LIFE GOALS AND WELL-BEING: THE ROLE OF WORK-FAMILY INTERFERENCE

A thesis submitted for the degree of Doctor of Philosophy in PSYCHOLOGY

> by GAMZE IPEK

Defense Committee Members

Prof. Dr. Sonia Lippke, Constructor University Bremen (Supervisor/Reviewer)
Prof. Sonja Drobnič, PhD, University of Bremen (Supervisor/Reviewer)
Prof. Roy F. Baumeister, University of Queensland (Supervisor/Reviewer)
Prof. Dr. Klaus Boehnke, Constructor University Bremen (Examiner)
Prof. Dr. Ulrich Kühnen, Constructor University Bremen (Examiner)
Dr. Franziska Deutsch, Constructor University Bremen (Examiner)
Carina Hartz (Advisory Member)

Date of Defense: December 6, 2024 Bremen International Graduate School of Social Sciences

Slow down and enjoy life.

It is not only the scenery you miss by going too fast

- you also miss the sense of where you are going and why.

-Eddie Cantor

Acknowledgments

First and foremost, I would like to extend my heartfelt thanks to my advisors for their unwavering support, invaluable contributions, and trust. I am grateful to Prof. Dr. Sonia Lippke for her incredible support and guidance. Her encouragement and the confidence she instilled in me have been instrumental in my development as a researcher, which I have become after three years. I also extend my gratitude to Prof. Sonja Drobnič, PhD for her insightful guidance and valuable feedback. Her attention to detail and analytical approach significantly enhanced the quality of my thesis. Meeting and working with Prof. Roy F. Baumeister has been a great honor and a source of immense pride for me. I am profoundly thankful for his inspiring suggestions and motivational support.

I am also grateful to the esteemed Principal Investigators of the BIGSSS. I extend my gratitude to Prof. Klaus Boehnke for his unwavering support and for his consistent encouragement, which emphasized his confidence in my abilities. My sincere thanks go to Dr. Franziska Deutch for her open-door policy, constant support, and motivational attitude, which were vital to me.

I would like to thank all my friends at BIGSSS, particularly dear Carina Hartz, Tanya Keni, Pay Laurin Jessen, and Dariya Koleva, with whom I navigated this challenging journey. Their friendship and support made this journey more manageable and enjoyable.

I am appreciative to my dear friend Dr. Ferhat Ayyıldız, who has been by my side at every milestone of my academic career. His continuous encouragement and friendship have always been a source of inspiration for me to reach my academic goals.

To my dear husband and best friend, Barış Açık, I owe immense gratitude for his emotional support and love. His patience and understanding gave me strength throughout this challenging process and were key to my success. Lastly, I extend my deepest thanks to my beloved parents, Meltem Ipek and Hüseyin Ipek, and my brother, Şafak Göktuğ Ipek, who have always believed in me and supported me in all my endeavors. Their unconditional love and support have been a constant source of inspiration throughout my life.

Table of Contents

Chapter 1: General Introduction
1.1. Background and Outline of the Dissertation
1.2. Theoretical Framework
1.3. Work-Family Interference
1.4. Subjective Well-Being
1.5. Life Goals
1.6. Inter-Cultural Differences: Societal Expectations and Non-Traditional Goal Pursuit27
1.7. Personal Differences: Self-Control
1.8. Aim of the Dissertation
Chapter 2: Work-Family Interference and Life Satisfaction: The Role of Life Goals45
2.1. Introduction
2.2. Theoretical Framework
2.3. Method
2.4. Results
2.5. Discussion
Chapter 3: Exploring the Mediating Role of Work-Family Interference in the Complex
Dynamics of Life Goals and Well-Being
3.1. Introduction
3.2. Theoretical Framework
3.3. Method
3.4. Results

3.5. Discussion	
Chapter 4: Examining the Impact of a Self-Regulation Intervention on Work-	to-Family
Interference	114
4.1. Introduction	116
4.2. Method	119
4.3. Results	126
4.4. Discussion	133
Chapter 5: General Discussion	139
5.1. Main Research Objectives and Outcomes	140
5.2. Main Contributions	148
5.3. Practical Recommendations	152
5.4. Main Limitations and Suggestions for Future Research	154
References	158
Appendix	

List of Abbreviations

BCT: Behavior Change Techniques FtWI: Family-to-Work Interference HA_HC: High Agentic–High Communal Goal Endorsement HA_LC: High Agentic–Low Communal Goal Endorsement LA_HC: Low Agentic–High Communal Goal Endorsement OECD: Organization for Economic Cooperation and Development pairfam: German Family Panel Survey WEF: World Economic Forum WtFI: Work-to-Family Interference

List of Tables

Table 2. 1: Path Analysis of Primary Relationships
Table 2. 2: Nested Model Comparisons in The Relationship Between Work-Family Interference
and Life Satisfaction among Individuals with Different Life Goals
Table 2. 3: Path Analysis of The Moderating Role of Life Goals
Table 2. 4: Nested Model Comparisons in The Relationship Between WtFI and Life
Satisfaction among Men and Women with Different Life Goals
Table 2. 5: Nested Model Comparisons in The Relationship Between FtWI and Life
Satisfaction among Men and Women with Different Life Goals
Table 2. 6: Path Analysis of The Moderating Role of Life Goals and Gender
Table 3. 1: Hypothesis 1: The Relationship Between Well-Being and Goal Endorsement85
Table 3. 2: Hypothesis 2: The Mediating Role of Work-Family Interference 89
Table 3. 3: Hypothesis 3: The Moderating Role of Self-Control 91
Table 3. 4: Hypothesis 4: The Moderating Role of Societal Expectations Concerning Gender
Roles
Table 4. 1: Socio-Demographic Characteristics of The Participants: Number of Study
Participants and Percentages Per Group
Table 4. 2: Correlation Matrix 127
Table 4. 3: Means and Standard Deviations
Table 5. 1: Research Questions and Hypotheses Across Studies 141
Table 5. 2: Summary of Hypotheses and Their Empirical Results 142
Table A2. 1: Descriptive Statistics 186
Table A2. 2: Correlation Matrix 187
Table A3. 1: Descriptive Statistics 191
Table A3. 2: Mean Level Differences 192

Table A3. 3: Correlation Matrix 193
Table A3. 4: The Relationship between Agentic Goals, WtFI, Life Satisfaction, and Stress after
Bootstrapping194
Table A3. 5: The Relationship between Communal Goals, FtWI, Life Satisfaction and Stress
after Bootstrapping
Table A3. 6: The Relationship between Agentic Goals, WtFI, Life Satisfaction, and Stress after
Bootstrapping for Low Self Control Group
Table A3. 7: The Relationship between Agentic Goals, WtFI, Life Satisfaction, and Stress after
Bootstrapping for High Self Control Group
Table A3. 8: The Relationship between Communal Goals, FtWI, Life Satisfaction, and Stress
after Bootstrapping for Low Self Control Group198
Table A3. 9: The Relationship between Communal Goals, FtWI, Life Satisfaction, and Stress
after Bootstrapping for High Self Control Group199
Table A3. 10: The Relationship between Agentic Goals, WtFI, Life Satisfaction, and Stress
after Bootstrapping for German Women
Table A3. 11: The Relationship between Agentic Goals, WtFI, Life Satisfaction, and Stress
after Bootstrapping for Turkish Women
Table A3. 12: The Relationship between Communal Goals, FtWI, Life Satisfaction and Stress
after Bootstrapping for German Men
Table A3. 13: The Relationship between Communal Goals, FtWI, Life Satisfaction and Stress
after Bootstrapping for Turkish Men
Table A4.: Intervention and Control Group Differences Regarding Baseline Self-Control
Scores

List of Figures

Figure 1. 1: Work-Home Resources Model
Figure 1. 2: Life Goals
Figure 2.: Research Model
Figure 3. 1: Research Model
Figure 3. 2: The Relationship between Agentic Goals, WtFI, Life Satisfaction, and Stress 100
Figure 3. 3: The Relationship between Communal Goals, FtWI, Life Satisfaction, and Stress
Figure 4. 1: Flow Chart and Study Design120
Figure 4. 2: Estimated Means of Work-to-Family Interference at T1 and T2 for Improvers 130
Figure 4. 3: Estimated Means of Work-to-Family Interference at T1 and T2 for Reducers131
Figure 4. 4: Estimated Means of Work-to-Family Interference at T1 and T2 for Stables132
Figure A2. 1: Changes in Mean Values of WtFI from T1 to T2 for Different Life-Goal Groups
Figure A2. 2: Changes in Mean Values of FtWI from T1 to T2 for Different Life-Goal Groups
Figure A2. 3: Changes in Life Satisfaction from T1 to T2 for Different Life-Goal Groups 190
Figure A4.: Action Planning and Action Control Exercise

Executive Summary

The objective of this doctoral thesis is to empirically test theory-based hypotheses related to the role of work-family interference in the relationship between life goals and well-being. While the relationship between work-family interference and well-being is well-documented in the literature, the role of individual motivational function of life goals has been largely overlooked. Life goals are conceptualized as agentic, representing independence and success, and communal, representing interdependence and care. Additionally, while some studies have explored the relationship between life goals and well-being, little is known about how these life goals affect different life domains. The study draws on established theoretical frameworks, including the Work-Home Resources model (ten Brummelhuis & Bakker, 2012), Conservation of Resources theory (Hobfoll, 1989), Personal Resource Allocation framework (Grawitch et al., 2010), and the Strength Model of Self-Control (Muraven & Baumeister, 2000), to highlight the critical role of managing personal and contextual resources in mitigating negative well-being outcomes while balancing work-family dynamics and pursuing life goals.

The overarching research question this thesis addresses is: What is the relationship between life goals, work-family interference, and well-being? Three empirical studies were conducted as part of this thesis, testing the following core hypotheses: (1) Life goals modify the impact of work-family interference on well-being; (2) Life goals act as antecedents to workfamily interference, with effects varying across cultural contexts (Turkey and Germany) and levels of self-control; (3) Self-regulation interventions aimed at reducing work-family interference may have varying effects depending on individual differences in self-control.

The first study, utilizing secondary data from the German Family Panel (pairfam) with N=2656 participants residing in Germany, supports the hypothesis that life goals significantly moderate the relationship between work-family interference and life satisfaction. Specifically, individuals who prioritize agentic goals tend to experience lower life satisfaction when work

interferes with family ($\beta = -0.18$, $p \le 0.001$), while those who prioritize communal goals may face a similar decline in life satisfaction when family interferes with work ($\beta = -0.07$, $p \le 0.05$). These results underscore the importance of understanding the type of life goals individuals prioritize as they critically shape how work-family interference affects overall life satisfaction.

The second study, using primary data from N=1252 white-collar employees in Germany and Turkey, expands on the findings of the first study by providing further support for the hypothesis, showing that life goals not only moderate the relationship between work-family interference and well-being but also serve as antecedents to it. Specifically, focusing on agentic goals is positively associated with work-to-family interference ($\beta = 0.30$; p < .001), while focusing on communal goals is positively associated with family-to-work interference (β = 0.19; p < 0.001), both of which contribute to a decline in life satisfaction ($\beta = -0.12$; p < .001; $\beta = -0.05$; p < .001) and an increase in stress ($\beta = 0.16$; p < .001; $\beta = 0.08$; p < .001). Furthermore, cultural context and self-control levels significantly moderate these relationships. For example, Turkish women experience a greater positive impact on stress when their agentic goals conflict with societal gender role expectations ($\beta = .14$; p < .05). Individuals with high self-control, despite higher life satisfaction ($\beta = .17$; p < .01), often underestimate the demands of communal goals, resulting in increased family-to-work interference ($\beta = .29$; p < .001). These findings highlight the complex interplay between life goals, cultural context, and selfcontrol in shaping the experience of work-family interference and its impact on well-being, suggesting the need for targeted interventions that address the specific challenges posed by agentic and communal goals within different cultural contexts. To address these challenges, the third study implements action planning and action control interventions aimed at helping individuals better manage work-family interference and enhance well-being.

The third study assessed the effectiveness of self-regulation intervention designed to reduce work-to-family interference in N=662 white-collar employees residing in Germany.

Contrary to the hypotheses, the results showed no significant differences in work-to-family interference between the intervention and control groups (F(1, 427) = .055, p = .82), indicating that the intervention had no overall effect. However, further analysis revealed that changes in self-control were key. Participants in the intervention group who showed moderate improvements in self-control initially experienced an increase in WtFI, suggesting that high self-control may, under certain conditions, exacerbate rather than alleviate role interference. In contrast, those with greater self-control improvements experienced a decrease in WtFI over time, suggesting long-term benefits. This challenges the assumption that self-control always leads to better management of work-family demands and highlights the need for more nuanced interventions tailored to individual differences.

These three studies collectively draw on and extend the theoretical frameworks introduced earlier. The findings demonstrate that life goals play a pivotal role in both moderating and predicting the relationship between work-family interference and well-being, highlighting the importance of resource allocation between work and home, the preservation and enhancement of resources, as well as the role of life goals in guiding resource investment across different life domains. Additionally, the findings challenge the strength model of selfcontrol by revealing that, under certain conditions, high self-control can exacerbate rather than alleviate role interference. These insights underscore the need for tailored interventions that consider personal and cultural differences in life goals and self-control to effectively manage work-family interference and improve well-being, providing actionable insights for future research and practical interventions. **Chapter 1: General Introduction**

1.1. Background and Outline of the Dissertation

Imagine individuals striving for a doctoral degree, which requires them to spend long hours at their desks while also being responsible for the daily care of their elderly parents. Picture a person who yearns for regular contact with friends but is challenged by a demanding six-day workweek. Such scenarios are commonplace in daily life, where personal goals, professional commitments, and family responsibilities intertwine. The effort to balance these aspects has not only been a significant concern from academic and professional viewpoints for decades but failing to achieve such balance can also have harmful effects on well-being.

Particularly with technological advancements and the increasing prevalence of remote work options, the boundary between life domains has become increasingly blurred. For instance, for white-collar professionals who predominantly work via computers, seamlessly checking emails at home with a click of a button or swiftly shifting focus to their children's situation in the kindergarten with a phone call from the office has become effortless. According to the American Psychological Association's 2023 Work in America Survey, employees lacking the flexibility to balance work and personal life were significantly more prone to report negative impacts on their mental health (67%) compared to those with such flexibility (23%). Thus, it is understandable that employees were attracted to their current positions more because of the work-life balance opportunities (41%) than their salary (36%; Aviva, 2022). Economic reasons also appear to play a role in the need to improve the health and well-being of employees. In the McKinsey Report, it was stated that improving the health and well-being of employees on a global scale has the potential to generate economic value ranging from \$3.7 trillion to \$11.7 trillion (Brassey et al., 2024). Thus, exploring avenues for achieving a balance between work and personal life becomes increasingly crucial, particularly considering their profound implications for individual well-being.

Taking all these situations/conditions into consideration, the purpose of this thesis is to understand how the interference between different life domains, such as work and family, impacts the well-being of white-collar employees. It also aims to understand the role of life goals as factors that may both contribute to or modify these effects, as well as how these life goals themselves may influence individual well-being. In doing so, it considers not only the importance of individuals' life goals but also their ability to pursue them with the ability of selfcontrol. Additionally, this thesis takes into account the influence of the cultural context in which individuals live, as it can either support or burden them in their efforts to achieve their life goals while also affecting their ability to cope with the work-family dynamics. By addressing both the personal and cultural dimensions, the thesis aims to provide an understanding of the factors influencing employee well-being.

The relationship between work-family interference and well-being has been frequently addressed in academic literature (Mullen et al., 2008; Obrenovic et al., 2020; Reimann & Diewald, 2022) and in various theories and models, such as the Conservation of Resources Theory (Hobfoll, 1989), Boundary Theory (Ashforth et al., 2000), and Job Demands-Resources Model (Bakker & Demerouti, 2007). In fact, Amstad et al. (2011) conducted a meta-analysis that revealed work-family interference has the greatest impact on individuals' overall well-being, compared to its effects on work- or family-related outcomes such as job satisfaction and marital performance. This finding highlights how work-family interference can substantially affect the quality of life. However, the effect on well-being can still vary significantly among individuals. For instance, factors such as spousal support and personality traits can either buffer or exacerbate the impact of work-family interference (Aycan & Eskin, 2005; Kinnunen et al., 2003). Researchers have also focused on identifying antecedents to understand the emergence of work-family interference, highlighting both domain-specific factors, such as workload and family responsibilities, and dispositional factors, such as stress management and personality

traits (Allen et al., 2012; Michel et al., 2011). While these studies have provided valuable insights, there appears to be a need for further exploration into the role of individual-related and particularly motivational factors in the work-family interference literature. Despite numerous calls from scholars (Carlson & Kacmar, 2000; Hauser et al., 2018; Senecal et al., 2001; Shaffer et al., 2011; Xanthopoulou et al., 2007), the existing literature largely overlooks how individuals' life goals (motivational factors) might influence their experience of work-family interference across diverse contexts. This thesis aims to contribute to the understanding of how individuals' life goals, which influence decision-making and purposive actions (Emmons, 1996; Gollwitzer & Bargh, 1996), shape work-family dynamics.

Life goals are internal representations of desired states (Austin & Vancouver, 1996) that individuals strive to achieve, preserve, or avoid (Emmons, 1996, p. 314), and they can directly impact individual well-being as the process of achieving them provides personal meaning and satisfaction (Deci & Ryan, 2000). However, individuals pursuing specific life goals may be more sensitive to the interaction between work and family domains and thus may be more negatively affected by this interaction, which suggests a moderating role for life goals in shaping well-being. Additionally, while some studies have explored the relationship between life goals and well-being (Macleod, 2012; Saragovi et al., 2002; Sheldon & Elliot, 1999), little is known about how these life goals affect different life domains. This thesis investigates the moderating and predecessor roles of life goals in the relationship between work-family interference and well-being, underscoring the need for further research on how to manage work-family boundaries, effectively achieve life goals, and maintain well-being in this process.

When considering the relationship among life goals, work-family interference, and wellbeing, it is crucial to acknowledge the significant roles of cultural values because they dictate societal expectations regarding gender roles and acceptable behavior. These expectations profoundly influence how individuals prioritize their life goals and navigate the balance between work and family responsibilities. The misalignment between personal aspirations and societal expectations can result in stress, reduced well-being, and a sense of failure in fulfilling both work and family roles (Hagqvist et al., 2017; Kaufman & Taniguchi, 2019). The profound effect of these expectations lies in their ability to shape not only individual behaviors but also societal structures, creating barriers to pursuing life goals that conflict with traditional roles. Furthermore, as societal norms evolve, they can either mitigate or exacerbate the effect of work-family interference on well-being, illustrating the dynamic interplay among cultural values, societal expectations, and individual well-being. Conducting studies across diverse cultural contexts is essential for understanding this complexity and revealing how these forces impact the endorsement of life goals and well-being in different societies. In this regard, this thesis particularly focuses on the impact of pursuing life goals that represent non-traditional gender roles on experienced work-family interference and its effect on well-being in Germany and Türkiye. Based on Hofstede's (2001) cultural dimension of individualism and collectivism, it may be assumed that in Germany, where individualistic values dominate, pursuing life goals aligned with independence and success may result in lower levels of work-family interference and higher well-being, as societal norms are more supportive of individual choices. In contrast, in Türkiye, where collectivistic values prevail, pursuing life goals that emphasize interdependence and caring may lead to less work-family interference and enhanced wellbeing, as these goals align more closely with societal expectations. Examining these issues in countries with differing cultural characteristics will reveal the influence of cultural context on the relationship among life goals, work-family interference, and well-being.

Besides cultural differences, personal differences, such as individuals' self-control skills, play a crucial role in how effectively people pursue their life goals, as research indicates that individuals with higher self-control are better equipped to manage competing demands, resulting in more successful goal achievement (Tangney et al., 2004; Hofmann et al., 2012). In this thesis, the focus is on self-control as a mechanism to overcome obstacles that may arise while pursuing life goals (de Ridder et al., 2012). The strength model of self-control (Muraven & Baumeister, 2000) suggests that self-control can be exercised and strengthened over time, much like muscle. Targeted self-control training not only improves regulation across various domains of life but also contributes to overall well-being (Beames et al., 2017; Friese et al., 2017). These interventions offer individuals the tools to pursue their life goals more effectively while maintaining a healthy balance between different life domains without the need to retreat from external pressures. For example, studies evaluating systematic self-monitoring and reflection strategies have shown improvements in employee well-being and reductions in sick leave (Krampen, 2010). The literature highlights the need for interventions that help employees manage conflicting demands more efficiently (Richardson, 2017). However, research on the moderating effect of self-control and targeted interventions designed to improve self-control for better goal pursuit and management of the dynamics between different life domains remains limited. Therefore, this thesis seeks to test the effectiveness of a self-regulation intervention aimed at promoting self-control and reducing work-family interference.

Based on these explanations, this doctoral thesis tests specific hypotheses concerning the relationship among life goals, work-family interference, and employee well-being. It examines how work-family interference affects employee well-being and how life goals contribute to managing work and family demands. Additionally, the moderating effects of cultural and personal differences are analyzed to determine how these factors enable employees to address work-family challenges. By testing these hypotheses, this thesis provides evidence-based insights into how employees achieve a balance between their professional and personal responsibilities. To achieve these goals, the current chapter, Chapter 1, outlines key theoretical frameworks and reviews the literature on the relationship among life goals, work-family interference, and well-being, considering the role of intercultural and personal differences. It

concludes by outlining the research aims and objectives. Chapter 2 presents a longitudinal study comparing work-family interference and well-being among employees with different life goals, investigating how life goals may moderate work-family interference experiences. Chapter 3 explores cross-sectional data from white-collar employees in Germany and Türkiye, focusing on how life goals drive work-family interference and influence well-being. Chapter 4 assesses a self-control intervention aimed at helping employees improve their ability to manage competing demands. Finally, Chapter 5 summarizes the key findings, outlines limitations, and discusses the theoretical and practical contributions, providing recommendations for future research and applications.

1.2. Theoretical Framework

Before moving forward, this section will present a broad theoretical framework that explains the key conceptual relationships discussed in the following chapters and clarifies the rationale behind the choice of these concepts. Several resource-based theoretical models are used as foundational frameworks to help clarify the relationships between key concepts.

The Work-Home Resources model (ten Brummelhuis & Bakker, 2012) is a model that examines the interplay between work and home domains, focusing on the allocation and utilization of resources across these domains to understand their impact on individual wellbeing (see Figure 1.1). The model defines work-family interference as a process whereby demands in either the work or home domain can drain personal resources, reducing effectiveness in the other area (ten Brummelhuis & Bakker, 2012). It is grounded in the Conservation of Resources theory (Hobfoll, 1989), which posits that individuals strive to obtain, preserve, and enhance their resources, such as time and energy, which are critical for effectively managing both work and home responsibilities. Hobfoll (2011) categorized the resources into two types: contextual resources, external to the self and located within one's social environment, and personal resources, which are personal characteristics and energies (Hobfoll, 2011). The Work-Home Resources model introduces two additional subtypes of resources: key resources and macro resources. Key resources are placed at a higher level within personal resources, as they encompass stable personality traits like the intensity of goal pursuit. Likewise, macro resources are positioned at a higher level within contextual resources, representing more enduring and uncontrollable factors such as the cultural system (ten Brummelhuis & Bakker, 2012).

Figure 1.1

	Work-Home	Resources	Model
--	-----------	-----------	-------

Macro Resources Culture Social equality Wealth Public policies Marr Emp Horr Social	Objects/ Conditions iage loyment e al network	Constructive Resources Skills Knowledge Experience Mental resil Health	Key Re Self-eff Self-es Optimi Social	sources ficacy teem sm power
 Social S Affe Advi Resp Instr from other 	upport ct, Love ce ect umental help significant 's	Energies Mood Physical energies Cognitive energies Attention Time 	ergy hergy	
(Contextual So	Person urce	nal	

Note: Taken from ten Brummelhuis and Bakker (2012).

Self-regulation can be considered one of the personal resources in the Conservation of Resources theory and key resources in the Work-Home Resources model. It is defined as the

ability to control emotions, thoughts, and behaviors in pursuit of life goals (Zimmerman, 2000). It has four components: Individuals establish specific standards and desired behaviors, such as life goals. They maintain motivation to achieve their life goals or meet these standards. They regularly assess and track their actions and thoughts to monitor progress. Finally, they exercise willpower to resist temptations or impulses (Baumeister & Vohs, 2007). Self-regulation encompasses more than just suppressing automatic responses (referred to as self-control); it also involves directing behavior toward desired outcomes (referred to as goal striving; Berkman, 2016). According to the limited-resource model of self-regulation, the self consists of energy and is governed by depletable, limited self-regulatory resources (Baumeister & Vohs, 2007). This depletion is referred to as ego depletion and is explained through muscle metaphor. Just as muscles tire when heavily used and require rest to be effectively reused, the resources necessary for self-regulation can also diminish over time (Muraven & Baumeister, 2000). Limited self-regulation resources lead individuals to be motivated to conserve them rather than expend them until depleted (Muraven et al., 2006).

The Personal Resource Allocation framework (Grawitch et al., 2010), which is similar to and based on the models mentioned above, also focuses on the concept of resources and places significant emphasis on resource investment. According to the Personal Resource Allocation framework, effective resource allocation across all life domains is the key to maintaining balance between those domains. Individuals have limited resources, and as they face repeated demands on these resources, they must choose how to allocate them effectively (Grawitch et al., 2010). Self-regulation as a key personal resource enables resource management by facilitating the selection and application of resources across domains (Hagger, 2015). Nevertheless, the act of choices depletes essential resources required for self-regulation, suggesting that it potentially leads to resource depletion (Vohs et al., 2008). Therefore, whether individuals choose to invest their resources or are compelled to do so, this investment ultimately drains their personal resources (Grawitch et al., 2010). There is a paradox here: self-regulation requires resources to function effectively, yet the very act of self-regulation also consumes these limited resources (Baumeister et al., 1998).

It is important to emphasize that each choice of where, when, and how resources will be invested involves some trade-offs. Tradeoffs are central to self-regulation, and paradoxically, the self depletes its resources when deciding how to allocate them (Baumeister & Alquist, 2009). Decision-making and self-regulation are integral components of the self's executive function; thus, effective self-regulation often necessitates sacrificing one desire in order to achieve another that is more important (Vohs et al., 2008). The pursuit of a significant, longterm goal can provide a clear sense of direction and purpose, which helps individuals prioritize their efforts and make necessary sacrifices more easily (Baumeister & Vohs, 2007). Selfregulation decisions involve value-based judgments, where choices are assessed according to their subjective worth (Berkman et al., 2017). This idea is consistent with the Conservation of Resources theory, which posits that individuals typically allocate resources to preferred activities while minimizing expenditure on those considered necessary but less favored (Hobfoll, 2011). In the realm of achieving a balance between life domains, when individuals allocate their personal resources broadly to meet all desires rather than prioritizing essential ones, they may experience dissatisfaction and psychological distress due to resource insufficiency and unmet expectations (Grawitch et al., 2010).

All the outlined approaches have also addressed the origins of stress, as well as individuals' reactions to it. These models do not treat demands and stressors as the same concepts. On the contrary, while demands are factors or responsibilities that compete for personal resources, stress represents the mismatch between these demands and the available resources (Grawitch et al., 2010). Stress occurs when there is a perceived mismatch between demands and available resources, a threat of resource loss, an actual loss of resources, or a

failure to gain resources following their investment (Hobfoll, 1989). Resource availability emerges as a crucial factor in dealing with stress; thus, the more individuals have selfregulation skills, the better they can cope with stress (Hagger, 2015). Cyclically, coping with stress requires self-regulation, and subsequent attempts at self-regulation are more likely to fail after such exertions (Muraven & Baumeister, 2000). Hobfoll defined this process as the loss spiral. The loss spiral describes a process in which an initial loss in personal resources leads to further losses because there are fewer personal resources available to cope with ongoing demands effectively (Hobfoll, 2011). Theoretical frameworks that apply the loss spiral process from the Conservation of Resources theory to specific domains like work-family interference explain how demands in one area deplete personal resources. This depletion leaves insufficient resources for other areas, thereby elucidating the stressors that arise. These stressors manifest in outcomes related to work, home, and overall well-being (Grawitch et al., 2010; ten Brummelhuis & Bakker, 2012). Thus, the negative consequences in the work, family, and health-related areas result from ineffective management of life given a finite number of personal resources (time, energy, and money) (Grawitch et al., 2010).

In conclusion, the theoretical frameworks discussed in this chapter provide a comprehensive foundation for understanding the interplay among life goals, work-family interference, and individual well-being. By examining resource-based models, valuable insights are offered into how individuals allocate and manage their limited resources across life domains. This selective resource investment opens the door for trade-offs, where certain life goals and domains are prioritized at the expense of others. The concepts of self-regulation, resource depletion, and the loss spiral are crucial for explaining how resource imbalances can lead to stress and dissatisfaction in both work and personal life. These theories serve as a guide in shedding light on the mechanisms that influence individual well-being in the context of life goals and work-family interference.

In the subsequent sections of this chapter, each key concept of the thesis is explored in depth, offering a thorough review of the current knowledge in the field.

1.3. Work-Family Interference¹

In organizational and social psychology contexts, a role is conceptualized as a set of activities (Kahn et al., 1964). Within role theory, the scarcity hypothesis is another model grounded in resource allocation that suggests that the concurrent management of multiple roles typically leads to strain, given that individuals have a limited number of resources to allocate across their various roles (Goode, 1960). Another perspective, known as the expansion approach, proposes that human energy and time are not fixed but rather flexible and capable of growth. This perspective posits that being involved in multiple roles can be perceived as enriching and fulfilling, offering significant benefits to individuals (Marks, 1977). Given that negative experiences tend to have a more profound impact on human lives than positive ones (Baumeister et al., 2001), this doctoral thesis will mainly focus on the negative interactions between roles.

Kahn et al. (1964) defined the notion of negative spillover between roles as "role conflict," which occurs when two or more demands from different roles arise simultaneously, making it difficult to meet one without compromising the others. They identified several forms

¹ In the academic discourse, there's an awareness that the term "family" might not fully capture the experiences of individuals outside traditional family units, prompting suggestions for broader terms like "non-work," "home," and "private life" to describe personal life domains more inclusively. Despite the importance of these broader concepts, this doctoral thesis employs the term "work-family interference." This choice is grounded in its prevalent use among scholars in the field. Nevertheless, it still encompasses not only traditional family units but also other forms of life arrangements.

of role conflict, including inter-role conflict, which specifically refers to clashes between demands stemming from different group memberships (Kahn et al., 1964). Building on these definitions by Kahn et al. (1964), Greenhaus and Beutell (1985) defined work-family interference as a specific type of inter-role conflict where the demands from work and family spheres are mutually incompatible to some extent. This means that participation in either domain (work or family) is made more challenging by obligations in the other. They identify three types of interference: time-based, where time commitments to one role impair fulfillment in another; strain-based, where stress from one role impedes performance in another; and behavior-based, where the behaviors requisite in one role hinder fulfillment in another role (Greenhaus & Beutell, 1985).

Furthermore, work-family interference is a bi-directional phenomenon, comprising work-to-family interference (WtFI), where professional obligations encroach on personal life, and family-to-work interference (FtWI), where personal life duties impede professional responsibilities (Frone et al., 1992). Research has also unveiled a reciprocal relationship between these two types of interference (Frone et al., 1992; Gutek, 1991), suggesting that experiencing one form of interference is likely to predispose an individual to encounter the other. A meta-analysis has confirmed the presence of both overlapping (reciprocal) elements and distinct aspects between these interferences, underscoring the importance of treating them as separate constructs in research designs (Mesmer-Magnus & Viswesvaran, 2005). When the classifications regarding the form and direction of interference are combined, six dimensions emerge, including time-based, strain-based, and behavior-based work-to-family interference for behavior-based forms of interference is limited (Geurts & Demerouti, 2003), with research predominantly focusing on conflicts rooted in time and strain. The current doctoral thesis follows that tradition, acknowledging the challenges of distinguishing the direction of

behavior-based conflicts due to the intertwined nature of work and family behaviors, and therefore focuses specifically on time- and strain-based interferences, which are widely recognized as the primary sources of conflict that individuals find most difficult to manage (Wei et al., 2022).

The antecedents contributing to work-family interference have been divided into three categories: domain-specific factors associated with work and family domains, as well as individual differences (Byron, 2005; Eby et al., 2005). The domain specificity hypothesis (Frone et al., 1992; 1997) suggests that work-specific characteristics are more strongly linked to WtFI, while family-specific characteristics are more strongly associated with FtWI. This hypothesis finds empirical support through meta-analytic investigations, consistently indicating that work-related factors (such as working hours, work involvement, and work support) have stronger correlations with WtFI than with FtWI and that family-related factors (such as the number of children, family commitment, and spousal support) exhibit stronger correlations with FtWI than with WtFI (Byron, 2005; Mesmer-Magnus & Viswesvaran, 2005; Michel et al., 2011). Thus, it is important to note that these effects are not exclusively one-way; both work and family characteristics can influence both WtFI and FtWI, though the strength of these associations differs. When examining studies that focus on individual characteristics, it is observed that common factors such as internal locus of control, positive/negative affect, and self-efficacy are associated with both types of interference (Allen et al., 2012; Michel et al., 2011). In addition to these findings, as emphasized by Byron (2005) and Allen et al. (2012), the majority of existing literature has primarily concentrated on antecedents based on situational factors, creating a significant gap in comprehending the influence of individualrelated or dispositional factors, such as motivational ones.

The resource-based theoretical frameworks, as discussed in the previous section, all emphasize that negative outcomes in different life domains occur when there is a misalignment between resource allocation and the demands placed on individuals (Grawitch et al., 2010; ten Brummelhuis & Bakker, 2012). Work-family interference is a form of resource depletion, which in turn creates stress and negatively impacts individuals' well-being. Previous studies have found that when resources are allocated to one domain at the expense of neglecting the demands of the other domain, interference between work and family can lead to various negative outcomes, such as decreased organizational commitment, reduced family performance, and increased burnout. The consequences of interference between the work and family domains can be categorized into two main types: work and family-related (domainspecific) and health-related (domain-unspecific) outcomes (Allen et al., 2020; Amstad et al., 2011).

The work-family interference literature explores how the interplay between work and family is associated with outcomes across different domains, with a particular emphasis on two key theories: the cross-domain principle and the matching hypothesis. The former, which is the more widely recognized of the two (Frone et al., 1992; 1997), suggests that WtFI is associated with family outcomes (receiving domain) more significantly than work outcomes (originating domain), whereas FtWI is associated with work outcomes (receiving domain) more strongly than family outcomes (originating domain). This is because when individuals face overwhelming demands in one domain, their ability to fulfill responsibilities diminishes, resulting in decreased performance and quality of life in the receiving domain. Meta-analytic results conducted by Ford and colleagues (2007) support this hypothesis, showing that variables in the work domain explain a substantial amount of variance in family satisfaction, while variables in the family domain explain a significant portion of the variance in job satisfaction.

On the other hand, according to the matching hypothesis (Amstad et al., 2011), WtFI is more strongly associated with work-related outcomes (originating domain) compared to family-related outcomes (receiving domain). Conversely, FtWI is more strongly associated with family-related outcomes (originating domain) than work-related outcomes (receiving domain). This pattern is due to cognitive evaluations, where blame is attributed to the domain where the interference originates, which consequently leads to more negative associations with outcomes in that domain compared to the receiving domain. Previous findings provide further support for the matching hypothesis and suggest that WtFI has stronger associations with work-related outcomes, such as job satisfaction, than with family-related outcomes, while FtWI is more closely associated with family-related outcomes, such as family satisfaction, than with work-related outcomes (Amstad et al., 2011; Shockley & Singla, 2011). Additionally, a meta-analysis by Nohe et al. (2015) found that the association between WtFI and job strain is stronger than that between FtWI and job strain. Even though this suggests that interference originating in one domain tends to be more strongly associated with outcomes within that same domain, it has been found that both forms of interference are most strongly associated with outcomes that are not specific to any domain but are domain-unspecific, such as overall well-being (Amstad et al., 2011).

Delving into the domain-unspecific outcomes of work-family interference reveals significant associations with key aspects of life. Meta-analyses have consistently shown that both WtFI and FtWI are negatively associated with life satisfaction, underscoring the detrimental associations of such interference with overall well-being (Allen et al., 2000; Amstad et al., 2011; Kossek & Ozeki, 1998). Furthermore, research by Amstad and colleagues (2011) highlights that both WtFI and FtWI correlate positively with stress, psychological distress, depression, and anxiety. The study by Yavas et al. (2008) also illuminates the positive correlation between both directions of interference and emotional exhaustion, further emphasizing the significant psychological toll. Additionally, evidence suggests that work-

family interference can have not only cross-sectional but also both immediate and long-term adverse effects on individuals' well-being (Grant-Vallone & Donaldson, 2001).

In summary, this section explored the core concepts of role theory, focusing on the dynamics between work and family domains and how conflicting demands lead to strain and negative outcomes. The section also covered the nature of work-family interference, emphasizing time-based and strain-based interference and their reciprocal relationship. Additionally, the antecedents that contribute to this interference were discussed. The work-family interference literature illustrates its impact on domain-specific outcomes, such as work and family performance, as well as domain-unspecific outcomes, including general well-being. Moving forward, the next section will delve deeper into these domain-unspecific outcomes, particularly the concept of subjective well-being. It will explore how work-family interference influences life satisfaction and stress, offering insights into the motivational perspective of subjective well-being.

1.4. Subjective Well-Being

Subjective well-being is an umbrella concept that reflects individuals' evaluations and feelings regarding their lives (Diener & Ryan, 2009). It is deeply rooted in both emotional and cognitive dimensions and involves individuals' emotional reactions—ranging from positive to negative affect—as well as their cognitive judgments of life satisfaction and fulfillment (Diener, 1984).

According to Diener and Ryan (2009), subjective well-being encompasses emotional reactions to life events, including feelings of joy and sadness, contributing to an individual's overall sense of well-being. Besides, life satisfaction —a cognitive dimension— emphasizes individuals' subjective evaluations of the pleasant and unpleasant aspects of their lives and their satisfaction across various life domains such as work, family, and personal life (Diener & Ryan, 2009). These judgments typically rest on comparisons with personal standards and

expectations, underlining the subjective nature of well-being (Diener et al., 2018). Within subjective well-being literature, a notable distinction is made between the top-down and bottom-up models. The top-down perspective posits that overall life satisfaction influences satisfaction within specific life domains. Conversely, the bottom-up approach views life satisfaction as a cumulative assessment derived from individual satisfaction across various important life domains (Schimmack et al., 2002). This model suggests that changes within these domains, as well as diverse life events, may cause variations in the overarching sense of life satisfaction (Pavot & Diener, 2008). Supporting the top-down perspective, this thesis follows the idea that understanding overall life satisfaction can offer valuable insights into other areas of life.

There are two different perspectives on well-being and happiness, originating from ancient Greek philosophy: The hedonic view and the eudaimonic approach. Subjective wellbeing offers a comprehensive perspective on well-being that adheres to the hedonic view, positing well-being as deriving from experiencing life positively through both affective evaluations and cognitive judgments (Diener, 1984). The hedonic view centers on subjective feelings and thoughts, which equates subjective well-being with happiness through pleasure and pain avoidance, even though eliminating negative states would not necessarily foster positive states (Diener et al., 2002). In contrast, the eudemonic approach considers well-being in terms of realizing one's potential and cultivating desirable qualities beyond just subjective happiness (Das et al., 2020). Unlike purely hedonic happiness, which is primarily subjective and based on personal pleasure, eudaimonic happiness is also tied to objective standards of what is considered a "good life" (Sheldon, 2018). In the current thesis, emphasis is placed on the former viewpoint—the personal/subjective perception of individual well-being, with a particular focus on life satisfaction and immediate perceptions of well-being rather than the pursuit of meaning. According to the catalog study of Das and colleagues (2020), the theories of subjective well-being offer a multifaceted understanding of how personal fulfillment, emotions, and evaluations intertwine to shape overall happiness—a fluid and continuous process where each aspect of subjective well-being informs and reshapes the others. Personal goals lead to various degrees of fulfillment, which, through personal orientation, can lead to further goals and activities, completing the cycle. Emotions and evaluations are both outcomes and informers of affective and cognitive experiences (Das et al., 2020).

Motivational theories of subjective well-being (Diener, 1984) delve into the role of individual goals, needs, and activities in fostering subjective well-being, suggesting that the pursuit and attainment of personal goals and the satisfaction of needs through various activities are central to one's sense of well-being (Steca et al., 2016). The personal goals that people set for themselves, which involve what they want to accomplish or avoid in different life domains, are one key to understanding human behavior and emotions (Diener, 1984). The attainment and accomplishment of important personal goals typically correlate with higher levels of subjective well-being (Diener & Ryan, 2009). This idea also aligns with the resource theory of subjective well-being (Diener & Fujita, 1995), which asserts that the level of satisfaction one experiences in life is directly related to the sufficiency of personal resources that enable people to fulfill their needs or make progress toward their personal life goals (Diener et al., 2018).

To sum up, this section has explored the concept of subjective well-being, emphasizing its emotional and cognitive dimensions and its role in shaping individuals' overall life satisfaction. Additionally, the section has addressed the hedonic and eudaimonic views of happiness, with the current thesis focusing on the subjective, hedonic perspective. Motivational theories of subjective well-being have underscored the importance of personal life goals and the fulfillment of needs in fostering well-being, emphasizing that the attainment of these goals significantly contributes to an individual's overall happiness. Building on these ideas, the next section will delve into the role of life goals, examining how the pursuit and achievement of these goals intersect with work-family interference and subjective well-being. This will provide a deeper understanding of how individuals navigate conflicting demands across life domains while striving for personal fulfillment.

1.5. Life Goals

Goals are essential components that give meaning to people's lives and contribute to the process by which individuals construe their lives as meaningful or worthwhile (Baumeister, 1991; Emmons, 2003). They are the internal representations of desired states (Austin & Vancouver, 1996) that "people seek to obtain, maintain, or avoid" (Emmons, 1996, p. 314). Understanding an individual's goals allows for a deeper insight into their current and future actions (Emmons, 1996), as these objectives serve as a guide for behavior (Gollwitzer & Bargh, 1996). Goals provide structure and meaning to daily life and help people construe their existence as meaningful, serving as vital components in the pursuit of well-being and a positive life (Emmons, 2003).

Clarifying the terms related to motivation is essential, as they were often used interchangeably. Motive dispositions represent mainly unconscious desires and are revealed indirectly (McClelland, 1985). In contrast, goals are explicitly conscious and differ from motive dispositions (Schmuck & Sheldon, 2001). Values are cognitive representations of what is desirable and indicate what one should do, as opposed to goals that indicate what one wants to do (Emmons, 1989). Values are abstract concepts that represent desirable end-states (Schwartz, 2012) and are broader in scope than goals (Emmons, 1989). While motives and values form the basis of broad, indirect principles that guide behavior, goals are more connected to an individual's intentions than motive dispositions and values (Schmuck & Sheldon, 2001). They serve as the link between motives/values and actions (Locke & Kristof, 1996).

Goals are structured hierarchically, highlighting the importance of grasping an individual's unique goal hierarchy for a comprehensive understanding of them, as noted by Wadsworth and Ford (1983). Lower-order goals serve higher-order goals, and these lower-order goals are context-specific, short-term, and can be easily substituted. On the other hand, higher-order goals are fewer in number, more abstract, can remain salient for longer periods, and are more important to the individual (Duckworth & Gross, 2014). According to Austin and Vancouver (1996), high-level goals are representations of the ideal self and world views and can be found at the top of the hierarchy. Below the high-level goals, there are personal strivings (Emmons, 1986), and at the bottom of the hierarchy, there are current concerns (Klinger, 1975), personal projects (Little, 1983), and life tasks (Cantor, 1990). Finally, the term life goals, which is used in this doctoral thesis, can be seen as high-level goals that individuals set as a fundamental reference point for planning their lives (Pöhlmann, 2001).

Researchers have explored the content of goals to understand what individuals are pursuing and identified several distinctions, such as approach versus avoidance goals (Elliot, 1999), self-determined versus externally controlled goals (Deci & Ryan, 2000), and intrinsic versus extrinsic goals (Kasser & Ryan, 2001). Bakan (1966) also made a distinction between two themes of motivation, naming them agency and communion. In Bakan's (1966) terms, agency refers to an organism's identity as an individual, while communion describes an individual's involvement in larger groups to which they belong. The distinction between agency and communion is based on where the focus is directed - either toward oneself or toward others (Helgeson, 1994). In other words, agentic goals are related to the self and focus on achieving personal success, autonomy, and demonstrating competence. Communal goals, on the other hand, are concerned with relationships and involve seeking affiliation, fostering cooperation, and supporting the community (Pöhlmann, 2001).
The literature has established various distinctions based on the concepts of agency² and communion, each encompassing a range of terminologies to describe related pursuits (see Figure 1.2). Agentic pursuits have been characterized by labels like "getting ahead" (Hogan, 1982), "individual strivings" (Wicker et al., 1984), "self-enhancement goals" (Schmuck & Sheldon, 2001), "self-achievement, power, variation" (Pöhlmann, 2001), "zero-sum goals" (Headey, 2008), "recognition and mastery" (Diekman et al., 2010), as well as "assertiveness and competence" (Abele & Wojciszke, 2014). Conversely, communal pursuits have been characterized by labels like "getting along" (Hogan, 1982), "harmony seeking" (Wicker et al., 1984), "group-enhancement goals" (Schmuck & Sheldon, 2001), "intimacy, affiliation, altruism" (Pöhlmann, 2001), "non-zero-sum goals" (Headey, 2008), "helping others and connecting with others" (Diekman et al., 2010), as well as "warmth and morality" (Abele & Wojciszke, 2014).

Figure 1.2

Life Goals

² Agency refers both to an individual's focus on achieving personal success and autonomy (e.g., life goals like self-achievement and power) and to the active initiation of actions, decision-making, and exertion of control. This thesis approaches the concept of agency in the former sense, as a focus on the self.



Originally, the concepts of agency and communion were theorized to operate within orthogonal (separate) dimensions of social cognition, representing distinct yet essential aspects of human functioning (Abele & Wojciszke, 2014). Although these two types of goals may seem to be on opposite ends of the spectrum, they are not mutually exclusive (Trapnell & Paulhus, 2011). As Wiggins (1991) explained, all possible combinations of agency and communion are feasible. This flexibility suggests that development in one domain does not inhibit growth in the other, indicating no essential conflict between them. Namely, while agency and communion are fundamentally orthogonal in content, their relationship is dynamic rather than static. They can be positively correlated due to shared positive aspects. Also, their relationship can manifest negatively due to differing perspectives of actors and observers (Abele & Wojciszke, 2014). This dynamic interplay, characterized by both theoretical and empirical inconsistencies, may suggest a curvilinear relationship between agency and communion (Imhoff & Koch, 2017). Thus, agentic and communal goals can be complementary and interrelated in many aspects of human motivation and behavior. The integration or balance of these goals can be key to personal well-being.

The link between life goals and well-being is complex and multifaceted. Striving for personal, enjoyable, and achievable goals fosters psychological well-being (Brunstein, 1993; Pöhlmann, 2001). Pursuing a goal—identifying what goal is being pursued (content),

understanding why it is being pursued (importance), and determining how well the actions taken to achieve it (achievability)-results in various impacts on well-being (Schmuck & Sheldon, 2001). Research indicates that the content of the goal has different effects on wellbeing. For instance, intrinsic goals usually enhance psychological well-being more than extrinsic ones (Kasser & Ryan, 1993). Studies have also highlighted how pursuing communal and agentic goals affects well-being differently. For optimal well-being, both agentic and communal goal pursuits are required. Helgeson (1994) introduced the unmitigated agency, an excessive focus on self, and the unmitigated communion, an excessive focus on others, to explain imbalances in focus. When one dominates without the other, it can lead to negative health outcomes such as poor social support, unhealthy behaviors, increased psychological distress, and lower well-being (Helgeson & Fritz, 1999). A higher proportion of intimacyrelated goals, compared to other goals, is associated with increased well-being (Colby et al., 1994). In contrast, a higher focus on achievement and power goals has been linked to reduced well-being (Emmons, 1991). However, agency-motivated goals have also been associated with positive mental health, as individuals who pursue these goals tend to be more effective than people who pursue other goals in managing both internal and external demands in their lives (Pöhlmann, 2001). Inconsistently, other findings suggest that both intimacy/altruism and achievement/power goals can be positively correlated with positive affect and life satisfaction (Saragovi et al., 2002; Sheldon & Cooper, 2008; Tucak Junaković, 2015). Goals aligned with core values significantly boost well-being by way of greater goal-attainment (Sheldon & Elliot, 1999; Sheldon et al., 2004). The essence of well-being lies not only in the pursuit of specific goals but also in fulfilling one's life plan (Sheldon & Kasser, 1995).

Moreover, the more personal resources people have, the more positive well-being outcomes will occur while pursuing life goals (Diener & Fujita, 1995). People often have multiple goals that intersect in many ways, and pursuing one goal can have negative consequences for the pursuit of others (Cantor & Blanton, 1996; Emmons & King, 1988). This is because pursuing multiple life goals requires long-term commitment and can create multiple demands (Cantor & Blanton, 1996). It aligns with role accumulation theories that suggest having multiple roles leads to role strain (Goode, 1960) and with stress models that emphasize the importance of resource depletion (Hobfoll, 1989). This means that when faced with multiple goal pursuits that hinge on limited resources, priority management becomes essential. Goal prioritization is crucial in such contexts, and it is a function of the goal's importance. In multiple goal contexts, valences and expectancies play significant roles in affecting goal prioritization and resource allocation (Sun & Frese, 2013). Unsworth et al. (2014) noted that goals linked to positive emotional values are more likely to receive prioritization. Furthermore, Fishbach et al. (2009) highlighted that individuals dedicate more resources to goals they regard as more important, whereas less salient goals are allotted fewer resources. While pursuing multiple life goals can lead to role strain and resource depletion, prioritizing goals based on their importance and emotional value is essential to manage resources effectively and achieve positive well-being outcomes.

Nevertheless, Greenhaus and Beutell (1985) argued that the more prioritized motivation may increase time commitment and produce a strain that may interfere with another domain. Building on this idea, considering that individuals generally adopt roles in alignment with their goals (Diekman et al., 2017; 2020), it can be posited that pursuits represented by agentic life goals, such as power, achievement, and mastery, are predominantly related to the professional domain. Consequently, individuals who prioritize these goals are likely to allocate their resources toward career development. Similarly, pursuits associated with communal life goals, such as affiliation, intimacy, and altruism, are inherently linked to the family domain. Consequently, those who prioritize communal goals are likely to allocate their resources within the home domain³. These relationships have been similarly established in prior research (e.g., Cinamon & Rich, 2002; Cohen, 2009). Regarding subjective well-being, it can be pointed out that subjective well-being is predicted to be influenced by the interaction between life events and life goals, with these events evaluated based on their significance to an individual's life goals (Emmons, 1996). Events are considered important when they impact what individuals are striving for. Research has shown that daily moods fluctuate based on positive and negative occurrences in domains that are significant to one's life goals. For example, people who value relationships feel more impact from social events, while those focused-on achievements react more to school or work events (Emmons, 1991). Similarly, it has been found that negative events that impact personal goals can deeply influence self-perception, often leading to increased self-focus and rumination (Lavallee & Campbell, 1995).

In summary, this section has highlighted the pivotal role of life goals in giving meaning to individuals' lives and shaping their well-being, focusing on the distinction between agentic life goals (personal success, autonomy, competence) and communal life goals (relationships, affiliation, community support). These distinctions provide a framework to examine how individuals allocate resources between professional and personal domains, particularly in the context of work-family interference. This thesis will explore how a focus on agentic goals may shift resources toward work life, potentially increasing family strain, while a focus on

³ Obviously, an individual might pursue agentic goals, like career advancement, with communal outcomes in mind, such as supporting their family. Conversely, they might seek communal goals, such as volunteering, with agentic outcomes in mind, like skill development. In this thesis, my focus is primarily on the goals individuals pursue, rather than their reasons, and the additional benefits to the unprioritized domain are beyond the main focus of my doctoral thesis.

communal goals may shift resources toward family life, impacting professional pursuits. The existing literature on life goals and well-being is inconsistent, especially regarding their role in work-family dynamics, which still remains underexplored. This thesis aims to investigate not only the complex relationship among life goals, well-being, and work-family interference but also the role of intercultural differences. The following section will provide a detailed explanation of these cultural differences and their potential impact on these relationships.

1.6. Inter-Cultural Differences: Societal Expectations and Non-Traditional Goal Pursuit 1.6.1. Individualistic Cultures vs. Collectivistic Cultures

This thesis explores how individuals' pursuit of non-traditional goals can be evaluated within the contexts of relative individualism and collectivism and how societal expectations from different gender roles affect individuals' well-being. Given the traditional gender roles that categorize men as providers and women as caregivers (Eagly & Wood, 2012), it is suggested that women engaging in communal goals, whereas men pursuing agentic goals meet the societal expectations for their gender role (Diekman & Eagly, 2008). In recent times, new goals and expectations have emerged for both men and women that go beyond their traditional roles (Boehnke, 2011). As people increasingly balance both their career and family responsibilities, the traditional division of labor is beginning to lose its significance, and traditional gender role expectations are questioned (Frear et al., 2019). This shift leads individuals to experience interference between work and family in many ways.

The concept of culture has been previously addressed through the individualistic and collectivistic classification (e.g., Hofstede, 2011; Triandis, 2001). According to this, individualistic cultures view people as independent units focusing on personal interests and emphasizing individuals' internal feelings and thoughts. In comparison, collectivistic cultures value extended families and prioritize in-group relationships (Hofstede, 2001). Individualism represents a cultural value system that prioritizes individual rights, freedom, and personal

goals. In contrast, collectivism represents a cultural value system where personal goals are subordinated to group expectations (Suh & Oishi, 2002).

Markus and Kitayama (1991) highlighted how individualistic cultures view the self as independent, emphasizing personal traits and accomplishments, whereas collectivistic cultures view the self as interdependent, prioritizing relationships and the well-being of the group. Both independent -closely associated with the agency- and interdependent -closely associated with the communion- self-construal exist within everyone, but cultural contexts shape their expression (Markus & Kitayama, 1991; Wojciszke, 1997). Individuals from individualistic cultures derive their sense of self-worth from their capacity to express themselves and affirm their inner qualities. Conversely, individuals from collectivistic cultures find their self-worth in their ability to adapt and sustain harmony within their social environments (Markus & Kitayama, 1991). Challenging this binary, Kagitcibasi (1996; 2005) critiques the Western psychological framework that contrasts autonomy (i.e., agency) and relatedness (i.e., communion) as opposing traits. She introduces a model comprising three self-structures: the first type leads to a heteronomous-related self, the second to an autonomous-separate self, and the third indicates an autonomous-related self, prevalent in collectivist societies where industrialization and urbanization have reduced material dependencies, but emotional interdependence remains valued (Kagitcibasi, 2005).

Triandis (1995) differentiated individualist cultures from collectivist cultures by the goals people pursue, suggesting that individuals in individualistic cultures aim for personal goals reflecting their desires and needs. In contrast, those in collectivistic cultures pursue communal goals reflecting the desires and needs of in-group members. In individualistic cultures, personal goals are not typically associated with communal goals, but in collectivistic cultures, they tend to be (Triandis, 1995). Consequently, in individualistic cultures, pursuing agentic goals is valued, whereas in collectivistic cultures, engaging in communal goals is valued (Gebauer et

al., 2013; Sedikides et al., 2003), thereby influencing the well-being through the expectations associated with the goals pursued.

Building on this, subjective well-being is found to differ across nations, influenced in part by cultural values and norms (Diener et al., 2018). Even though research shows that there is a worldwide trend toward individualism and away from collectivism (Santos et al., 2017), and people in individualistic cultures generally report higher levels of happiness compared to those in collectivistic cultures (Diener & Diener, 1995; Suh & Oishi, 2002), acting in harmony with the values of one's culture still enhances an individual's well-being (Myers & Diener, 1995). Oishi and Diener (2001) discovered that in more collectivistic cultures, individuals' subjective well-being was boosted by pursuing relationship-oriented goals, while in more individualistic cultures, personal interest-driven goals were more intricately linked to enhanced subjective well-being. This means that individuals who adopt goals that diverge from traditional expectations may confront challenges related to acceptance and social harmony, potentially affecting their well-being. Research suggests that conformity to societal expectations tends to elicit positive emotional responses, whereas deviation often leads to negative emotional outcomes, such as "trouble" in life (Haines & Stroessner, 2019; Rudman et al., 2012; Sczesny et al., 2018).

From the perspective of work-family interference, which may be another "trouble" in life, as previously mentioned, the Work-Home Resources model discusses the concept of macro-resources, which are characteristics of the cultural system embedding an individual; these are stable and beyond individual control but determine the extent to which other resources can be effectively utilized (ten Brummelhuis & Bakker, 2012). The interplay between workfamily interference and cultural values, particularly individualism and collectivism, plays a crucial role in shaping the outcomes of this interference on well-being. The review by Shockley et al. (2017) highlights studies that examine cultural values as moderators in the outcomes of work-family interference. For instance, Aycan (2008) highlighted that WtFI is more negatively correlated with well-being in collectivistic cultures, where family is essential, and WtFI may threaten family identity. She further explains that in collectivistic cultures, family is deeply intertwined with personal identity, and WtFI is seen as a threat to this identity, amplifying its negative impact on well-being (e.g., Aycan & Eskin, 2005). Additionally, maintaining harmony and avoiding conflict in workplace relationships is considered a significant cultural demand, further contributing to strain-based WtFI. Conversely, in individualistic cultures where work is prioritized, FtWI emerges as more detrimental to well-being. Aycan (2008) explains that in these societies, work often defines personal identity, and when family interferes with work, it is perceived as an intrusion on one's sense of self. As a result, FtWI tends to have a more detrimental effect on well-being in individualistic contexts, where the importance of work outweighs other life domains. Similarly, it is hypothesized that in collectivistic cultures, WtFI negatively affects life satisfaction due to the threat it poses to family life, while in individualistic cultures, FtWI has a stronger impact on life satisfaction because of its influence on work identity (Poelmans et al., 2003). However, Allen et al. (2020) only partially support this perspective, as their findings suggest that the negative effects of both WtFI and FtWI on job, life, and family satisfaction are actually less pronounced in collectivistic societies. Similarly, in collectivistic cultures, work is often seen as a contribution to the family's welfare, which fosters a greater tolerance for spillover and a consequently weaker association between both WtFI and FtWI and stress-related outcomes (Spector et al., 2007; Yang, 2005). The collectivistic viewpoint that integrates work and family roles reduces the perceived interference between these domains (Allen et al., 2020). It has also been found that individualists tend to focus more on their own needs than collectivists, and thus, WtFI is viewed as a threat to their well-being (Aycan & Korabik, 2017). In individualistic societies, where a clear distinction between work and family exists, the balance between work and family is crucial and directly influences positive outcomes. Conversely, in collectivistic cultures, where such boundaries are more fluid, the emphasis on balance between work and family is less pronounced, reflecting a cultural adaptation that views work and family as integrated parts of life (Haar et al., 2014). Thus, just as the specific life goals one pursues are important, so is the harmony between these life goals and cultural expectations.

Finally, the concept of work-family guilt is identified as an emotional response triggered by the need to balance work and family commitments (Aycan, 2008; Korabik, 2017). This guilt primarily arises from failing to meet traditional gender expectations, where deviations from roles—men as providers and women as caregivers—challenge established norms (Livingston & Judge, 2008). Such deviations include men focusing on communal strivings and women on agentic strivings, leading to varied impacts on work-family interaction and overall well-being across different cultural contexts. Research by Aycan and Eskin (2005) in Türkiye revealed that women experience more employment-related guilt than men, with that guilt being more strongly associated with women's WtFI than FtWI. Furthermore, Korabik (2017) discovered that guilt from WtFI was more significant than guilt from FtWI across ten countries studied. In more individualistic and egalitarian societies, the higher levels of guilt associated with work interfering with family were directly correlated with adverse well-being outcomes. Conversely, those in more collectivistic countries reported lower levels of guilt associated with family interference with work, which was also linked to better well-being outcomes (Korabik, 2017).

These findings illustrate the complexity of how cultural values, gender roles, and workfamily interference interact to influence well-being. Specifically, the differences between how individualistic cultures emphasize personal achievement and agency while collectivistic cultures prioritize harmony and communion reveal the varying pressures individuals face in balancing work and family responsibilities. The apparent contradictions between these cultural expectations underscore the need to examine how the pursuit of non-traditional goals—such as women focusing on agentic life goals or men engaging in communal ones—can lead to different well-being outcomes. In this thesis, these cultural, gender, and work-family dynamics will be explored in the specific contexts of Germany and Türkiye, representing individualistic and collectivistic cultures, respectively. By focusing on these two countries, this thesis aims to investigate how societal expectations and cultural values shape individuals' experiences of work-family interference and the pursuit of non-traditional goals, contributing to a deeper understanding of the well-being outcomes associated with these cultural frameworks.

1.6.2. The Special Cases: Germany and Türkiye

It is essential to underscore that the comparisons drawn within this thesis, focusing on Germany and Türkiye, are relative and not meant to be exclusive or definitive. Each country inherently embodies a spectrum of values, notably individualism, and collectivism, which coexist within their societal frameworks. One of the main aims of this thesis is to examine how the cultural norms of two different countries shape expectations around gender roles and the impact of these expectations. It also seeks to provide a better understanding of the cultural dynamics involved.

According to Hofstede's (2001) comparison of cultural dimensions, Germany values individualism, small family units, and self-actualization, prioritizing personal loyalty and direct communication. German culture emphasizes work, performance, decisiveness, and the demonstration of status through material possessions. It shows that people rather "live in order to work." In contrast, Türkiye leans towards collectivism, highlighting the importance of group harmony, indirect communication, and loyalty within extended families and groups. Turkish culture values consensus avoids conflicts, and places importance on leisure time and relationships, with the status being more influenced by power distance (Hofstede, 2001).

Both Germany and Türkiye are members of the Organization for Economic Cooperation and Development (OECD). Based on the OECD Better Life Index 2020, which evaluates 41 countries, Türkiye is among the lowest four in terms of life quality, while Germany ranks in the top 13. When it comes to overall life satisfaction, Turkish people average a score of 4.9 out of 10, which is below the OECD's average of 6.7, positioning Türkiye at the bottom, in 41st place. In contrast, German people score an average of 7.3, which ranks Germany 8th among the 41 countries. The analysis reveals that Türkiye has the most significant gender gap in life satisfaction, with a great advantage for women and marking it as the country with the highest gender inequality in this regard. Germany, however, stands out for gender equality in life satisfaction, with a slight advantage for women, making it 10th in gender equality among the reviewed nations. Furthermore, 25% of Turkish employees work exceptionally long hours, one of the OECD's highest rates, placing Türkiye at 40th out of 41 countries. Meanwhile, in Germany, only about 4% of employees face such long hours, well below the OECD norm, placing it in the 16th spot. Similarly, Türkiye's employment rate is 47.52%, ranking it 40th, with a significant gender inequality leaning towards men, putting it at the very bottom in this aspect. Germany, on the other hand, boasts an employment rate of 76.69%, making it the 6th best among the nations studied, with its gender equality in employment placing it as the 18th best, reflecting a moderate advantage for men (OECD, 2020).

In the 2023 Global Gender Gap Report by the World Economic Forum (WEF), Türkiye ranks 129th out of 146 countries, showing the least progress in Economic Participation and Opportunity—being the only country with less than 60% of this gap closed. The country ranked 130th in terms of women's labor force participation rate, 91st in pay equality for the same job, 128th in estimated income, and 119th in access to executive positions. In contrast, Germany, ranking 6th, has closed over 80% of its gender gap, including a notable rise in women's labor force participation rate. The country ranked 49th in terms of women's labor force participation (WEF), and 95th in access to executive positions (WEF, 2023).

These factors-characteristics, employment trends, well-being, and gender equality statistics within Turkish and German settings-significantly influence personal strivings, as well as work-family dynamics. Türkiye's strategic position as a bridge between the East and West not only enhances its cultural diversity but also shapes its work-family practices. Turkish society integrates traditional values of family orientation and collectivism with modern, egalitarian, and individualistic values (Erarslan-Başkurt & Aycan, 2017). This synthesis of autonomy and relatedness, as conceptualized by Kagitcibasi (1996; 2005), may be a unique feature of Turkish culture. Similarly, Uskul and colleagues (2023) demonstrated that participants from the Mediterranean region (including Türkiye) tended to be relatively more independent in terms of social orientation and self-construal compared to those from Anglo-Western societies. However, they also exhibited stronger interdependence in measures emphasizing the connection between individuals and their groups (e.g., ingroup closeness, connection to others, commitment to others). Additionally, all these numbers can be seen as a reflection of how society prioritizes and values various life goals within itself. Therefore, the pursuit of non-traditional goals, leading to potential guilt and a 'double trouble' effect from work-family interference and not meeting cultural expectations, is worth examining for its negative impacts on well-being.

Given the distinctions outlined, it is crucial to recognize that both agentic and communal values are not mutually exclusive and coexist across various cultures. Individuals and societies have the ability to balance these values depending on the specific social circumstances. It is acknowledged that even in individualistic cultures like Germany, social responsibilities are still upheld, while in collectivistic cultures like Türkiye, the pursuit of personal achievement and self-development is not neglected. In both cultures, individuals strive for a blend of agentic and communal life goals, but the priority and significance of these goals shift based on cultural context. In Germany, people often prioritize agentic goals, shaping their worldview around

personal success and autonomy. Meanwhile, in Türkiye, people often prioritize communal goals, tending to act in accordance with societal norms and expectations.

1.7. Personal Differences: Self-Control

This thesis, to account for individual differences, explores the capacity for self-control as a moderating factor, specifically moderating the relationships between personal goals and work-family interference, as well as well-being, while the other moderators influence other specific relationships. Drawing from the Work-Home Resources model, which posits that certain personal characteristics serve as key resources enabling individuals to manage stressful situations better (ten Brummelhuis & Bakker, 2012), and management resources enable the selection, alteration, and application of other resources (Thoits, 1994), self-control capacity has been proposed as a pivotal 'key resource' within both the Conservation of Resources theory and the Work-Home Resources model. This capacity likely enhances functioning and behavioral regulation abilities, particularly when resource levels are high (Hagger, 2015).

Self-control is defined as the ability to monitor, modify, and regulate one's thoughts, emotions, and behaviors (de Ridder et al., 2012). This involves controlling impulses and habits in practice. Behaviors governed by self-control aim to secure the individual's long-term goals. People with high self-control exhibit superior performance, fewer problems with impulse control, and better psychological adjustment, including lower levels of depression and anxiety, higher self-acceptance or self-esteem, and enhanced interpersonal relationships, as evidenced by stronger family cohesion and reduced family conflict. Individuals with high self-control can exert it when needed and suspend it when not; namely, it is positively correlated with conscientiousness but less linked to perfectionism (Tangney et al., 2004). A meta-analysis shows that higher self-control correlates with improved behavior management, aligning with personal goals and societal norms, enhancing psychological well-being and life satisfaction. The study found a medium to strong effect of self-control on achievement and task performance

and a medium effect on interpersonal functioning, suggesting that individuals with greater selfcontrol are more successful in both work and social relationships (de Ridder et al., 2012). It has also been shown that self-control is necessary when shifting from one task to another, whether it be in a work or non-work setting (Baumeister et al., 2007).

Muraven and Baumeister (2000) conceptualized self-control as a crucial component of executive functioning with limited resources. It is like a muscle with limited capacity, which means that every act that requires self-control uses up this shared reserve of strength. Therefore, the effectiveness of self-control efforts depends on the individual's available self-control strength. Those who have greater reserves are more capable of achieving self-control objectives than those with lesser strength (Muraven & Baumeister, 2000). This fundamental understanding of self-control as a depletable resource sets the stage for examining how challenges, such as work-family interference, arise as individuals allocate this precious capacity towards achieving specific life goals.

A research study by Hofmann et al. (2014) underscores the importance of self-control in promoting affective well-being and life satisfaction through the effective management of multiple life goals. Individuals with higher self-control are better at managing resources between goals, leading to increased happiness and satisfaction. The concept of key resources explains how people utilize their resources. Individuals with more key resources may plan activities more efficiently, thus conserving time and energy for other pursuits. Notably, workhome interference is less probable among those with key resources, as these resources reduce the pressure of demands on personal resources like time and energy (ten Brummelhuis & Bakker, 2012). Similar to my perspective, Hirschi et al. (2019) discussed self-control within the framework of an action regulation model for balancing work and family, which views balance as successfully pursuing both work and family goals. It highlights self-control's significance in resource management, indicating that individuals with higher self-control are

better at using and optimizing these resources effectively. The article emphasizes the need for adaptability and flexibility, which are intricately linked to self-control. Maintaining workfamily balance involves regular progress monitoring, strategy adjustments, and decisionmaking to properly align and prioritize goals (Hirschi et al., 2019).

According to the strength model of self-control (Muraven & Baumeister, 2000), it might be possible to increase people's ability to exercise self-control over time. If self-control works like a muscle, then exercising it could make it stronger. Although initially, exerting self-control may deplete and weaken one's capacity, in the long term, it might have the opposite effect. Regular exercise of self-control followed by periods of rest and replenishment could gradually enhance an individual's overall capacity for self-control (Muraven & Baumeister, 2000). Metaanalysis studies have found that self-control can be enhanced through practice, exhibiting small to medium effects on various outcomes. It has been suggested that training in self-control can foster self-control across multiple domains, as well as general well-being, highlighting the significant potential of such interventions (Beames et al., 2017; Friese et al., 2017).

Berkman (2016) explained the fundamental theoretical frameworks detailing the improvement of self-control through training. He highlighted three primary models: the strength model, the motivational model, and the cognitive model. The strength model suggests that self-control across different domains relies on a finite, shared resource. Training is believed to strengthen this resource, leading to domain-general improvement. Research using the strength model has shown that any activity utilizing this shared self-control resource could serve as effective training (Berkman, 2016). Empirical studies often involve participants engaging in everyday tasks that require self-control, such as practicing speech regulation by avoiding slang (Gailliot et al., 2007), resisting sweets, or maintaining a handgrip (Muraven, 2010). These activities are followed by assessing their performance in a different, untrained self-control task (Berkman, 2016).

Motivational models concentrate on the willingness to self-control, proposing that aligning personal goals with one's self-concept, such as enhancing intrinsic motivation, can boost self-control (Berkman, 2016). Empirical evidence for these models demonstrates that interventions designed to increase autonomy in achieving goals (Williams et al., 2006), positively affirm their core values and self-identity (Schmeichel & Vohs, 2009), or contingent rewards (Prendergast et al., 2006), and monetary incentives (Volpp et al., 2008) can enhance goal-directed motivation and improve self-control. The key difference between the strength and motivational models is their focus: the strength model targets an individual's overall capacity for self-control, aiming to impact a wide range of targets, while motivational models focus on specific goals, targeting enhancements related to those goals (Berkman, 2016).

Cognitive models underscore the importance of cognitive processes, such as attention, beliefs, and habit learning, in self-control. Training interventions often involve cognitive strategies aimed at reinforcing goal-consistent behaviors (Berkman, 2016). For example, from a social-cognitive perspective, action planning and self-monitoring are identified as vital self-control skills (Schwarzer et al., 2011). Following the same logic, if-then statements can increase individuals' self-control abilities by creating plans to deal with potential obstacles on the way to the goal (Gollwitzer & Sheeran, 2006). Similarly, in the literature on work-family interference, some cognitive models aim to mitigate cross-domain interactions by enhancing self-control. For instance, individuals who engage in more planning behaviors are likely to be more proficient in exercising control to prevent work and family domains from conflicting (Lapierre & Allen, 2012). Furthermore, research has delved into how mindfulness can shape self-control processes, including decision-making regarding resource allocation and its connection to work-family interference (Kiburz et al., 2017). Developing mindfulness can serve as an effective self-regulatory practice, promoting better focus and attentional control, which, in turn, can positively alter perceptions of work-family balance (Allen & Paddock,

2015). Although not exclusively centered on self-control, the Selection, Optimization, and Compensation (Baltes & Heydens-Gahir 2003) framework is also fundamentally rooted in cognitive models, as these interventions are based on how individuals will select, optimize, and compensate their resources while pursuing their life goals (Müller et al., 2016).

Last but not least, self-control, while beneficial for achieving long-term goals, comes with significant costs. The exertion of self-control depletes mental and physical resources, temporarily weakening one's ability to make decisions and respond to challenges. Physiological effects include reduced glucose levels and compromised brain function, which can impair decision-making and social interactions (Baumeister & Alquist, 2009; Hofmann et al., 2014). Additionally, people with high self-control often face greater expectations from others, leading to heavier workloads and feelings of burden in both personal and professional relationships (Koval et al., 2015). People may also overestimate their capacity for self-control, particularly when planning for future tasks, which can result in unrealistic performance expectations and an increased likelihood of overwork (de Ridder et al., 2012; Del Líbano et al., 2012). In some cases, a combination of high self-control and low self-regulation fuels work cravings, contributing to psychological distress (Wojdylo et al., 2017). Beyond these social and cognitive demands, self-control may also come at an emotional cost. High levels of trait selfcontrol can limit one's emotional experience, leading to a life that is less rich in feelings and satisfaction (Layton & Muraven, 2014). Moreover, excessive self-control may lead to longterm regret, as individuals who consistently choose responsibility over pleasure might later feel they missed out on important experiences. For example, someone who works during a vacation may regret not fully enjoying the holiday years later (Kivetz & Keinan, 2006). Finally, selfcontrol's benefits vary across individuals. Individuals who prioritize rationality tend to feel true to themselves when they exercise self-control. In contrast, those who rely more on emotions often feel more satisfied and authentic when they give in to their desires (Kokkoris et al., 2019).

1.8. Aim of the Dissertation

1.8.1. Research Objectives

Building on the theoretical framework and literature presented in the previous section, this thesis seeks to understand the relationship among life goals, work-family interference, and well-being. An extended discussion has been provided on how individuals' life goals can influence the way they balance work and family responsibilities and how this balancing act affects their well-being.

Bolger and Zuckerman (1995) explored how individual-related factors can shape wellbeing, focusing on two key concepts: stressor exposure and stressor reactivity. Stressor exposure refers to how likely individuals are to perceive a situation as stressful, while stressor reactivity refers to their likelihood of reacting emotionally or physically to stress. They emphasized the importance of examining both how individuals encounter stress and how they respond to it in order to understand the role of dispositional factors in stress reactions. Viewing work-family interference as a source of stress underscores the extent to which individuals experience conflict between their work and family roles, contributing to their overall stress levels. The way individuals respond emotionally or physically to this stress, in turn, plays a critical role in determining their psychological well-being (Bolger & Zuckerman, 1995).

While prior research has shed light on work-family interference, there remains a notable gap in understanding the role of dispositional/motivational factors, particularly life goals, in shaping these dynamics. Despite calls from scholars (Carlson & Kacmar, 2000; Hauser et al., 2018; Senecal et al., 2001) to explore the association between life goals and work-family interference, much of the literature continues to focus on external, situational factors (Allen et al., 2020, 2023). Although some studies have examined the link between life goals and well-being and consider content, significance, and achievement of life goals (Macleod, 2012; Saragovi et al., 2002), they often overlook the ways in which these life goals impact multiple

domains of life. Inconsistencies in the relationship between work-family interference and wellbeing necessitate a deeper examination of how personal strivings, as well as inter-cultural and personal differences, can modify this connection. This implies that, depending on inter-cultural and personal differences, work-family interference may be experienced more or less frequently and have varying degrees of negative impact on an individual's well-being (Kinnunen et al., 2013). This thesis addresses this gap by investigating the relationship among life goals, workfamily interference, and well-being, offering new insights into the complex interplay between life goals and the balance between work and family roles.

This thesis has four objectives, which are offering a new perspective on the role of dispositional/motivational factors in shaping work-family interference, evaluating the influence of culture, not only in terms of gender role expectations but also in the pursuit of non-traditional life goals, investigating the role of personal differences in managing work-family interference while pursuing life goals, and testing how self-regulation interventions manage work-family interference. To research these research objectives, the following research questions will be answered through a hypothesis-driven approach:

- I. How does work-family interference mediate the relationship between life goals and wellbeing, and in what ways do life goals influence work-family interference and their subsequent impact on well-being?
- II. How do different gender roles in different cultural contexts (i.e., Turkish and German cultures) modify the relationship between the pursuit of different life goals, work-family interference, and overall well-being?
- III. How do varying levels of self-control modify the relationship between the pursuit of different life goals, work-family interference, and overall well-being?
- IV. What role do self-regulation interventions play in managing work-family interference?

1.8.2. Hypotheses and Methodological Approaches

In order to answer the research questions mentioned above, this thesis comprises three empirical studies. The hypotheses presented here are grounded in the theoretical frameworks previously discussed, and the full derivation and justification for each hypothesis will be elaborated upon in later chapters.

Chapter 2, "Study 1: Work-Family Interference and Life Satisfaction: The Role of Life Goals," explores how life goals and gender roles influence the relationship between WtFI and FtWI on life satisfaction. It addresses the first research question using quantitative data from Waves 8 and 10 of the German Family Panel Survey (pairfam). This study examines why individuals experience the effects of WtFI and FtWI differently, with some showing resilience while others succumb to negative impacts. Drawing on the Conservation of Resources theory (Hobfoll, 1989), which argues that resource loss directly diminishes well-being, and the Work-Home Resources model (ten Brummelhuis & Bakker, 2012), which posits that resource depletion from one life domain (work or family) impacts the other, it is expected that both WtFI and FtWI will reduce life satisfaction. These theories suggest that individuals allocate resources selectively, with those pursuing agentic goals investing more in the work domain and those with communal goals prioritizing the family domain. In line with these models, individuals who invest more heavily in either work or family roles may become more susceptible to interference originating from the domain in which they are most invested, leading to the hypothesis that individuals with agentic goals will experience a greater negative impact on life satisfaction from WtFI, while those with communal goals will be more affected by FtWI. Furthermore, based on gender role theories (Bakan, 1966; Eagly, 2020), deviations from traditional gender expectations—such as women pursuing agentic goals or men pursuing communal goals-are expected to create additional stress and conflict. As a result, individuals who deviate from these societal expectations are likely to experience heightened negative effects from work-family interference due to the "double trouble" of conflicting social norms and work-family interference. This study positions life goals as moderators that can alter the intensity or direction of the effect of work-family interference on well-being, reflected through life satisfaction.

Chapter 3, "Study 2: You are Free to Choose Your Struggle: Exploring the Mediating Role of Work-Family Interference in the Dynamics of Life Goals and Well-Being," explores the question of how agentic and communal life goals affect well-being through the lens of work-family interference, within the diverse cultural settings of Germany and Turkey. This study also investigates how culture and individual trait self-control moderate these effects, responding to the first three research questions with primary self-report quantitative data from white-collar employees in both countries. Following the Conservation of Resources theory (Hobfoll, 1989) and the Work-Home Resources model (ten Brummelhuis & Bakker, 2012), pursuing life goals requires the investment of personal resources, which increases the likelihood of resource depletion and interference between work and family roles. Therefore, it is assumed that pursuing life goals, whether agentic or communal, requires significant personal resource investment. This resource investment can lead to depletion, which increases the likelihood of work-family interference. While life goals provide individuals with a sense of purpose and fulfillment, they may also indirectly harm well-being by amplifying work-family interference due to the strain on resources across work and family domains. In line with research on cultural contexts (Oishi & Diener, 2001), the assumption is that individualistic cultures emphasize agentic goals, while collectivistic cultures emphasize communal goals, suggesting that cultural context significantly shapes the relationship among life goals, workfamily interference, and well-being. These relationships are particularly pronounced when gender roles conflict with societal expectations (Eagly, 2020) due to the "double trouble" of conflicting social norms and work-family interference. Third, building research on selfregulation, it is assumed that self-control acts as a buffer in managing work-family interference. Individuals with higher self-control are better equipped to manage conflicting demands from work and family, allocating their resources more effectively (de Ridder et al., 2012; Tangney et al., 2004). As a result, they are expected to experience less interference and maintain higher levels of well-being compared to individuals with lower self-control, who may struggle more with balancing these competing demands. This study adds to the previous one and extends the scope by positioning life goals as a predictor and introducing cultural context and individual self-control as additional moderating factors.

Chapter 4, "Study 3: Examining the Impact of a Self-Regulation Intervention on Workto-Family Interference", aims to improve self-control skills through an intervention intending to mitigate work-family interference and boost well-being, answering the fourth research question, using the longitudinal self-report survey data collected from the same group of German white-collar employees from the second study. A randomized controlled trial with a two-group pre-post design is being implemented. Drawing on the strength model of self-control (Muraven & Baumeister, 2000), which posits that self-control can be developed and strengthened through practice, this study assumes that a self-regulation intervention incorporating action planning and action control strategies will reduce WtFI. By enhancing self-control, employees are expected to improve their ability to manage work demands and set clear boundaries between work and family life, thereby decreasing time-based and strain-based WtFI. It is further assumed that individuals who improve their self-control through the intervention will experience a more significant reduction in WtFI compared to those whose self-control remains stable or declines. This study adds a practical application to the theoretical frameworks developed in previous studies by testing the effectiveness of the self-regulation intervention in real-world settings.

Chapter 2: Work-Family Interference and Life Satisfaction: The Role of Life Goals

This chapter is based on the article: Ipek, G., Drobnič, S., & Boehnke, K. (2024). Workfamily interference and life satisfaction: the role of life goals. *Community, Work & Family*, 1– 23. https://doi.org/10.1080/13668803.2024.2424358

Abstract

The interface between work and family domains has received considerable scholarly attention in recent decades due to its substantial impact on outcomes related to work, family, health, and well-being. However, individuals experience this impact differently, with some demonstrating resilience while others are more vulnerable to negative effects. In this study, we investigate the role of individuals' life goals and gender in the work-family interface, using the framework of the work-home resources model and the self-regulation theory. Specifically, we examine how life goals and gender moderate the impact of work-to-family interference (WtFI) and familyto-work interference (FtWI) on life satisfaction. We analyze these associations for men and women using Structural Equation Modeling and data from Wave 8 and Wave 10 of the German Family Panel Survey (*pairfam*). Our findings show that life goals indeed significantly moderate these relationships. Individuals prioritizing agentic goals feel the detrimental impact of WtFI on life satisfaction more intensely than those who prioritize communal goals and those whose goals are balanced. Similarly, individuals with high communal goals are more affected by the adverse effects of FtWI on their life satisfaction than those who prioritize agentic goals and those whose goals are balanced. We observed no significant gender difference in the impact of WtFI on life satisfaction for women and men who primarily pursue agentic life goals, nor in the effect of FtWI on life satisfaction between those who primarily pursue communal goals. We also discuss potential theoretical and practical implications in light of our findings.

Keywords

Life goals; agency; communion; work-family interference; life satisfaction

2.1. Introduction

Recent decades have witnessed significant social and demographic changes, particularly in the work environment and family dynamics. These changes, characterized by increased work demands and evolving family structures, have made balancing work and family life a significant challenge. The growing difficulty in navigating these dual responsibilities has sparked interest in the work-family interface. This field delves into the interplay between work and family roles and their impact on life satisfaction. Within this framework, life goals serve as a lens through which individuals perceive and navigate the challenges posed by the workfamily interface. Building upon the proposition put forth by Carlson and Kacmar (2000), variations in work-family interference can be attributed to individuals' distinct set of life goals. However, a notable gap persists in the existing literature concerning the specific ways life goals modulate the effects of work-family interference on life satisfaction.

The interaction between work and family domains can be comprehensively examined using the Work-Home Resources model (ten Brummelhuis & Bakker, 2012), which acknowledges both positive (enhancement) and negative (conflict) aspects. Although engaging in multiple roles can yield benefits and resource gains (Greenhaus & Powell, 2006; Marks, 1977), it also presents challenges when resources allocated to one domain diminish the availability for another (Goode, 1960; Greenhaus & Beutell, 1985). Our research primarily focuses on the conflict perspective, guided by the understanding that negative experiences in this interplay often impact life satisfaction substantially more than positive ones (Baumeister et al., 2001; Drobnič et al., 2010). Numerous studies have demonstrated that work-family interference, characterized by resource loss, induces stress and adversely affects life satisfaction (Hobfoll et al., 2018). This notion is supported by a recent meta-analysis, which found that work-family interference has a negative impact on overall well-being (Biswas et al., 2022).

While an extensive body of research (see Amstad et al., 2011, and Mullen et al., 2008 for an overview) provides insights into the effects of work-family interference on life satisfaction, it indicates that individuals experience these impacts to varying degrees. This has led researchers to examine various factors, including work-related resources like job control (Mauno et al., 2006), family-related resources such as spousal support (Aycan & Eskin, 2005), and individual characteristics such as personality (Kinnunen et al., 2003). The role of dispositional characteristics, particularly life goals, in influencing the experience and impact of work-family interference on life satisfaction is an area of growing interest (Hirschi et al., 2019). Life goals, whether focused on independence (agentic) or interdependence (communal) (Bakan, 1966), can shape how individuals allocate their resources and perceive stressors from different domains (Grandey & Cropanzano, 1999). We argue that goal strivings in different life domains may threaten self-regulatory personal resources. Therefore, in this study, we aim to explore the moderating role of life goals in the relationship between work-family interference and life satisfaction.

Set within the context of the gendered nature of life experiences, we examine the intricate relationship between life goals and gender, focusing on their combined impact on work-family interference and life satisfaction. We explore how societal expectations for gender roles—typically communal characteristics for women and agentic characteristics for men (Eagly et al., 2020)—interact with personal life goals. This interplay within the broader societal context is crucial as it can amplify or mitigate the strain caused by work-family interference stressors. Individuals with life goals that align or conflict with societal gender expectations might experience work-family interference differently, influencing their life satisfaction.

In this study, grounded in the Work-Home Resources model (ten Brummelhuis & Bakker, 2012), which can be integrated with the self-regulation theory (Baumeister & Vohs, 2007), we focus on managing resources between work and family life and their impact on life

satisfaction. We emphasize the importance of understanding the longitudinal effects of workfamily interference on life satisfaction over two years. By concentrating on individual characteristics, particularly life goals, and gender, our study brings a new perspective. It aims to clarify which individuals find these interferences more unsettling and why.

2.2. Theoretical Framework

2.2.1. Work-Family Interference and Life Satisfaction

Work-family interference is characterized by conflicting pressures between work and family roles, resulting in incompatibilities related to time, strain, and behavior (Greenhaus & Beutell, 1985). This interference can flow from work to family (WtFI) or from family to work (FtWI)⁴, forming a cyclical process (Frone et al., 1992) where work demands impact family obligations and family needs affect work responsibilities (Netemeyer et al., 1996). The cyclical nature of this process has been substantiated by empirical evidence (Beham & Drobnič, 2011).

The interference between work and family domains often leads to various adverse outcomes across work, family (i.e., domain-specific), and health (i.e., domain-unspecific), as indicated by multiple meta-analyses (Allen et al., 2020; Amstad et al., 2011). The relationship between work-family interference and its domain-specific effects remains a considerable scholarly debate. The traditional cross-domain perspective (Frone et al., 1992; 1997) suggests that conflict from one domain predominantly causes issues in the opposite domain (e.g., WtFI influencing family-related outcomes). However, the more recent matching hypothesis (Amstad et al., 2011; Shockley & Singla, 2011) argues that conflict from one domain primarily affects the domain where it originates (e.g., WtFI influencing work-related outcomes). Nonetheless, a

⁴ In this paper, the term "work-family interference" is utilized to encompass the two-way interrelationships between work and family domains, while "work-to-family interference" (WtFI) and "family-to-work interference" (FtWI) refer to one-way processes.

meta-analysis shows the most significant relationship between work-family interference and domain-unspecific outcomes, such as life satisfaction (Amstad et al., 2011).

Life satisfaction is the cognitive aspect of the composite well-being measure (Riva et al., 2019), encompassing a person's evaluation of their life against personal standards (Diener et al., 1985). It is considered the sum of domain-specific (e.g., job, marital) satisfactions (Stubbe et al., 2005). Changes in these domains and various life events may cause a change in overall levels of life satisfaction (Pavot & Diener, 2008). Focusing on life satisfaction instead of narrower, domain-specific well-being indicators helps capture the long-term effects of work-family interference across various life domains.

The relationship between work-family interference and life satisfaction has mainly been studied within the framework of the Conservation of Resources theory (Hobfoll, 1989), which forms the foundational theory of the Work-Home Resources model (ten Brummelhuis & Bakker, 2012). According to the Conservation of Resources theory, individuals strive to attain, protect, and accumulate personal resources such as time and energy. If they fail in their conservation efforts, their resources begin to deplete. Using the loss spiral concept, Hobfoll (1989; 2011) argues that initiating a cycle of resource depletion induces stress, and ongoing stress reduces well-being. The conflict between work and family domains manifests as a situation where requirements in one domain deplete personal resources, leaving insufficient resources to fulfill the needs of the other domain (Grandey & Cropanzano, 1999). The persistent demands in one domain require ongoing personal resource investment, leading to negative long-term outcomes (ten Brummelhuis & Bakker, 2012). Accordingly, in a longitudinal study, Knecht and colleagues (2011) found that people who experienced more work-family interference is a longitudinal predictor of life satisfaction and mental health.

Starting from these premises, in this study, we propose that work-to-family interference (WtFI) and family-to-work interference (FtWI) experienced at Time 1 (T1) will significantly predict life satisfaction two years later at Time 2 (T2), even when life satisfaction at T1 is controlled.

Hypothesis 1: WtFI and FtWI at T1 predict life satisfaction at T2.

2.2.2. The Role of Life Goals

According to the Work-Home Resources model (ten Brummelhuis & Bakker, 2012), demands from different domains draw on individuals' resources, and a reduction in these resources can lead to interference between domains. However, individual differences can either help a person cope with this resource loss or make them more vulnerable to its effects. In this study, we examine how individual differences in goal endorsement affect the impact of between-domain interference on life satisfaction. While work-family interference represents the current state, life goals represent the desirable state. Therefore, to evaluate the alignment between the current/mandatory situational characteristic and the individual's preferences, life goals are considered as a moderator in the relationship between work-family interference and life satisfaction.

Humans are inherently goal-oriented, driven by the pursuit of desired states they seek to obtain, maintain, or avoid (Emmons, 1996). Bakan (1966) conceptualizes human duality in terms of (i) the agency construct and (ii) the communion construct, representing self-relevant goals. Agency focuses on self and independence, while communion emphasizes others and interdependency (Helgeson, 1994). Agentic goals include achievement, power, variation, assertiveness, and competence, while communal goals comprise intimacy, affiliation, altruism, warmth, and morality (Abele & Wojciszke, 2014; Pöhlmann, 2001).

In the context of the Work-Home Resources model, *resources* play a central role in goal attainment (Hirschi et al., 2019). Resources are broadly categorized into two types: contextual resources, external to the self and located within one's social environment, and personal resources, which are personal characteristics and energies (Hobfoll, 2011). Key resources, such as optimism and goal-pursuit intensity, are stable personal characteristics crucial for resource management (ten Brummelhuis & Bakker, 2012). Self-regulation, considered a key resource, is depicted as the ability to control emotions, thoughts, and behaviors in pursuit of life goals (Zimmerman, 2000). The self-regulation theory posits that self-regulatory resources are as limited as other personal resources, such as time and energy (Baumeister & Vohs, 2007). Thus, individuals must allocate these resources wisely, especially when facing conflicting demands from work and family domains.

Self-regulation models often highlight the importance of focusing on selected goals to conserve limited resources (Fishbach et al., 2009). Individuals allocate (or invest) resources based on the nature of their goals (Grawitch et al., 2010). This aligns with Hirschi et al.'s (2019) action regulation model, which posits that balancing domains requires effective resource allocation across life domains and making compromises when necessary. Achieving a balance across different life areas fosters a sense of fulfillment (Greenhaus & Powell, 2017). However, when individuals heavily invest in domains that align with their life goals and still encounter ongoing imbalances, they face challenges. People expect positive outcomes when they allocate more resources to a specific domain. However, if that domain remains a source of conflict and occupies a blamed position, it can be particularly distressing for individuals who attribute particular importance to it. This may lead to a more pronounced decline in their life satisfaction.

In our theoretical framework, self-regulation is proposed as an underlying mechanism that explains how life goals moderate the relationship between work-family interference and life satisfaction. Individuals with specific life goals utilize their self-regulation capabilities to allocate resources toward achieving these goals. We adopt the perspective that agentic goals, more related to the work domain, may lead to a focus on work-related activities. In contrast, communal goals, which are more related to the family domain, might result in greater allocation towards family endeavors. These individuals use their self-regulatory resources to make effective decisions in allocating resources between work and family domains while pursuing their agentic or communal life goals. Rothbard and Edwards (2003) support the idea that the importance attributed to work and family domains is positively associated with the time and energy allocated to those domains.

Furthermore, we argue that investing resources in agentic or communal life goals can render individuals more susceptible to stress in their prioritized domain. For those who place high importance on a specific life goal, whether agentic or communal, role pressures become particularly significant because success and rewards in that domain are crucial to them (Greenhaus & Beutell, 1985). As a result, they strive to manage these pressures by intensifying their efforts and investments in work or family. Li et al. (2020) emphasize that such investment is not without risks; greater investment can result in more substantial losses in the face of stress, particularly when it arises from a highly invested domain. This situation contributes to a decreased sense of control over that area, compounding the impact on well-being. Consequently, individuals in this scenario often grapple with the frustration of not receiving a proportional return on their investment, further intensifying the sense of dissatisfaction. We propose that when work interferes with family, individuals pursuing agentic life goals (investing more in the work domain) experience a sense of loss of control. Conversely, when family interferes with work, individuals pursuing communal life goals (investing more in the family domain) also experience a sense of loss of control, which aligns with self-regulation theory (Baumeister & Heatherton, 1996).

We, therefore, suggest life goals may moderate the relationship between WtFI/FtWI and life satisfaction in a direction different from the sensitization perspective (i.e., endorsement of agentic life goals should exacerbate the negative effects of FtWI on life satisfaction, and vice versa for endorsement of communal life goals; Carr et al., 2008). We follow the Work-Home Resources model by focusing on the amount of resources people invest in the work and family domains. Specifically, WtFI negatively impacts the life satisfaction of those with agentic life goals more than those with communal or balanced life goals. This is because they are more dissatisfied when their significant investment in work leaves them with insufficient resources for their family, leading to a sense of losing control over their work priorities. Similarly, FtWI negatively impacts the life goals. They are more dissatisfied when their significant investment in the family leaves them with insufficient resources for work, leading to a sense of losing control over their significant investment in the family leaves them with insufficient resources for work, leading to a sense of losing control over their significant investment in the family leaves them with insufficient resources for work, leading to a sense of losing control over their significant investment in the family leaves them with insufficient resources for work, leading to a sense of losing control over their significant investment in the family leaves them with insufficient resources for work, leading to a sense of losing control over their family priorities.

Hypothesis 2a: Individuals who primarily pursue agentic life goals are likely to experience more substantial negative effects of WtFI on life satisfaction than those who primarily pursue communal or balanced life goals.

Hypothesis 2b: Individuals who primarily pursue communal life goals are likely to experience more substantial negative effects of FtWI on life satisfaction than those who primarily pursue agentic or balanced life goals.

2.2.3. The Role of Gender

Researchers have noted that agency and communion exist in both men and women. However, women are expected to be more communal (reflecting a desire to be with others), and men are expected to be more agentic (reflecting a desire to master) (Bakan, 1966; Eagly, 2020). Stereotypical societal expectations influence people's actions, leading them to regulate their behavior in alignment with their gender identity (Sczesny et al., 2018). When individuals cannot regulate their actions in a way that aligns with societal expectations, negative affect is commonly experienced (Haines & Stroessner, 2019; Rudman et al., 2012).

The Work-Home Resources model provides insights into how cultural norms on gender roles may impact men and women differently (ten Brummelhuis & Bakker, 2012). This model facilitates the exploration of gender differences in accessing contextual and personal resources, emphasizing the importance of macro resources, such as societal and cultural systems. These macro resources are positioned at a higher level because they are more stable and beyond individual control (ten Brummelhuis & Bakker, 2012). We consider societal expectations as one of the macro resources and argue that the negative effect of work-family interference on life satisfaction will differ between men and women who align or deviate from societal expectations. Individuals with gender-atypical goals might already experience strain because their goals deviate from societal expectations. When faced with another source of strain, like work-family interference, they encounter double pressures.

Regarding interference between different domains, previous studies have yielded inconsistent findings on how men and women are affected by work-family interference. Wang and colleagues (2012) found that major depressive disorders were associated with FtWI for men but with WtFI for women. However, Frone and colleagues (1996) found that depression levels in men are more negatively affected by WtFI than in women. Yucel and Fan (2019) found no gender difference in the relationship between work-family interference (both WtFI and FtWI) and well-being. We argue that these inconsistencies stem from studies focusing solely on gender, overlooking the crucial impact of life goals on individuals' experiences with work-family interference.

This research proposes that women with agentic goals and men with communal goals deviating from societal norms may face more significant challenges in work-family interference. Based on our initial assumptions, we contend that the life satisfaction of individuals with agentic goals will be more affected by WtFI, while FtWI will have a greater impact on those with communal goals due to their sensitivity to conflict in domains where they have invested personal resources. However, we also argue that women primarily pursuing agentic goals not endorsed by societal norms will experience lower life satisfaction levels than men with similar life goals when experiencing WtFI. Similarly, men primarily pursuing communal goals, contrary to societal expectations, will have lower life satisfaction levels than women with communal goals while experiencing FtWI. We interpret this as individuals showing increased sensitivity to the life domain where they invest their resources (as guided by their life goals), heightened when social expectations contradict their personal goals, leading to additional stress or burden. Given the above, we propose that life goals and gender interact with each other and *together* moderate the relationship between work-family interference (both WtFI and FtWI) and life satisfaction.

Hypothesis 3a: Women who primarily pursue agentic life goals are likely to experience more substantial negative effects of WtFI on life satisfaction than men who primarily pursue agentic life goals.

Hypothesis 3b: Men who primarily pursue communal life goals are likely to experience more substantial negative effects of FtWI on life satisfaction than women who primarily pursue communal life goals.

The theoretical model in Figure 2. suggests that life goals and gender can moderate the relationship between work-family interference (both WtFI and FtWI) and life satisfaction.

Figure 2.

Research Model



2.3. Method

2.3.1. Data and Sample

The data for this study came from the German Family Panel (*pairfam*: "Panel Analysis of Intimate Relationships and Family Dynamics") (Brüderl et al., 2021). The survey began in 2008 with a nationwide random probability sample of 12,402 respondents selected from the population registers for three birth cohorts (1971–1973, 1981–1983, and 1991–1993). This was supplemented with a refreshment sample and a new birth cohort (2001–2003) in 2018. The primary mode of data collection was a computer-assisted personal interview (CAPI). Besides the annual core questionnaire, other questions were asked bi-annually or less frequently. We used only *pairfam*'s Wave 8 (2015/2016) and Wave 10 (2017/2018) because participants answered questions about work-family interference, life satisfaction, and essential goals and domains in life in the same survey year. Wave 8 included 5,461 respondents; Wave 10 included 4,750 respondents. Consistent with our research focus, we excluded non-employed individuals but included self-employed individuals or those with full-time or part-time work. We also excluded individuals whose marital status changed between Wave 8 and Wave 10 (n = 176).
This was done because we could not rule out the possibility that their level of life satisfaction was strongly affected by variables not included in Figure 2. or even assessed in Wave 8 and Wave 10. Participants with missing data for any of the variables in Figure 2. were also excluded from the analysis, as were respondents who did not participate in both data-gathering waves. This resulted in a final sample size of 2,656 participants.

According to descriptive statistics in Wave 8, 1361 (51.2%) of the 2656 participants in the analytical sample were men, and 1295 (48.8%) were women. The age range was 21–45 years (Mean (M) = 36.33; Standard Deviation (SD) = 7.40). There were 1052 (39.6%) respondents with a university degree or higher. A total of 1389 (49%) respondents were married or lived in a so-called civil union (a legal arrangement available to same-sex couples during Wave 8 and Wave 10). A total of 1,097 (38.7%) respondents were single, 161 (5.7%) were divorced (or in dissolved civil union), and 9 (0.3%) were widowed (or the surviving partner from a civil union). A total of 1,763 (66.4%) respondents were employed full-time, 642 (24.2%) were employed part-time, and 251 (9.5%) were self-employed. The average weekly working time was 38.6 hours (SD = 11.02). Only 49 (1.8%) respondents stated that they always worked from home. Most study participants had at least one child (1,592, 59.9%). A total of 1,279 (48.2%) participants stated that they had children younger than 14 living at home.

2.3.2. Measures

Dependent variable. In this study, general satisfaction with life is the measure of wellbeing. Life satisfaction is a one-item variable based on the question, "How satisfied are you with your life at the moment?" Participants responded on an 11-point Likert-type scale (0 = very dissatisfied to 10 = very satisfied). This measurement was used for both time points (Wave 8 and Wave 10). Specifically, the dependent variable was life satisfaction in Wave 10 (Time 2). *Independent variables.* The interference between work and family domains was measured with items that allowed us to assess the bidirectional inter-role conflicts of concern: work-to-family interference and family-to-work interference concerning time and strain (Carlson & Grzywacz, 2008). Our instrument contained eight items, four measuring WtFI and four measuring FtWI. Respondents were asked how their family life and work life influence one another and the extent to which the statements in the questionnaire describe their situation. Response options were presented as a 5-point Likert-type scale (1 = not at all to 5 = absolutely). An example item for WtFI was, "My work prevents me from doing things with my friends, partner, and family more than I would like." An example item for FtWI was, "Because I am often stressed in my private life, I have problems concentrating on my work." Cronbach's alpha was .765 for WtFI and .720 for FtWI. WtFI and FtWI were taken from Wave 8 (Time 1) and calculated by adding up the scores of the four domain-specific items. Due to a potential conceptual overlap between both variables, we conducted confirmatory factor analyses prior to hypothesis testing (results available on request). These confirmed that WtFI and FtWI are two different constructs.

Moderator variables. We used life goals and gender as moderator variables and took data from Wave 8 (Time 1). Huinink and Schröder (2003) developed the items for assessing life goals in their project 'Das Timing der Familiengründung' [The timing of family formation]. Respondents were asked to consider five life goals: (1) pursuing education or career interests, (2) pursuing hobbies and interests, (3) keeping in touch with friends, (4) living in a partnership, and (5) having a(nother) child. Respondents were asked to assess each goal's importance and indicate their relative importance by distributing 15 points to five goals. The more important a goal was, the more points they were asked to assign. Goals 1 and 2 were considered to be representative of agentic goals, while Goals 3, 4, and 5 were considered to be representative of communal goals. We calculated the sum of points a respondent gave to agentic and communal goals. We then determined the sample's median values for agentic and communal goals. Respondents were considered to have high agentic goals if their sum of points for agentic goals was at or above the sample median ($Mdn_{AG} = 7.00$). Respondents were considered to have low agentic goals if their sum of points for agentic goals was below the sample median. The procedure was repeated for communal goals ($Mdn_{CG} = 8.00$). Finally, we created three grouping variables, distinguishing between respondents with (1) low agentic–high communal goals, (2) high agentic–low communal goals, and (3) high agentic–high communal goals. Respondents could not have scores below the median for both types of goals.

Gender was assessed in the classic binary form (0 = man, 1 = woman). As mentioned, we expected that the effect of WtFI and FtWI on life satisfaction would vary between men and women. Based on individuals' life goals, we created six interaction variables by multiplying life goals with gender. We could then distinguish between (1) men with low agentic–high communal goals, (2) women with low agentic–high communal goals, (3) men with high agentic–low communal goals, (4) women with high agentic–low communal goals, (5) men with high agentic–high communal goals, and (6) women with high agentic–high communal goals.

Control variables. To control the baseline life satisfaction level, we included a respondent's life satisfaction level at Time 1 (Wave 8). We also used employment status (1 = regular full-time employment, 0 = other), marital status (1 = married or in a civil union, 0 = other), and having children (1 = yes, 0 = no children) as control variables. The data for all control variables were taken from Wave 8 (Time 1). We included employment status, marital status, and children because these factors can confound the relationship between work-family interference and life satisfaction. Some research has shown that married people are more satisfied with their lives than those who are divorced or have never married (Haring-Hidore et al., 1985; Helliwell, 2003). In addition, long working hours and having a child at home increase

work and family demands. This can lead to higher levels of work-family interference and negatively affect well-being (Byron, 2005; Kinnunen & Mauno, 1998).

2.3.3. Analytic Strategy

After documenting our descriptive findings, we conducted a Pearson correlation analysis to examine the relationships between the key variables (WtFI, FtWI, life goals, gender, and life satisfaction) in the study. Next, we used structural equation modeling to test and assess the effects of work-family interference (both WtFI and FtWI) on life satisfaction. Finally, we performed a multiple-group comparison analysis to examine whether the effect on life satisfaction differed for the six life goals by gender groups. We included all control variables in these analyses using SPSS and AMOS 28.

2.4. Results

Descriptive results for the overall sample and the various life-goal groups for men and women are presented in Table A2.1 in the Appendix. In both time points, women showed slightly higher life satisfaction (M = 7.66, 7.68) than men (M = 7.64, 7.61). Among the life-goal groups, the highest life satisfaction was observed in those with low agentic-high communal goals (M = 7.74, 7.74). WtFI was much more prevalent (M = 9.53) than FtWI (M = 6.51), with men reporting more interference in both directions (M = 9.82, 6.57) than women (M = 9.22, 6.46). We also analyzed correlations between WtFI, FtWI, life satisfaction (T2), and our proposed moderator variables (Table A2.2 in the Appendix). The correlation matrix revealed a negative relationship of life satisfaction (T2) with WtFI and FtWI (r = -.19, $p \le .001$; r = -.16, $p \le .001$), yet no significant correlation was found between different life goal combinations and interference types. Gender correlated with WtFI, indicating more interference experienced by men (r = -.08, $p \le .001$), but not with FtWI. The relationships between key variables like WtFI, FtWI, life goals, gender, and life satisfaction, in general, exhibit small to medium effect sizes (as per Cohen 1988).

In the next step, we examined the direct effects of WtFI and FtWI on life satisfaction by incorporating all control variables into the model (Table 2.1). The results confirmed that the negative effects of WtFI and FtWI on life satisfaction (T2) were statistically significant ($\beta = -0.06$, $p \le 0.001$; $\beta = -0.04$, $p \le 0.05$, respectively). The overall model explained 23% of the variance (\mathbb{R}^2) in life satisfaction. These findings suggest that interference between work and family domains is associated with decreased life satisfaction, even when controlling for baseline life satisfaction levels. Thus, these results support Hypothesis 1.

Table 2.1

Path	β	р
WtFI \rightarrow Life Satisfaction (T2)	064	.001
$FtWI \rightarrow Life Satisfaction (T2)$	040	.029
Life Satisfaction (T1) \rightarrow Life Satisfaction (T2)	.446	.001
Having a Child \rightarrow Life Satisfaction (T2)	040	.062
Working Full-Time \rightarrow Life Satisfaction (T2)	009	.602
Being Married \rightarrow Life Satisfaction (T2)	.044	.036
R^2	.233	

Path Analysis of Primary Relationships

Note. WtFI: Work-to-family interference, FtWI: Family-to-work interference.

In the subsequent analysis, we explored whether prioritizing specific life goals moderates the relationship between WtFI as well as FtWI and life satisfaction while controlling for life satisfaction in T1, employment type, marital status, and children. Multi-group comparisons were conducted to assess the significance of group differences (Table 2.2). The slope was constrained equally among the groups compared to the unconstrained model. Specifically, we compared two models: one in which the slopes of the relationships between WtFI as well as FtWI and life satisfaction were constrained to be equal across life goals groups (constrained model), and another in which these slopes were allowed to vary freely (unconstrained model). The statistical differences between the models were determined using the chi-square difference test. Results for WtFI revealed significant differences in the effects of WtFI on life satisfaction among the three life goals groups ($\chi^2 = 13.863$; df = 2; $p \le 0.001$) between individuals with low agentic–high communal life goals and those with high agentic–low communal life goals ($\chi^2 = 11.107$; df = 1; $p \le 0.001$), as well as between individuals with high agentic–low communal life goals and those with high agentic–low communal life goals ($\chi^2 = .220$; df = 1; p = 0.64) were not significant. Results for FtWI indicated that only individuals with low agentic–high communal life goals and those with high agentic–low communal life goals and those with high agentic–low communal life goals and those with low agentic–high communal life goals ($\chi^2 = .220$; df = 1; p = 0.64) were not significant. Results for FtWI indicated that only individuals with low agentic–high communal life goals and those with high agentic–low communal life goals marginally differed ($\chi^2 = 2.887$; df = 1; p = 0.09) in terms of the relationship between FtWI and life satisfaction.

Table 2.2

Nested Model Comparisons in The Relationship Between Work-Family Interference and Life Satisfaction among Individuals with Different Life Goals

WtFI \rightarrow Life S	Satisfaction (T	2)	χ^2	DF	р
Life Goals					
LA_HC	HA_LC	HA_HC	13.863	2	.001
	LA_HC	HA_LC	11.107	1	.001
	LA_HC	HA_HC	.220	1	.639
	HA_LC	HA_HC	10.810	1	.001
$FtWI \rightarrow Life Sa$	atisfaction (T2)	χ^2	DF	р
FtWI → Life Sa Life Goals	ntisfaction (T2)	χ ²	DF	р
FtWI → Life Sa Life Goals LA_HC	ntisfaction (T2) HA_HC	χ ² 3.003	DF 2	р .223
FtWI → Life Sa Life Goals LA_HC	HA_LC LA_HC) HA_HC HA_LC	χ ² 3.003 2.887	DF 2 1	р .223 .089
FtWI → Life Sa Life Goals LA_HC	HA_LC LA_HC LA_HC) HA_HC HA_LC HA_HC	χ ² 3.003 2.887 .725	DF 2 1 1	р .223 .089 .395

Note. LA_HC: Low agentic-high communal goal endorsement, HA_LC: High agentic-low communal goal endorsement, HA_HC: High agentic-high communal goal endorsement. When assuming the unconstrained model to be correct, the results from a multiple-group comparison analysis indicate whether the relationship between work-family interference and life satisfaction differs significantly among individuals with different life goals. The rows represent pairwise comparisons in the relationship between work-family interference and life satisfaction among different life goal groups.

The impact of work-family interference on life satisfaction for different life-goals groups is presented in Table 2.3. The results for WtFI indicated a negative, highly significant impact

of WtFI on life satisfaction for individuals with high agentic–low communal life goals ($\beta = -0.18$, $p \le 0.001$). In contrast, the effect of WtFI on life satisfaction was insignificant for those with low agentic–high communal life goals ($\beta = -0.03$, p = 0.22) and for those with high agentic–high communal life goals ($\beta = -0.01$, p = 0.78). This pattern provides support for Hypothesis 2a.

Table 2.3

	LA_HC		HA_LC		HA_HC	
Path	ß	р	ß	р	ß	р
WtFI \rightarrow Life						
Satisfaction (T2)	032	.222	179	.001	011	.775
$FtWI \rightarrow Life$						
Satisfaction (T2)	065	.012	.014	.698	026	.498
R^2	.208		.280		.245	

Path Analysis of The Moderating Role of Life Goals

Note. LA_HC: Low agentic–high communal goal endorsement, HA_LC: High agentic–low communal goal endorsement, HA_HC: High agentic–high communal goal endorsement.

The effect of FtWI on life satisfaction was significant only for individuals with low agentic-high communal life goals ($\beta = -0.07$, $p \le 0.05$), while it was not significant for the other two groups ($\beta = 0.01$, p = 0.70 and $\beta = -0.03$, p = 0.50, respectively) (Table 2.3). This partially supports Hypothesis 2b. These findings suggest that concerning life satisfaction, individuals who mainly prioritize agentic life goals are most adversely affected when work interferes with family. In contrast, when family interferes with work, those who prioritize communal goals experience the most significant impact. The changes in scores for WtFI, FtWI,

and life satisfaction at two different time points by groups can be found in Appendix Figures A1-A3.

Next, we explored whether the interaction between life goals and gender moderated the relationship between WtFI and life satisfaction when all control variables were included in the model. Utilizing multi-group comparisons, we aimed to assess the significance of group differences in these relationships. The slope was constrained equally among the groups compared to the unconstrained model. Specifically, we compared two models: one in which the slopes of the relationships between WtFI, as well as FtWI, and life satisfaction, were constrained to be equal across the groups of interaction between life goals and gender (constrained model). The statistical differences between the models were determined using the chi-square difference test. Results displayed in Table 2.4 showed that when work interfered with family (WtFI), significant differences in the effects of WtFI on life satisfaction levels were observed among six groups with different interactions between life goals and gender (χ^2 = 18.867; df = 5; $p \le 0.01$).

The negative effect of WtFI on life satisfaction was significant for both men and women with high agentic–low communal life goals, but the impact was slightly more pronounced among women ($\beta = -.215$, p = .001) than men ($\beta = -.159$, p = .001). Women primarily pursuing agentic life goals did not significantly differ from men with similar life goals ($\chi^2 = 0.439$; df = 1; p = 0.51), failing to support Hypothesis 3a (Table 2.4). However, women primarily pursuing agentic life goals significantly differed from women prioritizing communal life goals ($\chi^2 =$ 4.270; df = 1; $p \le 0.05$) and those with balanced life goals ($\chi^2 = 9.534$; df = 1; $p \le 0.01$). Additionally, it was observed that men primarily pursuing agentic life goals and those prioritizing communal life goals significantly differed in the relationship between WtFI and life satisfaction ($\chi^2 = 8.157$; df = 1; $p \le 0.01$).

Table 2.4

Nested Model Comparisons in The Relationship Between WtFI and Life Satisfaction among Men and Women with Different Life Goals

$WtFI \rightarrow Life Satis$	faction (T2)					χ^2	DF	р
LA_HC_M	LA_HC_W	HA_LC_M	HA_LC_W	HA_HC_M	HA_HC_W	18.867	5	.002
				LA_HC_M	LA_HC_W	3.124	1	.077
				HA_LC_M	HA_LC_W	.439	1	.508
				HA_HC_M	HA_HC_W	1.093	1	.296
			LA_HC_M	HA_LC_M	HA_HC_M	8.188	2	.017
				LA_HC_M	HA_LC_M	8.157	1	.004
				LA_HC_M	HA_HC_M	1.252	1	.263
				HA_LC_M	HA_HC_M	2.260	1	.133
			LA_HC_W	HA_LC_W	HA_HC_W	9.639	2	.008
				LA_HC_W	HA_LC_W	4.270	1	.039
				LA_HC_W	HA_HC_W	2.313	1	.128
				HA_LC_W	HA_HC_W	9.534	1	.002

Note. LA_HC_M: men with low agentic-high communal goal endorsement, LA_HC_W: women with low agentic-high communal goal endorsement, HA_LC_M: men with high agentic-low communal goal endorsement, HA_LC_W: women with high agentic-low communal goal endorsement, HA_HC_W: women with high agentic-high communal goal endorsement, HA_HC_W: women with high agentic-high communal goal endorsement. When assuming the unconstrained model to be correct, the results from a multiple-group comparison analysis indicate whether the relationship between work-to-family interference and life satisfaction differs significantly among men and women with different life goals. The rows represent pairwise comparisons in the relationship between work-to-family interference and women with different life goals.

Similarly, several analyses were conducted on how the interaction between life goals and gender moderated the relationship between FtWI and life satisfaction, with all control variables included in the model. The effect of FtWI on life satisfaction was significant among men with low agentic–high communal goals ($\beta = -.093$, p = .008), while it was not significant among women with the same goals ($\beta = -.033$, p = .393). Results presented in Table 2.5 showed that men who mainly pursued communal life goals did not significantly differ from women with similar life goals ($\chi^2 = 1.696$; df = 1; p = 0.19), failing to support Hypothesis 3b. Men primarily pursuing communal life goals significantly differed from men prioritizing agentic life goals ($\chi^2 = 4.389$; df = 1; $p \le 0.05$) and those with balanced life goals ($\chi^2 = 5.360$; df = 1; $p \le 0.05$).

After exploring whether gender and life goals groups differ in the impact of work-family interference on life satisfaction, we examined the size of the effects for these groups. The effect of WtFI on life satisfaction was significant for both men and women with high agentic and low communal life goals ($\beta = -0.16$, $p \le 0.001$ and $\beta = -0.22$, $p \le 0.001$, respectively; Table 2.6). For women with low agentic and high communal life goals, the effect of WtFI on life satisfaction was significant ($\beta = -0.09$, $p \le 0.05$), while it was insignificant for those with balanced life goals ($\beta = 0.02$, p = 0.71). The effect was insignificant for men primarily pursuing communal life goals as well as those with balanced goals ($\beta = 0.01$, p = 0.72; $\beta = -0.06$, p = 0.26, respectively).

The impact of FtWI on life satisfaction was significant only for men with low agentic and high communal life goals ($\beta = -0.09$, p ≤ 0.01) and was insignificant for all other gender and life goals groups (Table 2.6). These findings suggest that when family interferes with work, the life satisfaction level of men prioritizing communal life goals is the most affected.

Table 2.5

Nested Model Comparisons in The Relationship Between FtWI and Life Satisfaction among Men and Women with Different Life Goals

$FtWI \rightarrow Life Satisfa$	action (T2)					χ^2	DF	р	
LA_HC_M	LA_HC_W	HA_LC_M	HA_LC_W	HA_HC_M	HA_HC_W	8.676	5	.123	
				LA_HC_M	LA_HC_W	1.696	1	.193	
				HA_LC_M	HA_LC_W	.253	1	.615	
				HA_HC_M	HA_HC_W	2.961	1	.085	
			LA_HC_M	HA_LC_M	HA_HC_M	7.366	2	.025	
				LA_HC_M	HA_LC_M	4.389	1	.036	
				LA_HC_M	HA_HC_M	5.360	1	.021	
				HA_LC_M	HA_HC_M	.001	1	.979	
			LA_HC_W	HA_LC_W	HA_HC_W	1.276	2	.528	
				LA_HC_W	HA_LC_W	.275	1	.600	
				LA_HC_W	HA_HC_W	.638	1	.425	
				HA_LC_W	HA_HC_W	1.253	1	.263	

Note. LA_HC_M: men with low agentic-high communal goal endorsement, LA_HC_W: women with low agentic-high communal goal endorsement, HA_LC_M: men with high agentic-low communal goal endorsement, HA_LC_W: women with high agentic-low communal goal endorsement, HA_HC_W: women with high agentic-high communal goal endorsement, HA_HC_W: women with high agentic-high communal goal endorsement. When assuming the unconstrained model to be correct, the results from a multiple-group comparison analysis indicate whether the relationship between family-to-work interference and life satisfaction differs significantly among men and women with different life goals. The rows represent pairwise comparisons in the relationship between family-to-work interference and women with different life goals.

Table 2.6

	LA_HC	_M	LA_HC	_W	HA_LC	_M	HA_LC	W	HA_HC	_M	HA_HC	_W
Path	ß	р	ß	р	ß	р	ß	p	ß	p	ß	p
WtFI \rightarrow Life Satisfaction (T2)	.013	.723	085	.034	159	.001	215	.001	059	.262	.020	.708
FtWI \rightarrow Life Satisfaction (T2)	093	.008	033	.393	.037	.447	.003	.951	.049	.365	081	.134
R ²	.269		.154		.276		.301		.305		.222	

Path Analysis of The Moderating Role of Life Goals and Gender

Note. LA_HC_M: men with low agentic-high communal goal endorsement, LA_HC_W: women with low agentic-high communal goal endorsement, HA_LC_M: men with high agentic-low communal goal endorsement, HA_LC_W: women with high agentic-low communal goal endorsement, HA_HC_W: women with high agentic-high communal goal endorsement.

2.5. Discussion

In this research, our objective was to examine whether life goals and gender moderate the long-term effects of work-family interference (both WtFI and FtWI) on life satisfaction. Our results indicate that life goals play an important moderating role in this relationship. Specifically, interference between life domains reduces satisfaction with life, especially when the domain containing primary life goals interferes with the other domain.

After controlling baseline life satisfaction levels at Time 1, our findings reveal that interference between work and family domains negatively predicts life satisfaction two years later (T2). These results align with previous studies (e.g., Grant-Vallone & Donaldson, 2001; Yucel & Fan, 2019). It should be noted that the effect sizes are relatively small, consistent with similar research results (Matthews et al., 2014; Yucel & Borgmann, 2022). This can be attributed to the multicausal nature of stressor-strain relationships, where various factors influence different aspects of life satisfaction, resulting in the relatively weak impact of a single stressor (Zapf et al., 1996). Beyond these primary impacts, we found that they can vary according to life goals.

Individuals who prioritize agentic life goals experience a significant decrease in life satisfaction when work interferes with family, in contrast to those with primarily communal life goals or a balanced set of goals. Similarly, when family interferes with work, individuals who prioritize communal goals over agentic goals experience a significant decrease in life satisfaction. This finding does not extend to individuals with primarily agentic life goals or a balanced set of goals where the effects were not significant. How can we understand the finding that the satisfaction levels of individuals are more affected by interference from their prioritized domain rather than personally less relevant domain? First, these findings resonate with Noor's (2004) study on role centrality, suggesting that prioritizing goals in a specific domain heightens sensitivity to stressors in that area, leading to increased negative impacts on life satisfaction

due to a perceived lack of control. Second, there appears to be a mutual relationship between the challenges of balancing different roles and a deficit in self-regulatory resources (Khatri & Shukla, 2022). This involves a mechanism not only of perceiving interference but also of being affected by it. While literature suggests that individuals are willing to tolerate higher levels of WtFI for domains they value more (Greenhaus et al., 2001; Matthews et al., 2012), if the interference persists, the lack of return on resource investment and a sense of losing control over the targeted domain result in a greater decline in satisfaction compared to those who have not invested resources in that domain. This dynamic resonates with Hobfoll's (1989) *loss spiral* concept. It illustrates a cyclical relationship where individuals feel vulnerable and lose control in their prioritized domain due to domain-specific stresses (instead of fulfillment) despite their investment efforts. This may lead to further investment in their life-goals-related domain to regain control. However, excessive investment can ultimately deplete their resources, leaving them more susceptible to interference from that domain.

The absence of a significant effect of WtFI on life satisfaction among individuals with primarily communal life goals, as well as the absence of the effect of FtWI on life satisfaction among individuals with primarily agentic life goals, may be surprising when considering previous research (e.g., Carr et al., 2008). In addition to the rationale that we presented above, figures A1-A3 in the Appendix may help to clarify these results. For individuals who prioritize communal life goals, there is a slight increase in WtFI between the two-time points, and similarly, there is an increase in life satisfaction between the two time points. Strong family bonds and social support systems can mitigate the negative effects of WtFI on these individuals' overall life satisfaction. Conversely, for those who prioritize agentic life goals, FtWI shows a decrease between the two-time points, and their life satisfaction also decreases over the same period. The career focus and access to professional support resources of individuals with agentic life goals can help alleviate the negative effects of FtWI, but not those of WtFI, on life

satisfaction. As their FtWI decreases, it might signal that they are dedicating more time and energy to work at the expense of family; they might not perceive FtWI as a primary factor affecting their life satisfaction because their main focus is on their career.

Another central objective of our study was to explore the interaction between life goals and gender. Contrary to expectations, neither the effect of WtFI nor FtWI on life satisfaction varies between women and men with the same life goals. Both agentic women and men experience similar levels of life satisfaction decline when work interferes with family life. The significant negative effect observed for all genders indicates that when individuals are highly focused on achieving agentic, career-oriented life goals, the demands of work that spill over into family life are equally disruptive, regardless of gender. In a comparatively genderegalitarian society like Germany, the shared challenges of balancing demanding professional aspirations with family responsibilities might explain this similarity.

On the other hand, the effect of FtWI on life satisfaction presents a different pattern. We observed that FtWI significantly impacts life satisfaction for men who prioritize communal life goals, but this effect is not significant for women with similar goals. This finding suggests that men who focus on communal, family-oriented goals are particularly sensitive to disruptions caused by family demands interfering with their work lives. The lack of a significant effect for women may indicate that women, even those with communal goals, might have developed more effective coping strategies or possess higher resilience in managing FtWI, possibly due to traditional socialization and experience in handling family responsibilities (Elman & Gilbert, 1984; Matias & Fontaine, 2015). While the groups individually show different relationships, with one being significant and the other not, there is no statistically significant difference between the groups, which may be due to small effect sizes. The absence of significant differences between women and men with communal life goals in the impact of FtWI on life satisfaction further underscores the nuanced interplay between gender and life goals. This

suggests that communal goals inherently involve a high degree of family involvement and commitment, leading to similar challenges for both men and women when family demands encroach upon work.

These results align with social role theory, which posits that differences in binary gender roles should be particularly notable in less egalitarian countries (Kosakowska-Berezecka et al., 2023; Wood & Eagly, 2012) and less notable in more gender-egalitarian countries like Germany, so that gender-related stereotypical expectations should be less dominant. Importantly, our study measured life goals rather than assuming predefined life goals for men and women, revealing that gender does not affect the relationship between work-family interference and life satisfaction for individuals with the same life goals. The interaction effect of life goals and gender, particularly when individuals prioritize life goals that contradict stereotypically gendered societal expectations, was not found to be significant. Instead, our findings indicate that it is the centralization of specific life goals, regardless of gender, that strengthens the negative impact of interference from the prioritized domain on life satisfaction.

In conclusion, our results affirm that life goals can be viewed as a resource shaping the relationship between work-family interference and life satisfaction. Building on the Work-Home Resources model, prior research has recognized individual differences, including psychological capital (Choi et al., 2018), self-efficacy (Bakker et al., 2019), and motivation (ten Brummelhuis et al., 2011), as resources in this relationship. Our research is the first to demonstrate that life goals and the self-regulatory resources utilized in pursuing them also play a significant role as critical resources in the work-family interference and life satisfaction relationship.

We acknowledge certain limitations of the study and propose possible alternative strategies for future research. While our sample, representative of Germany, provides valuable

insights, our rigorous exclusion of incomplete data might have slightly compromised representativity to an extent we consider negligible. Additionally, our findings may not readily generalize to cultures that are more collectivistic or adhere to different gender-role norms. Another limitation lies in using a one-item life satisfaction instrument, potentially introducing reliability concerns. Also, some may argue against measuring life goals interdependently with the distribution of 15 importance points among the domains, suggesting that future research should consider independent measurement. Lastly, we assessed the moderating effects of life goals and gender on the relationship between work-family interference itself and affect life satisfaction through work-family interference. Future research should explore these alternative relationships.

In summary, this study significantly contributes to the literature and has practical applications. We used longitudinal data spanning two years to examine the role of life goals on the effect of both WtFI and FtWI on life satisfaction. Going beyond the Work-Home Resources model, our study extends it by recognizing self-regulatory resources as key contributors. The work-family interference literature has often overlooked moderators related to individual identities and characteristics. Our study sheds light on the complex dynamics by employing the concept of life goals as an individual-related mechanism. While investing resources in a domain related to life goals may bring short-term satisfaction, our results indicate that the long-term impact - as interference with other domains increases - can potentially lead to a more significant decline in life satisfaction.

Our findings underscore the importance of considering individual differences to mitigate the adverse impact of work-family interference on life satisfaction. This insight is valuable for practitioners, suggesting the need to educate human resources professionals about the detrimental impact of work-family interference on employees and, thereby, the organization. Creating awareness and providing support training to stakeholders can empower individuals to take control of their life goals, potentially reducing the negative effects of work-family interference.

Chapter 3: Exploring the Mediating Role of Work-Family Interference in the Complex Dynamics of Life Goals and Well-Being

Abstract

In an era where balancing work and family has become increasingly challenging, this study examines the impact of agentic and communal life goals on well-being, with work-family interference serving as a mediating factor. The research is conducted within the diverse cultural settings of Germany and Türkiye and further explores the moderating roles of culture and trait self-control in these relationships. The study sample consists of white-collar employees (662 residing in Germany and 590 in Türkiye), and data were collected using cross-sectional, selfreport online surveys, including the Work-Family Conflict Scale (Carlson et al., 2000) and the GOALS Questionnaire (Pöhlmann & Brunstein, 1997). Structural equation modeling results reveal that prioritizing agentic goals intensifies work-to-family interference, thereby negatively affecting well-being. In contrast, prioritizing communal goals heightens family-to-work interference, indirectly reducing overall well-being. A notable finding is the adverse effect on women's well-being due to the misalignment between their agentic goals and societal gender role expectations. Interestingly, individuals with high self-control, while generally more satisfied with their lives, tend to underestimate the demands associated with communal goals, leading to increased family-to-work interference. The study concludes by discussing the implications of these findings, highlighting the complex interplay between life goals, workfamily interference, and overall well-being.

Keywords

Culture; life goals; self-control; well-being; work-family interference

3.1. Introduction

In the contemporary landscape of work and family dynamics, achieving a healthy balance between professional duties and personal life is a significant challenge, particularly for whitecollar workers whose work-life boundaries can blur easily (Peng et al., 2011). Advancements in technology have made it easier to bring work home, contributing to well-being issues among white-collar employees, as it has been discovered that 80% of white-collar employees experience at least one mental health issue (Deloitte, 2022). This underscores the critical importance of understanding the consequences of eliminating these boundaries—a relevance that persists in today's context.

Extensive research has delved into the adverse spillover effects between work and family on overall well-being (Amstad et al., 2011; Mullen et al., 2008), including psychological distress (Kinnunen et al., 2006) and overall life satisfaction (de Simone et al., 2014). However, there is a notable gap regarding individual characteristics that impact work-family interference. Interactions between work and family may be driven by personal aspirations, namely life goals. This study addresses this gap by considering life goals as a crucial individual factor. Additionally, while some studies have explored the relationship between life goals and wellbeing (Macleod, 2012; Saragovi et al., 2002; Sheldon & Elliot, 1999), the results are not consistent and comprehensive. Moreover, little is known about how these goals affect different domains. This research aims to elucidate why goal endorsement can negatively affect wellbeing through the mechanism of work-family interference.

According to the Work-Home Resources model, which is a model that establishes connection mechanisms between contextual demands and resources and work or home outcomes, persistent demands in one domain lead to negative outcomes in the long term, as personal resources are invested to meet these demands (ten Brummelhuis & Bakker, 2012). The pursuit of life goals, which can be categorized according to Bakan's (1966) agenticcommunal conceptualization, also requires personal resources, and self-regulation plays a vital role in conserving and managing these resources (Zimmerman, 2000). However, ongoing resource depletion while pursuing life goals can result in stress, ultimately impacting well-being (Baumeister & Vohs, 2007; Hobfoll, 2011).

In addition to the mediating role of work-family interference on the relationship between life goals and well-being, this study explores moderating relationships, focusing on macro resources (cultural environments) and key resources (individual traits; ten Brummelhuis & Bakker, 2012), examining how societal expectations around gender roles and individual traits like self-control interact and impact well-being. Cultural values, such as individualism in societies like Germany versus collectivism in societies like Türkiye, shape gender role expectations, influencing whether societies support agentic or communal goals (Ng et al., 2003; Sedikides et al., 2003). Deviation from these societal norms, such as women pursuing agentic goals or men pursuing communal goals, can adversely affect well-being (Haines & Stroessner, 2019; Sczesny et al., 2018) due to the additional strain beyond work-family interference and a lack of macro resources. Furthermore, self-control, essential for achieving life goals and managing stressors, is recognized as a protective factor against the detrimental impacts of stress (de Ridder et al., 2012; Clinton et al., 2020). High self-control is associated with better flexibility in goal setting and strategies under stress, thereby enhancing the balance between work and family and maintaining well-being (Johnson et al., 2018; Lanaj et al., 2014; Wrosch et al., 2003). Therefore, this research aims to examine the moderating effects of these cultural/gender-related factors and self-control traits on the relationship between life goals, work-family interference, and well-being.

Altogether, drawing on the theoretical frameworks of self-regulation and the Work-Home Resources model, this study aims to explore how white-collar employees strategically allocate their cognitive and emotional resources to manage the demands of work and family while pursuing life goals across various domains and to understand how the interference between these domains impacts their overall well-being. By integrating insights from cultural and gender-specific influences, as well as the role of self-control, this research contributes to understanding the well-being of white-collar employees navigating the complex intersection of work and family life. Before discussing theoretical insights, the conceptual model in Figure 3.1 of the relationships that will be examined can provide a comprehensive perspective.

Figure 3.1





3.2. Theoretical Framework

3.2.1. The Relationship Between Well-Being and Goal Endorsement

Subjective well-being refers to how individuals evaluate their own lives, including cognitive judgments of satisfaction, influenced by various life domains like job, family, or personal life (Diener et al., 2002). In the literature, there are two predominant perspectives on well-being: overall life satisfaction affecting well-being in life domains and satisfaction in life domains influencing overall life satisfaction (Heller et al., 2004). Stress also is a significant outcome of the interaction between an individual and their environment (Lazarus, 1999; Lazarus & Folkman, 1984). Hobfoll (2011) states that stress occurs as a response to personal

resource loss since individuals strive to obtain, retain, and enhance their resources. Persistent or chronic stress can exert a detrimental effect on overall life satisfaction. This interplay between satisfaction, stress, and overall well-being, as well as resource management, underscores the intricate relationship between one's life evaluations and mental and emotional states (Kahneman et al., 1999; Lazarus, 1999).

Given the pivotal role well-being plays in the essential spaces of individuals' lives, researchers have delved into what factors or influences can potentially undermine it. Wellbeing is shaped by personal experiences, each experience being significantly influenced and guided by overarching life goals. Life goals are concepts that give meaning to people's lives (Baumeister, 1991), and they can enhance the chance for long-term well-being (Brunstein, 1993; Emmons, 1996). They are described as internal representations of desired states (Austin & Vancouver, 1996) and are considered to be the directory of actions (Gollwitzer & Bargh, 1996). Thus, they can provide an understanding of people's behavior patterns (Emmons, 1996).

Researchers have employed various dichotomies to delineate goals, such as mastery versus performance goals (Poortvliet & Darnon, 2010), approach versus avoidance goals (Elliot, 1999), intrinsic versus extrinsic goals (Kasser & Ryan, 2001), and individual striving versus harmony seeking (Wicker et al., 1984). Among these, a prominent classification distinguishes between agency and communion (Bakan, 1966). Research has consistently identified two essential dimensions —often referred to as the "Big Two"—that pervade multiple areas of social cognition: agency and communion (Abele et al., 2016). These two dimensions are considered overarching categories that encompass all other dichotomies and are believed to correspond to two core components of self-concept (Wojciszke & Abele, 2008). The agency is employed to define an organism's existence as an individual, while communion determines an individual's engagement within a larger organism of which they are a part. The distinction between agency and communion lies in the former's focus on the self and the latter's

focus on others (Helgeson, 1994). In the literature, agentic strivings are depicted as selfachievement, power, variation (Pöhlmann, 2001), influence, control, and mastery (Horowitz et al., 2006), as well as assertiveness and competence (Abele & Wojciszke, 2014). On the other hand, communal strivings are depicted as intimacy, affiliation, altruism (Pöhlmann, 2001), connection, participation (Horowitz et al., 2006), as well as warmth, and morality (Abele & Wojciszke, 2014). In summary, theoretically, three categories of life goals (achievement, power, and variation) form a broader group known as the field of agentic life goals. The other three life goals (intimacy, affiliation, and altruism) constitute the field of communal life goals (Pöhlmann & Brunstein, 1997).

The pursuit of a goal—identifying which goal is being pursued (content), understanding why it is being pursued (importance), determining the actions taken to achieve it (achievability), specifying how it is being pursued (orientation), and assessing how it concludes (success)—results in various impacts on well-being. (Emmons, 1991; Pöhlmann & Brunstein, 1997; Sheldon et al., 2010). Previous studies have demonstrated that the pursuit of agentic and communal goals can impact well-being in several ways (Helgeson & Fritz, 1999; Hennecke & Brandstätter, 2017). For instance, it has been found that the proportion of the intimacy goal among the other goals has the most important positive effect on well-being (Colby, 1996). Conversely, it also has been found that agency-motivated goals were associated with more positive mental health. That is, agentic-motivated people were better at dealing with internal and external demands (Pöhlmann, 2001). In a recent study, it was found that both intimacy and altruism goals, as well as achievement and power goals, have a positive relationship with life satisfaction (Tucak Junaković, 2015; Sheldon & Cooper, 2008). From another perspective, setting various life goals may inherently signify a disparity between current circumstances and desired outcomes. This means that the behaviors guided by the life goals can even negatively affect well-being (MacLeod, 2012). In another study, the proportion of achievement and power goals have the most significant negative impact on well-being (Emmons, 1991). In essence, the intricate interplay of specific life goals—ranging from intimacy and altruism to achievement and power—profoundly influences well-being, offering a nuanced perspective on the intricate connections between goal pursuit, life satisfaction, and stress. The contradictory findings in the literature underscore the importance of understanding these nuances.

Self-regulation theory is frequently used to explain the link between life goals and wellbeing. This theory emphasizes the idea of guiding thoughts, feelings, and actions toward achieving goals (Zimmerman, 2000). Life goals guide individuals' actions, and individuals experience satisfaction with their lives when their life's trajectory aligns with the goals they have set for themselves (Oishi, 2000) and when they regulate their actions accordingly (Hofmann et al., 2014). Research has shown that individuals who actively pursue their goals tend to have higher well-being levels than those without concrete goals (e.g., Freund & Baltes, 2002). Goal striving has been found to promote improved psychological well-being, as demonstrated in a meta-analysis (Klug & Maier, 2015). People generally feel happy when they believe they are making reasonable progress toward their goals (Lazarus, 1991, p. 267). Here, coming from the idea that goal pursuing may lead to gains in the sense of fulfillment, and one can delve into the bright side of the narrative: Individuals with diverse goals may witness an uplifting improvement in their well-being, an elevation in life satisfaction, and a reduction in stress. Thus, based on self-regulation theory, we hypothesize the positive effects of pursuing life goals on well-being. This aligns with the idea that well-regulated goals often lead to positive outcomes in well-being. Both agentic and communal life goals provide individuals with a sense of meaning and purpose in their lives rather than acting as stressors, thus playing a significant role in reducing stress and increasing life satisfaction. Consequently, as indicators of a sense of fulfillment derived from investing in resource enrichment, this study focuses on

stress and life satisfaction as primary indicators of well-being. Specifically, we hypothesize the following (see Table 3.1):

Table 3.1

Hypothesis 1: The Relationship Between Well-Being and Goal Endorsement

Hypothesis	Goal Type	Relationship Type	Dependent Variable	Relationship Direction
H1a	Agentic	Direct	Life Satisfaction	Positive
H1b	Agentic	Direct	Stress	Negative
H1c	Communal	Direct	Life Satisfaction	Positive
H1d	Communal	Direct	Stress	Negative

The prominence of agency and communion aspects in an individual's self-concept reflects their tendency to engage in strategies to achieve these different life goals (Uchronski, 2008). Individuals with specific life goals use their self-regulation abilities to allocate resources toward achieving these goals. For those who place high importance on a particular life goal (whether agentic or communal), role pressures become particularly significant because success and rewards in that domain are crucial to them (Greenhaus & Beutell, 1985). As a result, they strive to manage these pressures by intensifying their efforts and investments in the work or family domain. Rothbard and Edwards (2003) support the idea that the importance attributed to work and family domains is positively associated with the time and energy allocated to those domains. Therefore, in this study, we adopt the perspective that agentic goals are more related to the work domain, while communal goals are more related to the family domain.

Pursuing the desired states mentioned above also requires an investment of resources, and investment in life goals may deplete the limited resources (Muraven & Baumeister, 2000). Namely, successful development in one goal might be associated with negative outcomes in another (Marsiske et al., 1995). Similarly, the allocation of resources to pursue specific goals in one life domain, whether it be in the realm of work or family, may unintentionally result in resource deficiencies in another. It means the goals pursued in one domain may drain an individual's resources, leaving less to fulfill the requirements in other domains. This scenario can give rise to conflicts between different life domains. Hence, the mechanism elucidating the relationship between life goals and well-being may involve the experience of conflict, and this experience can cast a shadow over positive relationships.

3.2.2. The Mediating Role of Work-Family Interference

Work-family interference refers to the conflicts or tensions that arise when the demands and responsibilities of work and family roles are in opposition or interfere with each other (Greenhaus & Beutell, 1985). There are two directions of interference: work-to-family and family-to-work (Frone et al., 1992). Work-to-family interference (WtFI) occurs when the demands and responsibilities of work interfere with family life. On the other hand, family-towork interference (FtWI) unfolds when family obligations hinder an individual's capacity to fulfill his/her occupational duties. Juggling responsibilities in multiple roles - such as work and family - is categorized into time-based (conflicts due to time constraints) and strain-based (stress spillover, making obligations challenging; Elloy & Smith, 2004).

The Work-Home Resources model defines work-family interference as a process whereby demands in either the work or home domain can drain personal resources, leading to reduced effectiveness in the other area (ten Brummelhuis & Bakker, 2012). This model is grounded in the Conservation of Resources (Hobfoll, 1989) theory, which posits that individuals strive to obtain, preserve, and enhance their resources, such as time and energy, which are critical for managing both work and home responsibilities effectively. However, any failure in these conservation efforts eventually leads to resource depletion (Hobfoll, 2011). Negative outcomes in the work, family, and health-related areas result from ineffective management of life given a finite number of personal resources (time, energy, and money) (Grawitch et al., 2010). Such a shortage causes distress due to an enhancement of loss over gain perception (Hobfoll, 2011). Thus, work-family interference arises when one domain is overinvested, depleting resources for the other (Grandey & Cropanzano, 1999), leading to long-term well-being reduction (Hobfoll, 2011).

The consequences of work-family interference can be grouped into work, non-work, and stress-related outcomes (Allen et al., 2000). Meta-analyses that support the matching hypothesis reveal that stronger interference relationships within the same domain, namely WtFI was more strongly associated with work-related than family-related outcomes, and FtWI was more strongly associated with family-related than work-related outcomes (Amstad et al., 2011; Nohe et al., 2014; Shockley & Singla, 2011). However, both types of interference have been found to be most associated with domain-unspecific outcomes, namely general life satisfaction and stress (Amstad et al., 2011).

When delving into the precursors of interference, the prevalent viewpoint in literature leans towards domain specificity (Frone et al., 1992; 1997). This implies that stressors and engagement in either work or family roles can lead to interference from the corresponding domain into the other rather than vice versa (Nohe et al., 2014). This situation is explained in the Work-Home Resources model with contextual demands and personal resources - the contextual demands from one domain lead to losing personal resources and resulting in diminished outcomes of another (ten Brummelhuis & Bakker, 2012). This perspective has also been corroborated in meta-analyses examining the antecedents of work-family interference, which typically encompass a wide array of factors falling under broad categories, including work and family-related sources (Byron, 2005). Work-related antecedents, including workrelated demands like extended work hours and unsupportive supervisors or coworkers, relate to WtFI more than FtWI. Meanwhile, family-related antecedents, including family-related

88

demands such as the number of children and unsupportive or demanding spouses, relate to FtWI more than WtFI (see Byron, 2005; Ford et al., 2007; Michel et al., 2011 for an overview).

Despite the numerous endeavors in extensive literature to unravel the antecedents of work-family interference, a significant gap remains in acknowledging individual differences. While some studies have explored dispositional variables like personality and gender (Miller et al., 2022; Shockley et al., 2017), as highlighted by Byron (2005) and Allen et al. (2012), the existing literature has predominantly focused on situation-based antecedents. This leaves a notable void in understanding the impact of individual-related or dispositional factors. Recognizing the critical importance of these factors is essential for a more comprehensive understanding of work-family interference (Hargis et al., 2011). Among these, personal life goals emerge as a significant and often overlooked contributor to work-family interference.

Work and family are two key life domains where people pursue their life goals, often leading to trade-offs between them due to conflicting demands (Hirschi et al., 2019; Mickel & Dallimore, 2009). People want to achieve their long-term goals and are willing to make sacrifices for them (Baumeister & Vohs, 2007). This means that they prioritize and actively engage in behaviors to achieve their personally valued life goals (Unsworth et al., 2014). As a result, individuals divide their time and energy between work and family responsibilities, highlighting the motivation to allocate resources toward a domain related to their life goals (Grawitch et al., 2010). However, this resource allocation process may come together with the challenge of overspending available resources to pursue one goal at the cost of other life goals (Hirschi et al., 2019). It is also mentally exhausting and often leads to resource depletion and increased stress levels, ultimately impacting well-being (Hobfoll, 2011; Muraven & Baumeister, 2000).

Thus, individuals who pursue agentic goals related to their job, career, or education, as well as those who prioritize communal goals linked to personal relationships and caregiving, are likely to experience work-family interference in distinct ways. In the context of self-regulation and the Work-Home Resources model, we believe that pursuing agentic life goals focused on personal achievement, often linked to work, can result in a higher dedication of personal resources to work requirements, thereby increasing work demands. On the other hand, pursuing communal life goals centered around relationships, typically associated with family, might cause more personal resources to be allocated to family activities, leading to increased family demands. In support of the matching hypothesis, the unfavorable aspect of the story is that those pursuing agentic goals may face heightened WtFI due to the increased allocation of resources to work. Conversely, individuals pursuing communal goals may often grapple with FtWI as more resources are allocated to the family. The different dimensions of work-family interference will negatively influence the connection between different life goals and well-being (see Table 3.2).

Table 3.2

Hypothesis 2: The Mediating Role of Work-Family Interference

Hypothesis	Goal Type	Indirect Relationship	Dependent Variable	Relationship Direction
H2a	Agentic	WtFI	Life Satisfaction	Negative
H2b	Agentic	WtFI	Stress	Positive
H2c	Communal	FtWI	Life Satisfaction	Negative
H2d	Communal	FtWI	Stress	Positive

The Work-Home Resources model defines the conditions in which work-family interference emerges and is experienced through key resources - the characteristics of individuals, and macro resources - the contexts in which individuals live (ten Brummelhuis & Bakker, 2012). Building on this, the study examines self-control traits as key resources and

societal expectations concerning gender roles as macro resources. It investigates how these conditional factors moderate the relationship between life goals, work-family interference, and dimensions of well-being.

3.2.3. The Moderating Role of Self-Control

Key resources, such as self-control, explain why some people are better than others at coping with stressful situations, such as self-control (ten Brummelhuis & Bakker, 2012). Self-control is the remarkable ability to regulate one's own emotions, thoughts, and behaviors in pursuit of long-term goals (Goschke & Job, 2023). Research has shed light on its protective role in the face of stressful situations (Clinton et al., 2020; Tangney et al., 2004). It has been shown in a meta-analysis that self-control can bring about desired outcomes by helping individuals regulate their behavior, thoughts, and emotions in a way that promotes improved psychological well-being and overall life satisfaction. It guides goal-directed behavior and serves a valued long-term goal, leading to a happier and more fulfilling life (de Ridder et al., 2012). Like every resource, the self-control strength model (Muraven & Baumeister, 2000) suggests that self-control functions like a muscle, with a limited, consumable strength similar to a muscle's capacity. This implies that individuals may experience ego depletion when facing multiple demands simultaneously, even if they would have otherwise been successful (Baumeister et al., 2018).

The Personal Resource Allocation framework argues that balancing the interaction between work and family is about how individuals allocate their personal resources when confronted with various demands (Grawitch et al., 2010). In other words, individuals may be motivated to invest their personal resources, such as time and energy, into specific domains to achieve their life goals. However, the depletion of these resources can make them more vulnerable and fragile in work and family interactions (Lanaj et al., 2014; Rothbard, 2001). Therefore, the level of dispositional self-control is crucial as a key resource in managing all other personal resources. Hofmann et al. (2014) found that self-control influences well-being by balancing resources in main life domains, such as work and family.

Hence, it is hypothesized that individuals with high self-control abilities, while pursuing life goals, can allocate certain resources to domains beyond these goals, resulting in a lower experience of work-family interference compared to individuals with low self-control abilities. Similarly, individuals with high self-control abilities are expected to manage better external temptations in their journey towards achieving these goals, leading to a higher level of wellbeing compared to those with lower self-control abilities (see Table 3.3).

Table 3.3

Hypothesis	Goal Type	Relationship Type	Dependent Variable	Moderator	Effect Direction
H2a	Agantia	Direct	Life	Self-	Stronger (High
пза	Agentic	Positive	Satisfaction	Control	Self-Control)
U2b	Agantia	Direct	Strong	Self-	Stronger (High
пзо	Agentic	Negative	511688	Control	Self-Control)
Ш2а	Agontia	Direct	W/+FI	Self-	Weaker (High
1150	Agennie	Negative	vv tr 1	Control	Self-Control)
Ш24	Communal	Direct	Life	Self-	Stronger (High
1150	Communai	Positive	Satisfaction	Control	Self-Control)
Ш2а	Communal	Direct	Strong	Self-	Stronger (High
пзе	Communai	Negative	511688	Control	Self-Control)
H3f	Communal	Direct		Self-	Weaker (High
	Communal	Negative		Control	Self-Control)

Hypothesis 3: The Moderating Role of Self-Control

3.2.4. The Moderating Role of Societal Expectations Concerning Gender Roles

Macro resources, such as cultural values, refer to the macro-level factors surrounding the interface between different domains (Brummelhuis & Bakker, 2012). Different cultural values shape the expectations associated with gender roles in different cultures and significantly impact how individuals manage work-family interactions, thereby influencing their overall well-being. Individualism-collectivism and egalitarianism-hierarchy can be considered cultural

values (Hofstede, 2001). Individualistic cultures prioritize personal independence, leading to higher reported happiness levels compared to collectivistic cultures, which prioritize group harmony (Diener & Diener, 1995). In societies with high power distance, hierarchy is prevalent, while those with low power distance emphasize equality (Hofstede, 2001). The link between the Big Two (agentic and communal; Abele & Wojciszke, 2014) and the standard dimensions of cross-cultural comparisons, individualism, and collectivism provides a framework for understanding how agency and communion are associated with these cultural dimensions (Markus & Kitayama, 1991). Individualism aligns with agency, as individualists prioritize differentiation from others, while collectivism aligns with communion, as collectivists prioritize assimilation with others (Gebauer et al., 2013). In more individualistic and egalitarian societies, there is a greater emphasis on agentic traits. On the other hand, in societies that are more collectivistic and hierarchical, communal traits tend to be more supported (Ng et al., 2003; Sedikides et al., 2003; Wojciszke & Bialobrzeska, 2014). In more individualistic cultures, work is seen as a personal achievement, and individuals prioritizing work over family are more sensitive to family demands, experiencing the negative effects of FtWI on well-being (Spector et al., 2007). Conversely, in more collectivistic cultures, individuals may be more sensitive to work demands, and experiencing WtFI may threaten family identity and affect well-being (Aycan, 2008).

Gender differences arise from stereotypical expectations associated with societal roles, often categorized as agentic (power-oriented) and communal (connection-oriented), contributing to gender-related behavioral disparities (Bakan, 1966; Eagly & Wood, 2012). Therefore, men and women may be influenced differently by these cultural values on gender roles. Considering the stereotypical gender roles where men are viewed as breadwinners and women are viewed as caregivers (Eagly & Wood, 2012), it can be argued that as women pursue communal goals, they will be compatible with societal expectations for their gender role, while as men pursue agentic goals, they will be compatible with societal expectations for their gender role (Diekman & Eagly, 2008; Wood et al., 1997). Studies indicate that aligning one's actions with these expectations leads to positive emotional experiences, while deviating from them results in negative emotions (Haines & Stroessner, 2019; Rudman et al., 2012; Sczesny et al., 2018). Therefore, it can be assumed that when life goals align with both cultural and gender expectations, it is more likely to experience positive or beneficial effects on well-being. However, failure to meet these expectations can result in a lack of access to macro resources, which can eventually have a negative impact on well-being. Those who do not behave in line with societal expectations, namely women who pursue agentic goals and men who pursue communal goals, face some penalties (Haines & Stroessner, 2019; Sczesny et al., 2018).

Germany is characterized by more individualistic, egalitarian, and agency-oriented practices. In contrast, Türkiye emphasizes collectivistic, hierarchical, and communion-oriented practices (Hofstede, 2001). Therefore, in Türkiye, women pursuing agency goals may face extra challenges due to the pressure to meet societal expectations of caring for the family, while in Germany, men pursuing communal goals may struggle more due to pressure to meet societal expectations of being breadwinners. It is hypothesized that as Turkish women give importance to agentic goals, they may face more significant challenges; also, if they experience WtFI, their well-being will be affected more negatively than German women. Similarly, as German men value communal goals, they may experience heightened challenges; also, if they experience FtWI, their well-being will be affected more negatively compared to Turkish men (see Table 3.4).
Table 3.4

Hypothesis	Independent Variable	Relationship Type	Dependent Variable	Moderator	Effect Direction
		Direct	Life	Culture x Gender	Weaker (Turkish
H4a	Agentic	Positive	Satisfaction	Interaction	Women)
	C	Direct		Culture x Gender	Weaker (Turkish
H4b	Agentic	Negative	Stress	Interaction	Women)
		Direct	Life	Culture x Gender	Stronger (Turkish
H4c	WtFI	Negative	Satisfaction	Interaction	Women)
		Direct		Culture x Gender	Stronger (Turkish
H4d	WtFI	Positive	Stress	Interaction	Women)
		Direct	Life	Culture x Gender	Weaker (German
H4e	Communal	Positive	Satisfaction	Interaction	Men)
		Direct		Culture x Gender	Weaker (German
H4f	Communal	Negative	Stress	Interaction	Men)
		Direct	Life	Culture x Gender	Stronger (German
H4g	FtWI	Negative	Satisfaction	Interaction	Men)
		Direct		Culture x Gender	Stronger (German
H4h	FtWI	Positive	Stress	Interaction	Men)

Hypothesis 4: The Moderating Role of Societal Expectations Concerning Gender Roles

3.3. Method

3.3.1. Procedure

An online questionnaire was prepared using the UNIPARK system. It is available in English, German, and Turkish. All measurements that had not been previously translated into the relevant language were translated by the researcher using the back-translation method (Brislin, 1970). Participants were given the option to choose their preferred language before starting the survey. The initial intention was to access white-collar employees through direct collaboration with their employers. However, faced with an inadequate number of participants in this category, a strategic shift occurred. Participants residing in Germany were reached through Norstat GmbH. The survey company shared the online questionnaire through relevant links, thereby employing a convenience sampling technique. A snowball sampling technique was used to recruit participants in Türkiye. Consequently, the researcher shared the online questionnaire with white-collar employees living in Türkiye within their own networks,

requesting them to share the questionnaire further within their networks. In both cases, ethical considerations emphasizing the rights and privacy of participants, including informed consent and confidentiality, were communicated to the participants at the beginning of the survey. After the distribution of the online surveys, data collection in both countries was completed within approximately two months, from June till the end of August 2023.

The data analysis process began with an examination of descriptive findings. Subsequently, confirmatory factor analysis was conducted for general data, as well as for each country, and the scales' reliability was checked. Upon identifying sub-dimensions, various interdependent t-tests were employed to explore mean-level differences. Following this, structural equation modeling was utilized for model testing and assessing the indirect effects of life goals on well-being (i.e., life satisfaction and stress) through work-family interference. Finally, multiple-group comparison analysis was performed to investigate whether the effect on well-being differed across cultural, gender, and self-control groups. All control variables were included in these analyses, which were conducted using SPSS and AMOS 29.

3.3.2. Participants

This study was conducted with white-collar workers living in Türkiye and Germany. A total of 1252 participants were reached, with 662 residing in Germany and 590 in Türkiye. The gender distribution includes 599 (47.8%) women, 650 (51.9%) men, and 3 (0.2%) individuals identifying as diverse. Gender distributions are equal in both countries. The birth year of the oldest person in the dataset is 1955, while the youngest is 2005. The number of individuals with at least a bachelor's degree in the German sample is 268 (40.5%), and in the Turkish sample is 497 (84.2%).

According to the country-specific household income deciles, it is observed that the median monthly household income for German participants corresponds to the 7th highest

income bracket with a range from $\in 3,321$ to $\notin 3,880$. In contrast, the median monthly household income for Turkish participants aligns with the 8th highest income bracket and ranges from $\pounds 39,000$ (~ $\notin 1,300$) to $\pounds 45,999$ (~ $\notin 1,533$).

Half of the participants (n = 627) are married. Parental status is evenly distributed, with 605 individuals (48.3%) having at least one child. A large majority of the participants (n = 980, 78.3%) work full-time. While 122 participants (9.8%) work from home every day, 497 (39.8%) state that they never work from home. Most participants' spouses/partners (n = 647, 80%) are currently employed, with a large portion (n = 494, 76.5%) working full-time. The number of spouses/partners who consistently work from home is 73 (11.3%), while the number of those who never work from home is 286 (44.3%).

3.3.3. Measures

Life Goals: The importance assigned to goals was measured using the GOALS questionnaire developed by Pöhlmann and Brunstein (1997; for more details, see Pöhlmann et al., 2010). Participants rated the 24 goal items based on importance by responding to the question, "How important is it for you to reach this goal in your lifetime?": *power* (e.g., "I would like to be able to exert influence."), *achievement* (e.g., "I would like to continuously improve myself."), *variation* (e.g., "I would like to have adventures, live an adventurous life."), *intimacy* (e.g., "I would like to give affection and love."), *affiliation* (e.g., "I would like to be friends with many people."), and *altruism* (e.g., "I would like to help other people who are in need.") with four items each. All items were answered on a 5-point Likert scale, ranging from 1 (not important) to 5 (very important). Confirmatory factor analyses confirmed the six-factor structure both in the total sample and across different cultural groups (results available on request). As previously mentioned in the literature (Pöhlmann, 2001), the first three subdimensions represent agentic goals, while the last three represent communal goals according to Bakan's taxonomy (Bakan, 1966). Based on this, in the current study, the first

three dimensions were loaded onto the agentic goals second-order factor, and the last three dimensions were loaded onto the communal goals second-order factor. The results indicated no significant statistical difference between using a first-order or second-order factorial model, demonstrating robustness in the factorial structure. The Cronbach's alpha was 0.91 for agentic goals and 0.88 for communal goals.

Work-Family Interference: The interference between work and family/private life domains was assessed using a scale that measured bidirectional inter-role conflicts: work-to-family interference (WtFI) and family-to-work interference (FtWI) concerning time and strain, developed by Carlson et al. (2000). The German translation of the survey was conducted by Wolff and Höge (2011), and the Turkish translation by Erdoğan (2009). The scale consisted of twelve items, with six measuring WtFI and six measuring FtWI. Responses were collected on a 5-point Likert scale, ranging from 1 (not at all) to 5 (absolutely). An example item for WtFI was "My work keeps me from my family/private activities more than I would like," while for FtWI, it was "The time I spend on family/private responsibilities often interferes with my work responsibilities." Confirmatory factor analyses were conducted before hypothesis testing for the entire sample and different cultural groups, confirming that WtFI and FtWI are distinct constructs (results available on request). The scores for WtFI and FtWI were calculated by summing up the scores of the six domain-specific items and dividing them by six. The Cronbach's alpha was 0.91 for WtFI and 0.89 for FtWI.

Well-Being: Well-being was measured using a single item for each life satisfaction and stress. The question of life satisfaction was phrased as " All in all, how satisfied are you with your life at the moment?" Participants responded on an 11-point Likert scale ranging from 1 (very dissatisfied) to 11 (very satisfied). Stress, on the other hand, was assessed using a one-item developed by Elo et al. (2003), with the question, "Stress means a situation in which people feel tense, restless, nervous or anxious or are unable to sleep at night because their mind

is troubled all the time. Do you feel this kind of stress these days?" Participants responded on a 5-point Likert scale ranging from 1 (not at all) to 5 (very much).

Self-Control: Self-control was measured based on three single-item questions used by Wolff et al. (2022). Participants were asked three single-item questions related to self-control ("How much self-control do you have?"), self-discipline ("How much self-discipline do you have?") and willpower ("How much willpower do you have?"). They responded on an 11-point Likert scale ranging from 1 (not at all) to 11 (very much). Confirmatory factor analyses demonstrated that these three items converged into a single factor (results available on request). Cronbach's alpha coefficient was calculated as 0.81. A self-control score was obtained by averaging the scores of the three items. Subsequently, the median score for the entire sample was determined. Accordingly, participants scoring below the median were coded as the low self-control group, while those scoring above the median were coded as the high self-control group.

Culture and Gender: Culture was assessed by asking participants, "Of which country or countries are you a citizen?" The response options included Türkiye, Germany, or another country. Gender, on the other hand, was measured by asking participants, "Which of the following describes your gender identity?" Participants responded as female, male, or diverse. Given that the focus of the current study is on how different cultures may shape well-being for various gender groups, we created four interaction variables by multiplying cultures with gender: German-female, German-male, Turkish-female, and Turkish-male.

Confounding variables: We used employment status (1 = full-time employment, 0 = all other), marital status (1 = married, 0 = all other), and having children (1 = yes, 0 = no children) as control variables. We included employment status, marital status, and children because these factors can confound the relationship between life goals, work-family interference, and well-

being. Existing research indicates that married people tend to report higher life satisfaction compared to those who are divorced or have never been married (Helliwell, 2003; Stahnke & Cooley, 2021). Moreover, extensive work hours and the responsibility of caring for a child at home can elevate both work and family demands, potentially leading to increased levels of stress and consequently exerting a negative impact on overall well-being (Byron, 2005; Michel et al., 2011).

3.4. Results

3.4.1. Descriptive Statistics

Descriptive statistics and the differences in mean scores, both as directly observed variables and as latent factors, are significant and can be examined from Tables A3.1 and A3.2, shared in the appendices. The correlations between the main variables—agentic life goals, communal life goals, WtFI, FtWI, life satisfaction, and stress—are presented in Table A3.3 in the appendix.

3.4.2. Mediation Testing

We conducted mediation analyses to test our hypotheses. In these analyses, we included control variables in addition to agentic goals, which represent power, achievement, variation, and WtFI, as well as components of well-being like satisfaction and stress (Figure 3.2). The model fit the data well, indicating a good match (χ^2/df = 3.347; RMSEA= .043; CFI = .967; GFI = .950). We conducted the bootstrap analysis with 10000 samples using a 95% Bias-Corrected (BC) confidence interval. The coefficients and confidence intervals for indirect, direct, and total effect paths are presented in Appendix Table A3.4. As a result, we found that agentic goals were positively linked to higher levels of life satisfaction and stress; H1a is supported, and H1b is not supported. Moreover, it was clear that the indirect relationships between agentic goals and satisfaction (β = -.120; *p* < .001; 95% CI: -.155, -.089), as well as stress (β = .163; *p* < .001;

95% CI: .123, .205), through WtFI, were significant. Therefore, both H2a and H2b are supported.

Figure 3.2

The Relationship between Agentic Goals, WtFI, Life Satisfaction, and Stress



Note. Solid lines represent direct relationships, while dotted lines represent indirect relationships. Due to the study's cross-sectional design, no information could be derived about the causality of the hypothesized relationships, and the arrows in the figure should be analyzed with caution.

We followed a similar approach for the scenario where communal goals were considered as the independent variable (Figure 3.3). In this model, we included control variables alongside communal goals, which represent affiliation, intimacy, altruism, and FtWI, as well as components of well-being like life satisfaction and stress. The model fit the data reasonably well (χ^2/df = 4.961; RMSEA= .056; CFI = .933; GFI = .928). We conducted the bootstrap analysis with 10000 samples using a 95% Bias-Corrected (BC) confidence interval. The coefficients and confidence intervals for indirect, direct, and total effect paths are presented in Appendix Table A3.5. In this context, we found that communal goals were positively associated with higher levels of life satisfaction and stress; H1c is supported, and H1d is not supported. Moreover, it was evident that the indirect relationships between communal goals and satisfaction ($\beta = -.053$; p < .001; 95% CI: -.080, -.031), as well as stress ($\beta = .079$; p < .001; 95% CI: .048, .113) through FtWI, were significant. Therefore, both H2c and H2d are supported.

Figure 3.3

The Relationship between Communal Goals, FtWI, Life Satisfaction, and Stress



Note. Solid lines represent direct relationships, while dotted lines represent indirect relationships. Due to the study's cross-sectional design, no information could be derived about the causality of the hypothesized relationships, and the arrows in the figure should be analyzed with caution.

3.4.3. Group Comparisons for Self-Control Levels

We also examined how different levels of self-control interacted with the main variables to compare the groups. We looked at the low self-control group and the high self-control group separately, studying their differences in agentic goals, WtFI, life satisfaction, and stress. The model showed a good fit with the data, indicating a strong match (χ^2/df = 2.248; RMSEA= .032; CFI = .965; GFI = .936). The bootstrapping results supported the mediation relationship. For the low self-control group, the indirect relationships between agentic goals with life satisfaction (β = -.119; *p* < .001; 95% CI: -.169, -.078) and stress (β = .152; *p* < .001; 95% CI: .103, .208) through WtFI were significant (see Table A3.6 in the Appendix). Similarly, for the high self-control group, the indirect relationships between agentic goals with satisfaction (β = -.126; *p* < .001; 95% CI: -.179, -.082) and stress (β = .185; *p* < .001; 95% CI: .126, .247) through WtFI were also significant (see Table A3.7 in the Appendix).

For the low self-control group, the links between agentic goals and life satisfaction as well as stress became insignificant ($\beta = .06$; p = .182; $\beta = .07$; p = .138). For the high self-control group, connections between agentic goals and life satisfaction ($\beta = .16$; p < .001) as well as stress ($\beta = .09$; p < .05) remained significant. For both groups, the connections between agentic goals and WtFI ($\beta = .31$; p < .001, $\beta = .32$; p < .001), WtFI and satisfaction ($\beta = -.38$; p < .001, $\beta = -.39$; p < .001), and WtFI and stress ($\beta = .49$; p < .001, $\beta = .57$; p < .001) remained significant.

To examine the significance of group differences in these relationships, we conducted multi-group comparisons. We equalized the paths in the model for both groups and compared them to the unconstrained model. The results showed that there was no significant difference between the low self-control and high self-control groups when we controlled for being married, having children, and working full-time. Therefore, H3a, H3b, and H3c are not supported.

We also conducted the same test to determine the relationship between communal goals, FtWI, life satisfaction, and stress. The model fit the data well, indicating a strong match ($\chi^2/df=$ 2.995; RMSEA= .040; CFI = .931; GFI = .916). The bootstrapping results confirmed the mediation relationship. For the low self-control group, the indirect relationships between communal goals with satisfaction ($\beta = -.051$; p < .001; 95% CI: -.088, -.020) and stress ($\beta =$.062; p < .001; 95% CI: .025, .103) through FtWI were significant (see Table A3.8 in the Appendix). Similarly, for the high self-control group, the indirect relationships between communal goals with satisfaction ($\beta = -.065$; p < .001; 95% CI: -.116, -.031) and stress ($\beta =$.124; p < .001; 95% CI: .073, .190) through FtWI were also significant (see Table A3.9 in the Appendix).

For the low self-control group, the connection between communal goals and satisfaction $(\beta = .07; p = .138)$ was no longer significant, while it was significant for the high self-control group $(\beta = .17; p < .01)$. For the low self-control group, the link between communal goals and stress $(\beta = .14; p < .01)$ was significant while it became marginally significant for the high self-control group $(\beta = .11; p = .057)$. For both groups, the links between communal goals and FtWI $(\beta = .16; p < .001, \beta = .29; p < .001)$, FtWI and satisfaction $(\beta = -.31; p < .001, \beta = -.23; p < .001)$, and FtWI and stress $(\beta = .38; p < .001, \beta = .44; p < .001)$ were significant.

Multi-group comparisons were performed to assess the significance of group differences in these relationships. The paths in the model were equalized for both groups and compared to the unconstrained model. The results revealed a marginal significance in the relationship between communal goals and satisfaction ($\chi^2(1)=3.259$; p=.071); H3d is supported. However, there was no significant difference between low and high self-control groups regarding the relationship between communal goals and stress; H3e is not supported. Similarly, there was a significant distinction between the low self-control and high self-control groups in terms of the relationship between communal goals and FtWI ($\chi^2(1)=5.743$; p < .05). However, contrary to expectations, the relationship between communal goals and FtWI was stronger among people with high self-control than those with low self-control. Therefore, H3f is not supported. Additionally, there was a marginal significance in the relationship between FtWI and satisfaction ($\chi^2 = 2.979$; p = .084).

3.4.4. Group Comparisons for Culture and Gender Interactions

To compare the groups, we analyzed how nationality and gender interacted with the main variables. We looked at German and Turkish women separately, examining their differences in agentic goals, WtFI, and well-being (specifically, life satisfaction and stress). The model fit the data well, indicating a good match ($\chi^2/df=1.764$; RMSEA= .036; CFI = .948; GFI = .900). The bootstrapping results confirmed the mediation relationship. For German women, the indirect relationships between agentic goals with satisfaction ($\beta = -.156$; p < .001; 95% CI: -.262, -.076) and stress ($\beta = .168$; p < .001; 95% CI: .083, .270) through WtFI were significant (see Table A3.10 in the Appendix). Similarly, for Turkish women, the indirect relationships between agentic goals with satisfaction ($\beta = -.076$; p < .05; 95% CI: -.147, -.019) and stress ($\beta = .095$; p < .01; 95% CI: .022, .166) through WtFI were also significant (see Table A3.11 in the Appendix).

In both groups, except for the relationships between agentic goals, life satisfaction, and stress, all other connections in the main model remained significant. For German women, agentic goals were no longer linked to stress ($\beta = .04$; p = .585), while for Turkish women, this link remained significant ($\beta = .14$; p < .05). For Turkish women, agentic goals were no longer associated with life satisfaction ($\beta = .03$; p = .658), while for German women, this relationship still held ($\beta = .34$; p < .001). For German and Turkish women, the connections between agentic goals and WtFI ($\beta = .32$; p < .001, $\beta = .20$; p < .05), WtFI and life satisfaction ($\beta = .48$; p < .001, $\beta = -.38$; p < .001), and WtFI and stress ($\beta = .52$; p < .001, $\beta = .47$; p < .001) remained significant.

To assess the significance of group differences in these relationships, we conducted multi-group comparisons. We equalized the paths in the model for both groups and compared them to the unconstrained model. The results showed that German and Turkish women differed significantly in terms of the relationships between agentic goals and life satisfaction ($\chi^2(1)$ = 4.216; *p* = .040), H4a, but not H4b, is supported. However, there was no significant difference in the relationship between WtFI and any of the well-being dimensions. Therefore, both H4c and H4d are not supported.

The same analyses were repeated for German and Turkish male groups, considering communal goals, FtWI, life satisfaction, and stress. The model fit the data well (χ^2/df = 2.022; RMSEA= .040; CFI = .934; GFI = .899). The bootstrap results indicated that the mediation relationship could no longer be discussed for both German and Turkish men. Specifically, for German men, the indirect relationships between communal goals and satisfaction (β = -.017; *p* = .078; 95% CI: -.064, .002), as well as stress (β = .052; *p* = .110; 95% CI: -.013, .134) through FtWI were not significant(see Table A3.12 in the Appendix). For Turkish men, the indirect relationships between communal goals and satisfaction (β = -.016; *p* = .415; 95% CI: -.065, .026) as well as stress (β = .022; *p* = .434; 95% CI: -.036, .087) through FtWI were not significant (see Table A3.13 in the Appendix).

For both German and Turkish men, the relationship between communal goals and stress was no longer significant ($\beta = .05$; p = .493; $\beta = .09$; p = .188, respectively). However, communal goals were associated with higher levels of satisfaction ($\beta = .31$; p < .001, $\beta = .22$; p < .001; respectively) for both groups. The relationship between communal goals and FtWI was no longer significant for German men ($\beta = .12$; p = .122) and Turkish men ($\beta = .05$; p = .442). It was observed that FtWI had a significant dampening effect on satisfaction ($\beta = .14$; p < .05; $\beta = -.29$; p < .001), while it increased stress levels in the opposite direction ($\beta = .43$; p < .001; $\beta = .40$; p < .001) for both German men and Turkish men. Multi-group comparisons were performed to assess the significance of group differences in these relationships. The paths in the model were equalized for both groups, and these models were compared to the unconstrained model. There was no significant difference between German men and Turkish men regarding the relationship between communal goals and any of the well-being dimensions; neither H4e nor H4f were supported. However, the results showed that German and Turkish men differed significantly in terms of the relationships between FtWI and life satisfaction ($\chi^2(1)=5.741$; p < .05). However, contrary to expectations, the negative impact of FtWI on life satisfaction was stronger among Turkish men than among German men. Therefore, both H4g and H4h are not supported.

3.5. Discussion

In this study, we examined the effects of life goals on life satisfaction and stress, assessing their implications for well-being. We evaluated the mediating role of work-family interference as a negative impact of goal endorsements. It was also assumed that this effect would be moderated by both culture-gender interaction and self-control. In the current study, it was hypothesized that there are two facets to the narrative. Initially, it was posited that the direct correlation between life goals and well-being would yield positive outcomes, manifesting in increased life satisfaction and decreased stress. Conversely, the indirect relationship, mediated by work-family interference, was expected to exhibit a negative impact on well-being.

The analysis of the direct effect revealed that as individuals place increasing importance on a specific goal, life satisfaction levels increase, regardless of whether they are agentic or communal life goals. However, contrary to expectations, an increase in the importance attached to these life goals has also brought about increased stress levels. Within broader literature, numerous studies have explored the direct impact of diverse life goals on well-being. Notably, the Big Two framework, encompassing both agentic and communal goals, has been linked to well-being (Saragovi et al., 2002; Sheldon & Cooper, 2008). Confirming self-regulation theories, it seems that having goals brings about a sense of fulfillment and meaning in life, resulting in life satisfaction (Emmons, 2003). Although our results show that increasing the importance placed on agentic or communal goals leads to higher stress levels, these findings align with the literature. According to the cognitive activation theory of stress, life goals can serve as activators of stress. Short-term stress responses can help individuals cope with challenges and restore homeostasis (Ursin & Eriksen, 2010). On the other hand, there could be "too much of a good thing," meaning the demands associated with these goals, perhaps driven by a desire for attainment, also bring about stress (Toth et al., 2018). The more life goals indicating deficiencies in life receive importance, the more individuals may become stressed (Mayser et al., 2008). Therefore, life goals may be stressful, but they can simultaneously enhance life satisfaction, especially when they are perceived as achieved.

Existing literature lacks investigation into the relationship between goal endorsement and well-being across different domains, including the interference mechanism. Our study fills this gap by examining the direct effect of life goals on well-being through work-family interference, highlighting the unfavorable aspect of the story. In the current study, by identifying agentic goals with personal achievement and communal goals with relationships centered, we found results that support matching-domain effects in the literature (Amstad et al., 2011; Shockley & Singla, 2011). Increased importance on agentic goals leads to more WtFI, decreasing life satisfaction, and increasing stress. Similarly, emphasis on communal goals increases FtWI, further lowering life satisfaction and increasing stress indirectly. Negative experiences (WtFI or FtWI) tend to be based on areas with an excessive emphasis on a particular focus (agentic or communal life goals). This interaction with demands in that specific domain has repercussions in other aspects of life, leading to adverse outcomes such as reduced well-being, diminished life satisfaction, and increased stress. Our study demonstrated the existence of

matching-domain effects, and further research is needed to discuss the cross-domain effects of life goals on work-family interference and well-being.

It is found that as the importance of agentic goals increases, individuals invest more in the work domain, leaving them vulnerable to family demands, resulting in more WtFI, stress, and lower life satisfaction. Similarly, an emphasis on communal goals leads individuals to invest more in the family domain, causing FtWI, stress and decreased life satisfaction. These findings align with previous research indicating that higher income (the goal of making more money can be considered as an agentic goal) correlates with more WtFI (Rubenstein et al., 2022) and being ostracized by loved ones (individuals who do not want to feel excluded by their families are possibly who pursue communal goals) correlates with more FtWI (Babalola et al., 2020). These results also align with the boundary theory, which states that the more someone identifies with a role, the more they blend it with other roles (Ashforth et al., 2000). Similarly, Kossek et al. (2012) found that individuals with a greater work focus experience more WtFI, while those with a stronger family focus face more FtWI. Additionally, research indicates that individuals' perceptions of the usage of Information and Communication Technology are influenced by their motivations (Schöllbauer et al., 2021). People who continuously strive for career advancement tend to use Information and Communication Technology more during non-work hours, leading to increased WtFI (Schlachter et al., 2018). Conversely, we can extend these results by stating that those with communal goals may voluntarily use non-work-related technology during work hours, resulting in more FtWI. These situations also offer individuals a perspective that allows them to choose their struggles freely. Accordingly, giving more importance to goals in a specific domain opens the way for interaction with other domains of life, but this interaction makes individuals vulnerable by consuming their resources, negatively affecting their well-being. While goal endorsement ordinarily positively influences life satisfaction, an additional negative impact unfolds through the mediation of work-family interference. Moreover, work-family interference plays a

mediating role in amplifying the negative effect of goal endorsement on stress.

The current research also examined whether the indirect effect of life goals on well-being through work-family interference varies according to different self-control levels, which is called the key resources concept of the Work-Home Resources model (ten Brummelhuis & Bakker, 2012). Accordingly, when comparing the low self-control group with the high self-control group, no significant difference was found in the model that examined the relationship between agentic goals, WtFI, and well-being (i.e., life satisfaction and stress). Individuals who value agentic goals may be perceived as having a heightened sense of self-control. In other words, agentic life goals may already manifest themselves through a greater level of self-focus (Garcia et al., 2015; Saragovi et al., 1997). This could explain the lack of significant differences between groups with low and high levels of self-control, as the degree of importance attributed to agentic goals may be the primary factor influencing the relationships with WtFI and well-being dimensions.

While examining the model on communal goals, FtWI, and well-being (life satisfaction and stress), we discovered unexpected differences. Surprisingly, the high self-control group, when emphasizing communal goals, experienced more FtWI compared to the low self-control group. Initially, we assumed that individuals with high self-control would face less FtWI when pursuing communal goals due to their efficient resource allocation. This expectation was consistent with the literature, as Kossek et al. (2012) noted that individuals with high control profiles, regardless of their degree of boundary-crossing behaviors and the array of identity centralities, were more likely to report positive work-family outcomes compared to those with low control profiles. However, it seems this relationship goes beyond resource management, involving factors like self-confidence or over-commitment. Those with high self-control skills tend to take on more responsibility in pursuing goals (Koval et al., 2015), potentially leading to underestimating demands and overestimating capacities (Hetland et al., 2012). Paradoxically, an excessive dependence on one's own abilities may lead to elevated levels of FtWI. Consistent with existing literature, an emphasis on communal goals correlated with higher life satisfaction for the high self-control group compared to the low self-control group. Previous research indicates that individuals with high self-control skills are more successful in goal pursuit, leading to better health and well-being (de Ridder et al., 2012; Tangney et al., 2004). Additionally, self-control skills positively contribute to stress-coping abilities (Boals et al., 2011; Folkman, 1984). In line with this, our study found that FtWI has a less detrimental effect on life satisfaction for the high self-control group compared to the low self-control group. The results indicate that self-control does not suppress the level of cross-domain interference associated with life goals but rather mitigates the adverse outcomes of the experienced interference.

Finally, the present research explores how cultural and gender roles influence the relationship between life goals, work-family interference, and well-being in Germany and Türkiye, using the macro resources concept of the Work-Home Resources model (ten Brummelhuis & Bakker, 2012). Therefore, it was investigated if these relationships are significantly different for men and women in Germany and Türkiye by considering cultural values as one of the macro resources. Firstly, the study examined whether the importance placed on agentic goals and their impact on well-being (specifically life satisfaction and stress) through WtFI exhibited disparities among women in Germany and Türkiye. The findings revealed that as German women were placing greater importance on agentic goals, they exhibited higher life satisfaction than their Turkish counterparts. This could be related to appreciating individualism, self-focus, mastery, and power in German society. Upon examining subgroups, it appears that the effect of agentic goal endorsement on stress disappears as German women align themselves with social expectations. In Turkish women, however,

agentic goal endorsement increases stress while its impact on life satisfaction diminishes. This suggests that not pursuing goals aligned with societal expectations suppresses positive effects. Consequently, consistent with previous literature (Myers & Diener, 1995; Sedikides et al., 2003), German women meet societal expectations by pursuing their agentic goals in a manner valued by their community, which significantly positively impacts their life satisfaction. Significant differences were not found in the impact of WtFI on various dimensions of well-being among women living in different cultures. This outcome suggests that, regardless of cultural context, work life interference with family life is uniformly detrimental to the well-being of individuals, as supported by the findings of a previous study (Kusnierz et al., 2022).

Secondly, similar analyses were conducted to determine whether the impact of the importance given to communal goals on FtWI and well-being (i.e., life satisfaction and stress) differed for men in Germany and Türkiye. Surprisingly, the findings were contrary to our expectations: Turkish men, as they faced higher levels of FtWI, their life satisfaction level was negatively affected more than their German counterparts. This phenomenon can be explained by insights from previous studies, indicating that close family ties can act more as a demand than a source, potentially exerting a detrimental impact on life satisfaction (Allen et al., 2015; Kalliath et al., 2017). In societies characterized by pronounced gender inequalities, male individuals who perceive their role in providing for the family may experience a more significant impact on life satisfaction in the event of interference between family and work compared to their counterparts in societies with greater gender equality (i.e., Allen et al., 2015). Significant differences were not found in the effect of pursuing communal goals on well-being among men living in different cultures. Men who pursue these communal life goals do not experience increased stress, and their life satisfaction is positively influenced. These results indicate that valuing communal goals, such as cooperation, caring, and fostering relationships,

is directly crucial for well-being regardless of cultural background, consistent with the results of a meta-analysis study (Le et al., 2018).

Accordingly, the negative impact that emerges during the goal endorsement, which we view as the unfavorable aspect of the study, appears to affect individuals with different characteristics consistently. Interestingly, in relationships where group differences are significant, it has been observed that the dependent variable is mostly life satisfaction, not stress. These results are consistent with the meta-analysis, which found that the association between goal pursuit and well-being is larger when well-being is measured with positive indicators, as opposed to ill-being (negative indicators; Klug & Maier, 2015).

3.5.1. Limitations

It is important to consider several limitations of the study. Firstly, using a cross-sectional study design limits the comprehensiveness of data collection, as it captures a static snapshot of data and does not account for changes over time, thus preventing the establishment of causal relationships and only allowing the reporting of associations between variables. It can also be argued that the data from German participants is more representative than the data from Turkish participants. This assertion is based on the observation that a majority of Turkish participants in the present research possess higher education degrees and higher incomes and reside in urban areas, in contrast to the broader demographic of Turkish citizens. Additionally, using a one-item scale for measurement may limit the depth and accuracy of data collection. Furthermore, the study only focused on domain-unspecific outcomes, which may restrict the overall understanding of its implications on domain-specific outcomes. It is also essential to recognize that further research is imperative to fully unravel the cross-domain effects of life goals on work-family interference and well-being. While this study provides a valuable starting point, its limitations highlight the need for continued exploration in future research.

3.5.2. Contributions

This study contributes significantly to the literature by integrating the Work-Home Resources model with self-regulation theory, thereby bridging gaps in understanding how various life goals impact well-being. It establishes a crucial framework for discerning the diverse effects of different life goals on well-being, highlighting both their positive contributions and potential negative consequences mediated by work-family interference. This dual perspective enhances comprehension of the complexity inherent in goal pursuit. The research particularly examines how mismatches between individuals' goals and societal gender role expectations can detrimentally affect well-being, compounded by the tendency of individuals with high selfcontrol to underestimate demands. Practically, the study underscores the importance of awareness of potential stress and work-family interference of individuals pursuing their life goals. It advocates techniques from Hirschi et al.'s (2019) theoretical action regulation model, which can help them pursue their goals without interfering with work and family domains. Organizational implications include fostering supportive environments with flexible work arrangements and childcare resources, which are crucial to increasing both contextual and personal resources of employees. Policymakers are urged to promote social awareness and provide support for both individual and organizational well-being in the contemporary era, where all genders value both agentic and communal goals (Galinsky et al., 2013).

Chapter 4: Examining the Impact of a Self-Regulation Intervention on Work-to-Family Interference

Abstract

In the quest to improve employee well-being, the ability of self-control to balance work and family dynamics has become crucial, especially in times of skill shortages and increased stress among employees. Drawing on self-control theory (Muraven & Baumeister, 2000), we investigated the impact of an online intervention addressing self-regulation by means of action planning and action control. Self-control was hypothesized to lead to less work-to-family interference (WtFI) and enhanced well-being. The randomized controlled trial (RCT) included N = 662 white-collar employees with two measurement points over three months. Various analyses, including Multivariate Analysis of Variance and spline regression, were employed to evaluate the intervention's effectiveness. To test whether the intervention effects were consistent across all individuals, participants were categorized into Improvers (those whose self-control increased over time), Stables (those who maintained their existing self-control abilities), and Reducers (those whose self-control declined over time). However, the results indicated no significant differences in WtFI between the intervention and control groups, meaning there was no overall intervention effect. Further analysis revealed that the degree of self-control change played a crucial role. Specifically, participants in the intervention group who experienced moderate improvements in self-control initially reported an increase in WtFI. Conversely, those with higher increases in self-control showed a decrease in WtFI overtime, indicating long-term benefits. Reducers and Stables exhibited no significant changes in WtFI. These findings may highlight that while self-control improvements can be beneficial, they may initially increase WtFI before yielding positive outcomes. Such subgroups within the intervention group should be taken into account in the future to prevent such negative effects.

Keywords

Self-control, self-regulation, stress, well-being, work-to-family interference

4.1. Introduction

In the quest to improve employee well-being, the ability of self-control to balance work and family dynamics has emerged as a critical area of focus. In today's business landscape, technological advancements have significantly increased the integration of work into daily life, allowing for greater flexibility and efficiency. However, this increased accessibility also presents challenges to maintaining well-being, such as difficulty in setting boundaries between work and personal life. This study aims to test the hypothesis that a self-regulation intervention can help reduce work-to-family interference by means of action planning and action control.

Work-to-family interference (WtFI) occurs when work demands negatively impact family responsibilities, causing stress and reduced well-being (Greenhaus & Beutell, 1985). For example, an employee may struggle to attend their child's school event due to work deadlines, leading to feelings of guilt and frustration. Such interference, in turn, limits well-being and increases stress, which are major consequences of WtFI (Amstad et al., 2011; Kinnunen et al., 2006).

Self-control, the ability to regulate one's thoughts, emotions, and behaviors in the face of temptations and distractions (Baumeister et al., 1994), is increasingly recognized as a valuable asset in the workplace. High self-control enables employees to manage their resources and time better (Adams & Jex, 1999) and set effective boundaries between work and personal life (Rothbard & Ollier-Malaterre, 2016). This was shown to help to cope with stress (Tangney et al., 2004) and maintain beneficial routines (Muraven et al., 1999). The present study investigates the impact of a self-regulation intervention designed to enhance self-control and its subsequent positive effects on reducing WtFI.

Drawing on the *strength model of self-control* (Muraven & Baumeister, 2000), which posits that self-control can be strengthened through practice, the current intervention incorporates action planning and action control strategies. Action planning involves specifying when and where behaviors will be performed, which helps convert behavioral intentions into real activities. Implementation intentions (Gollwitzer et al., 2004), a subset of *action control*, involve planning responses to anticipated barriers, helping to overcome obstacles to action. Action control, particularly through self-monitoring, involves continuously observing and evaluating one's actions to ensure they align with action plans. These strategies are critical components of the Behavior Change Techniques (BCT) taxonomy (Michie et al., 2013), providing a structured framework for individuals to set goals, monitor progress, and establish consistent behaviors.

The relationship between WtFI, limited well-being, and stress underscores the necessity for an intervention that helps employees manage these conflicting demands more effectively (Richardson, 2017). This article focuses on time-based and strain-based interference, as its relationship with self-control is expected to be more pronounced. Research indicates that high self-control aids in time management and prioritization, helping to mitigate time-based interference (Gröpel & Kuhl, 2006; Mellner et al., 2014). It has also been shown that selfcontrol alleviates strain-based interference by enhancing stress management (Cheung & Tang, 2011; Wei et al., 2022). Training programs focused on developing self-control can equip employees with the tools needed to regulate their responses to stressors and maintain a healthier work-life balance (Kiburz et al., 2017). Research evaluating a program using systematic selfmonitoring and reflection strategies also found improved employee well-being and reduced sick leave (Krampen, 2010).

This study aims to test the nuanced effects of a self-regulation intervention specifically designed to promote self-control and reduce WtFI, incorporating elements of action control and planning. The ultimate goal is to optimize these programs to benefit employees. The impact of a self-regulation intervention was assessed by comparing it to a waitlist control group. To fully comprehend the intervention's effectiveness, we applied two complementary analytical

approaches to provide a deeper understanding of the outcomes. First, employees were categorized based on their self-control trajectories: Improvers, Stables, and Reducers. Improvers developed an increase in self-control, Stables maintained existing self-control abilities, and Reducers experienced a decline in self-control. This categorization allowed us to distinguish between changes attributable to the intervention itself and the inherent fluctuations in self-control levels that might occur naturally or due to external circumstances. By differentiating these groups, we were able to isolate the unique effects of the intervention on self-control, providing a clearer picture of its impact across diverse trajectories.

In the second analytical approach, we addressed the fact that the data did not follow a simple linear pattern. We anticipated that self-control would not influence WtFI uniformly across all levels but instead would exhibit varying effects depending on certain critical thresholds. To capture these nuances, we used spline regression, which allowed us to examine how the relationship between self-control changes and WtFI shifted at different points along the self-control spectrum. The categorical analysis offers a broad view of the overall trends, showing how distinct groups of employees responded to the intervention. However, the spline regression provided a more granular, detailed analysis, identifying specific threshold points where the effects of self-control changes became more pronounced or diminished. This dual approach allowed us to understand not only the general effectiveness of the intervention but also the specific conditions under which its impact was most significant.

We hypothesized that participants in the intervention group would develop more selfcontrol than those in the WLCG (T1-T2 = intervention check) and that this increase would result in less WtFI. We also hypothesized that the effects would be more pronounced among individuals who improved their self-control compared to those whose self-control remained stable or declined. The differences between Improvers and Stables were expected to be smaller than those between Improvers and Reducers.

4.2. Method

4.2.1. Participants and Procedure

The participants in this study were white-collar employees residing in Germany. They were recruited through a German-based survey company (Norstat Deutschland GmbH), which distributed the online questionnaire via relevant links, utilizing a convenience sampling technique. The online questionnaire was prepared using the UNIPARK system. All measures that had not previously been translated into German were translated by the researcher using the back-translation method (Brislin, 1970).

Participants were informed via email about the details of the study and received the baseline questionnaires. They gave their informed consent at the start of the online questionnaire by explicitly agreeing by hitting a "continue" button. Without this consent, they were unable to complete the questionnaire and were classified as dropouts. Occasional email reminders were sent to encourage the completion of the questionnaires or the intervention.

Initially, N = 662 employees were contacted at the baseline stage (T1). These participants were then randomly divided into two separate groups by the survey company: an intervention group and a waitlist control group, each consisting of N = 331 individuals. Approximately one month after completing the baseline questionnaires (see Figure 4.1 for the study flow chart), the two-week training content consisting of 2 hours in total was sent to the intervention group. One month after completing the training, the first post-questionnaire was sent to both the intervention and control groups. At this stage (T2), n = 183 participants from the intervention group and n = 258 individuals from the control group, totaling n = 441 participants, completed the questionnaires. The study continued with the waitlist control group.

Figure 4.1

Flow Chart and Study Design



The demographic characteristics of the intervention group (n = 331) and the control group (n = 331) were analyzed (see Table 4.1 for the descriptives). The proportion of married or living in partnership individuals is similar in both groups, with 64% (n = 211) in the intervention group and 65% (n = 215) in the control group. Full-time employees constitute the majority in both groups, with 77% (n = 255) in the intervention group and 83% (n = 274) in the control group at

54% (n = 177), while men are more prevalent in the control group at 62% (n = 204). The grouping is done to ensure an equal number of individuals in each group has shown that the majority of participants are in their 40s, with those aged 40-48 comprising 54% (n=180) of the intervention group and 50% (n = 165) of the control group. Regarding educational status, individuals with a university degree are 38% (n = 125) in the intervention group and 40% (n = 133) in the control group.

Table 4.1

		Interve	ntion	Cont	rol	Dropo	outs
Marital Status	Married/Living in A	211	64%	215	65%	141	64%
	Partnership						
	Other	120	36%	116	35%	80	36%
Employment	Full-Time	255	77%	274	83%	173	78%
Status							
	Part-Time	76	23%	57	17%	48	22%
Gender	Male	154	47%	204	62%	105	48%
	Female	177	54%	127	38%	116	53%
Age	Younger Than 31 Years	5	1.5%	11	3.3%	5	2.3%
	Old						
	31-40 Years	83	25.1%	98	29.6%	63	28.5%
	41-50 Years	220	66.5%	209	63.1%	133	60.2%
	Older Than 51 Years Old	23	6.9%	13	3.9%	20	9%
Education	University Degree	125	38%	133	40%	76	34%
	Other	200	60%	192	58%	137	62%
	Not Reported	6	2%	6	2%	8	4%

Socio-Demographic Characteristics of The Participants: Number of Study Participants and Percentages Per Group

Note. Significant differences were only found in gender distribution between intervention and control groups, $\chi^2(1) = 15.207$, *p*<.001. There were no group differences regarding other demographic factors $\chi^2(1-5) = .411-6.553$, *p* = .07-.65.

Participants who dropped out from the intervention group were largely similar to those who completed the intervention and provided T2 data regarding their socio-demographic characteristics, $\chi^2(1-3) = 0.28$ to 6.95, p = .073 to .596, with one notable exception: a higher proportion of women dropped out compared to men, $\chi^2(1)=5.791$, p=.016. Additionally, baseline scale results showed that participants who dropped out had higher levels of stress compared to those who completed the T2 questionnaires, t(325.487)=-3.410, p < .001. In the control group, participants who dropped out did not differ significantly from those who completed the T2 questionnaires in terms of socio-demographic data, $\chi^2(1-3) = 0.04$ to 5.38, p= .146 to .841, and baseline scale results.

4.2.2. Self-Regulation Intervention

Participants underwent an intervention aimed at developing action planning and action control behaviors. This process included exercises designed to help participants set clear goals, monitor their progress, and establish consistent behaviors. By fostering these behaviors, the intervention aimed to reduce work-to-family interference (WtFI) and improve overall wellbeing.

The intervention consisted of a comprehensive booklet on self-control and an exercise plan with action-control guidelines designed by the researcher and integrated into the UNIPARK system. Relevant Behavior Change Techniques (BCTs) were selected in accordance with the BCT taxonomy developed by Michie et al. (2013). Participants received a document of about two pages that detailed the definition, importance, physiological characteristics, development, and potential depletion of self-control (Shaping knowledge [BCT 4]). Following this, they were instructed to perform action control exercises over approximately two weeks using a template provided by the researcher. In these exercises, participants planned how to pursue their goals by specifying what they wanted to achieve, where and when they would work on their goals, and creating if-then scenarios to anticipate potential obstacles (Gollwitzer et al., 2004). They were also encouraged to reflect on how they could improve their goal achievement strategies for the future (Goals and planning [BCT 1], Feedback and monitoring [BCT 2], Repetition and substitution [BCT 8]).

4.2.3. Measures

Self-control was evaluated using three single-item questions from Wolff et al. (2022). Participants answered three specific questions about self-control ("How much self-control do you have?"), self-discipline ("How much self-discipline do you have?"), and willpower ("How much willpower do you have?"). The score was calculated by averaging the three items. They responded on an 11-point Likert scale ranging from 1 (not at all) to 11 (very much). Cronbach's α was .80 at T1 and .79 at T2. Participants were categorized into three categories based on changes in their self-control levels from T1 to T2. Specifically, the change in self-control was calculated by subtracting the T1 self-control score from the T2 self-control score (T2 - T1). Participants with positive change scores were classified as "*Improvers*," indicating an increase in self-control. Those with a change score of between ±.50 were categorized as "*Stables*," indicating no change in self-control. Finally, participants with negative change scores were labeled as "*Reducers*," indicating a decrease in self-control. This classification allowed us to analyze the effects of the self-control training intervention across different trajectories of self-control change.

The *interference between work and family/private life* was measured using the time and strain subdimensions of the WtFI subscale developed by Carlson et al. (2000). The German version of the survey was translated by Wolff and Höge (2011). These subscales included three items for time-based and strain-based WtFI. Responses were given on a 5-point Likert scale, from 1 (not at all) to 5 (absolutely). The score was calculated by averaging the six items. Cronbach's α was .91 at T1 and .91 at T2.

Stress was evaluated using a single-item measure created by Elo et al. (2003): "Stress means a situation in which people feel tense, restless, nervous, or anxious or are unable to sleep

at night because their mind is troubled all the time. Do you feel this kind of stress these days?" Participants answered using a 5-point Likert scale, from 1 (not at all) to 5 (very much).

Socio-demographics were measured in terms of gender (1 = woman, 0 = man), marital status (1 = married/living in partnership, 0 = all other), employment status (1 = full-time employment, 0 = all other), and age (1 = younger than 31 years old, 2 = 31-40 years, 3 = 41-50 years, 4 = older than 51 years old).

4.2.4. Data Analysis

Due to the multiple repeated measurements in the study, some outcome variables had missing data. This occurred primarily because some participants discontinued their participation or chose not to answer some of the regular questions. To address this issue, both the Intention-to-Treat and Complete-Case (Per Protocol) approaches were used for analysis and comparison, yielding nearly identical results. Participants who dropped out were compared with those who provided T2 data using χ^2 -tests for categorical socio-demographic data and independent *t*-tests for T1 scales, with analyses conducted in SPSS version 29.

For the complete case analyses, the training effects on these variables were examined using Chi-Square tests, Repeated Measures ANOVA, Spline Regression, and *t*-Tests, all conducted in SPSS version 29. For the 183 participants from the intervention group and the 258 participants from the control group who provided both baseline data (T1) and post-intervention data (T2), missing data was less than 5%; hence the list-wise deletion method was used for the missing data.

To validate the analyses using an Intention-to-Treat approach, the same models were applied to multiply imputed datasets, including the entire sample of 662 white-collar employees who provided baseline data (331 in the intervention group and 331 in the control group). The Fully Conditional Specification method was chosen for imputation, conducted in SPSS version 29. In the analysis, the group variable was excluded from the predictor and imputation processes. The Predictive Mean Matching method was selected for imputation, and categorical variables were defined as factors and included in the predictor matrix. The number of imputed datasets was set to 10, following prominent guidelines (Rubin, 1987). Finally, the maximum number of iterations for each dataset was set to 10. Among the imputed datasets generated, the one most similar to the original dataset was selected for reporting the results. This approach was taken because SPSS does not provide pooled results for the relevant analyses.

4.3. Results

4.3.1. Per Protocol Analyses⁵

The mean scores of variables between the intervention and control group and intercorrelations of self-control, work-to-family dynamics, and well-being, as well as their change scores between T1 and T2, are reported in Table 4.2 and Table 4.3. Initial self-control was correlated with the initial work-to-family interference scores, r = -.12 (p < .01).

In general, there were differences between the control group and the intervention group, not only with the mean scores but also with the intercorrelations between variables. The baseline results indicated no significant differences between the groups regarding self-control at T1, t(439) = -0.106, p = .92, and WtFI at T1, t(439) = -1.428, p = .15. However, there was a significant difference in stress levels at T1, t(364.770) = -2.748, p < .01 with higher stress levels in the control group. The correlations were always larger for the IG than the CG, indicating more variability/variance and changes over time.

⁵ The results obtained using Intention-to-Treat analysis were similar to those obtained using per protocol analysis.

Table 4.2

Correlation Matrix

	WtFI_T1	SCON_T1	Stress_T1	Change_WtFI	Change_SCON	Change_Stress
WtFI_T1	-	12**	.51**	40**	.02	08*
SCON_T1	05/22**	-	13**	.09	44**	.01
Stress_T1	.45**/.58**	08/19**	-	09	.05	46**
Change_WtFI	39**/42**	.16*/.00	06/13	-	.01	.20**
Change_SCON	.05/05	50**/37**	.13*/07	08/.12	-	.00
Change_Stress	.01/20**	.03/02	39**/53**	.14*/.27*	09/.13	-

Note. ** p < .01. * p < .05. WtFI: Work-to-family interference. SCON: Self-Control. Above the diagonal: Both groups are together; below, the

first is the control group, and the second is the intervention group.

Table 4.3

Means and Standard Deviations

	Control	Intervention	Total
WtFI_T1	2.50	2.37	2.45
SD	.97	.93	.95
SCON_T1	7.84	7.82	7.83
SD	1.56	1.66	1.60
Stress_T1	2.98	2.68	2.85
SD	1.07	1.19	1.13
WtFI_T2	2.52	2.41	2.48
SD	.98	.95	.97
SCON_T2	7.84	7.64	7.76
SD	1.40	1.69	1.53
Stress_T2	3.02	2.82	2.93
SD	1.16	1.12	1.14

4.3.2. Self-Control Changes

Over time and with all study participants taken together, 135 individuals (30.6%) exhibited improved self-control skills, referred to as *Improvers*. Conversely, 143 individuals (32.4%) showed a decrease in self-control skills, termed *Reducers*. Meanwhile, 159 individuals (36.1%) maintained the same level of self-control skills, identified as *Stables*. In the intervention group, 48 individuals (26.2%) were identified as Improvers, compared to 87 individuals (33.7%) in the control group. The intervention group had 64 Reducers (35%), whereas the control group had 79 (30.6%). Additionally, there were 68 Stables (37.2%) in the intervention group and 91 (35.3%) in the control group. These descriptive differences were not

statistically significant: $\chi^2(2) = 2.87$; p = .24 in baseline and post-intervention scores (see appendix). Additionally, there were no significant differences concerning gender, age, marital status, employment status, or education level: $\chi^2(2-6) = 0.59$ to 4.59, p = .46 to .74.

Repeated measures ANOVA was conducted to examine the effect of time (T1 and T2), group (control and intervention), and self-control changes (Improvers, Stables, and Reducers) on WtFI. The within-subjects factor was time, and the between-subjects factors were group and self-control changes. Since there was a difference between groups in the baseline scores on stress at T1, we treated stress at T1 and self-control at T1 as covariates.

The multivariate tests and within-subjects effects analysis showed that the main effect of time was not significant, indicating that WtFI did not change significantly from T1 to T2 across all participants ($\Lambda = .996$, F(1, 427) = 1.611, p = .205). The interaction between time and self-control at T1 was significant ($\Lambda = .988$, F(1, 427) = 5.330, p < .05), indicating that the initial levels of self-control at T1 had a significant effect on WtFI over time. Specifically, participants with higher self-control at T1 tended to show different changes in WtFI from T1 to T2 compared to those with lower self-control at T1.

The between-subjects effects analysis revealed a significant difference in stress levels between groups, F(1, 427) = 149.824, p < .001, with higher stress levels in the control group – replicating the group effects. However, there were no significant differences between the intervention and the control group in terms of WtFI (F(1, 427) = .055, p = .82), meaning that the intervention itself did not produce any significant overall effects. Despite the lack of intervention effects, the hypothesized three-way interaction between time, group, and selfcontrol changes was significant (F(2, 427) = 2.497, p = .04). This suggested a potential combined effect of group and self-control changes on WtFI over time and supported the hypothesized intervention's effects being more pronounced among individuals who improved their self-control compared to those whose self-control remained stable or declined.
Specifically, for participants who experienced positive changes in self-control (*Improvers*), there was a significant increase in WtFI from T1 (M = 2.42, SD = 0.12) to T2 (M = 2.73, SD = 0.13), $\Lambda = .983$, F(1, 427) = 7.238, p < .01, for Improvers in the intervention group but not in the control group (T1: M = 2.51, SD = 0.09; T2: M = 2.54, SD = 0.10), $\Lambda = 1.000$, F(1, 427) = .11, p = .75; see Figure 4.2).

Figure 4.2





For participants who experienced a decrease in self-control (*Reducers*), there were no significant changes over time or between the intervention group (T1: M = 2.55, SD = 0.10; T2: M = 2.47, SD = 0.11) and control groups (T1: M = 2.39, SD = 0.10; T2: M = 2.43, SD = 0.10; see Figure 4.3).

Figure 4.3



Estimated Means of Work-to-Family Interference at T1 and T2 for Reducers

Similarly, for participants who did not experience any changes in self-control (*Stables*), there were no significant changes over time or between the intervention (T1: M = 2.36, SD = 0.10; T2: M = 2.32, SD = 0.11), and the control group (T1: M = 2.43, SD = 0.09, T2: M = 2.45, SD = 0.09; see Figure 4.4). However, a pairwise comparison revealed a significant difference in WtFI between Improvers and Stables within the intervention group at T2, with Improvers reporting higher interference levels (Mean Difference = 0.41, SE = 0.17, p < .05, 95% CI [0.00, 0.81]).

Figure 4.4



Estimated Means of Work-to-Family Interference at T1 and T2 for Stables

Threshold Effects of Self-Control on WtFI: A Spline Regression Analysis

Although the previous results offered some insight into individuals whose self-control improved, a spline regression analysis was conducted to capture better the effect of varying degrees of change in self-control on WtFI (T2). Specifically, thresholds were set at values greater than 0 and greater than .75, and splines were generated accordingly. These splines, along with the change in self-control scores, were entered into the model while controlling for initial WtFI and stress levels.

The results provided more nuanced and precise insights, supporting the hypotheses. Specifically, for both the intervention and control groups, initial WtFI was a significant predictor of WtFI at T2 (B = 0.56, t(174) = 8.090, p < .001 for the intervention group; B = 0.62, t(249) = 11.899, p < .001 for the control group), as were initial stress levels (B = 0.17, t(174) = 2.402, p < .05 for the intervention group; B = 0.14, t(249) = 2.625, p < .01 for the control group). However, the effect of self-control change was not significant in either group (B = 0.04, t(174) = 0.366, p = .72 for the intervention group; B = -0.07, t(249) = -0.777, p = .44 for the control group).

In the control group, the effect of the splines was insignificant, both for changes greater than 0 (B = 0.12, t(249) = 0.882, p = .38) and changes greater than .75 (B = -0.15, t(249) = -1.445, p = .15). The overall model for the control group was significant, F(5, 249) = 47.221, p < .001, explaining 48% of the variance in WtFI.

In contrast, for the intervention group, changes in self-control above 0 had a marginally significant positive effect (B = 0.32, t(174) = 1.769, p = .08), while changes greater than .75 had a significant negative effect (B = -0.32, t(174) = -2.421, p < .05). The overall model for the intervention group was also significant, F(5, 174) = 28.182, p < .001, explaining 43% of the variance in WtFI.

4.4. Discussion

This study aimed to test the role of a self-regulation intervention in improving selfcontrol and overall well-being among white-collar employees, specifically in relation to WtFI. The aim was to address a lack of previous findings concerning work-family interference (Kossek, 2016; Richardson, 2017) by employing an intervention approach to assess the role of self-control. Despite the well-intentioned intervention, the results showed no clear intervention effects but rather revealed a complex picture supporting the hypothesized interaction effects.

Firstly, the intervention did not result in a significant overall improvement in self-control skills among participants. The numbers of Improvers, Reducers, and Stables were similar across the intervention and control groups, suggesting that the intervention might not have been robust enough to produce substantial differences in self-control compared to natural variations in the control group. Considering the theoretical and practical instructions provided by the intervention to facilitate action planning and action control, these outcomes were unexpected. Possible reasons for these unexpected outcomes, as we will discuss below, could include

insufficient duration of the intervention, lack of participant engagement, external factors influencing self-control, or the complexity of changing established skills.

It was demonstrated that self-control in the control group correlated with stress at T1 with r = .13 (p < .05): Those employees in the control group who had more stress at T1 were more likely to change their self-control scores while this was not the case in the intervention group (r = .07, p = .36). Although we hypothesized that the intervention would enhance self-control, leading to a relationship between changes in self-control and WtFI for the intervention group, this correlation was observed only in the control group, r = .16 (p = .01). While the intervention aimed to increase self-control, its effects on subsequent WtFI were multifaceted and warranted a detailed discussion.

Despite the absence of significant overall effects, a deeper analysis revealed notable interaction effects. When investigating the effects of the intervention accordingly in terms of self-control, it was found that there was a hypothesized interaction effect. By controlling for initial self-control scores, we aimed to account for the regression to the mean effect, ensuring that changes resulting from individuals with extreme initial scores moving towards the average are not misinterpreted as the effects of the intervention.

For Improvers in the intervention group, there was a significant increase in WtFI from T1 to T2. These improvements can be primarily attributed to (besides due to chance) the structured action planning and action control exercises that encouraged participants to set clear goals, monitor their progress, and develop consistent behaviors. This suggests that the initial phase of adopting new strategies to improve self-control may have temporarily increased cognitive and emotional demands, exacerbating feelings of interference between work and family responsibilities. The spline regression analysis provided additional insights, showing that the degree of self-control change had a significant impact on WtFI. Specifically, participants who experienced moderate improvements in self-control showed a positive

association with increased WtFI, meaning that as self-control improved, WtFI also increased. This finding aligns with existing literature, where greater exertion of self-control may deplete cognitive resources, leading to perceived interference due to the additional effort required (Externbrink et al., 2019). This result may highlight the potential "dark side" of self-control. Individuals who improved their self-control in the short term might have taken on additional responsibilities (Koval et al., 2015), leading to over-commitment (Hetland et al., 2012) and increased vulnerability to WtFI.

Interestingly, the results also showed that for participants who experienced higher increases in self-control, there was a negative relationship with WtFI, indicating that as self-control continued to increase significantly, WtFI actually decreased. This suggests that once individuals overcame the initial cognitive load associated with moderate self-control improvements, they may have developed more effective strategies to manage both work and family responsibilities. This finding supports the idea of Muraven and Baumeister (2000), who suggest that the benefits of self-control are most pronounced when there is a substantial change in self-regulatory behaviors.

These findings highlight a critical distinction in how self-control changes impact WtFI. Moderate increases in self-control may lead to short-term increases in WtFI due to the higher cognitive and emotional demands of self-regulation. However, larger increases in self-control seem to provide longer-term benefits, reducing WtFI as individuals develop more sustainable strategies for balancing these domains.

Notably, participants categorized as Reducers and Stables did not exhibit significant changes in WtFI over time, regardless of whether they were in the intervention or control group. These findings indicate that both a reduction and stability in self-control did not translate into noticeable fluctuations in WtFI, possibly due to established coping strategies or routines that mitigate the impact of fluctuations in self-control (Baumeister & Heatherton, 1996; Sonnentag & Fritz, 2007).

4.4.1. Limitations and Implications for Future Interventions

The findings underscore the complexities involved in changing self-control and WtFI through a self-regulation intervention. The results align with previous meta-analysis findings, indicating that repeated practice improves self-control with a smaller effect size than general treatment effects (Friese et al., 2017). This small effect size can be attributed to several factors: the design and implementation of the intervention, the intervention's length and duration, the measurement methods, and the inherent stability of the targeted behaviors and attitudes.

Previous research suggests that longer interventions are generally more effective (de Ridder et al., 2020). Although many previous training studies also used the strength model of self-control and a period of two weeks or less (Friese et al., 2017), this period may be too short in the current study to fully evaluate the training's effectiveness in achieving its goals. Future interventions should be designed with longer durations and increased intensity to produce more significant behavioral changes. Short-term interventions such as the one in the current study may not be sufficient to impact deeply ingrained behaviors and attitudes overall.

The self-control scale used may not have been ideal for capturing changes in our primary variable, with subjective measurements potentially leading to participants overestimating their abilities in initial assessments. Combining self-report measures with objective indicators (e.g., habit tracker, objective stress indicators) could offer a more comprehensive evaluation of the intervention's effects, mitigating biases associated with self-reports. The time and effort required to improve self-control, coupled with the absence of immediate benefits, may have led to participant disengagement, hindering the intervention's long-term positive effects (Ouellette & Wood, 1998). Moreover, the use of a single-item scale to measure stress presents another limitation. This choice was made for economic reasons, considering the length of the

questionnaire and the time required to complete it. Addressing these issues in future studies could enhance the understanding of the true efficacy of self-regulation interventions and their impact on self-control. Due to limited control over whether participants actually engaged in the intervention in the current study, incorporating advanced technological platforms for real-time monitoring and feedback on self-control in future studies can further improve adherence and data accuracy, helping to overcome limitations associated with remote interventions (e.g., Reinke & Ohly, 2024).

Despite these limitations, these findings underscore the importance of designing selfregulation interventions that not only enhance self-control skills but also address the cognitive and emotional demands placed on participants. While improving self-control is beneficial, moderate improvements may initially exacerbate WtFI as individuals adjust to new selfregulation strategies. Future interventions should aim to prevent over-commitment and cognitive overload, ensuring that the benefits of self-control on well-being are sustained in the long term. By considering the risk of taking on excessive responsibilities, we can develop more effective strategies that enhance employee well-being and work-life balance.

4.4.2. Conclusion

The findings of this study underscore the intricate nature of self-regulation interventions and their impact on WtFI. The data revealed that moderate improvements in self-control initially increase WtFI due to the cognitive demands of self-regulation; higher self-control gains appear to have long-term benefits, reducing WtFI. These findings underscore the importance of carefully considering the cognitive costs associated with self-regulation interventions and suggest that sustainable improvements in self-control may lead to better outcomes in managing work-family dynamics over time. The observed stability among Reducers and Stables indicates that existing coping mechanisms play a crucial role in managing work-family dynamics. This highlights the importance of considering individual differences in baseline self-control and coping strategies when designing interventions.

By delving into the role of self-control, this study contributes to a deeper understanding of how it interacts and influences work-family dynamics. This research lays the groundwork for developing more effective interventions aimed at improving health and well-being through sustainable behavior changes. By considering these factors, we can formulate strategies that better enhance employee well-being and work-life balance, ultimately leading to a more productive and satisfied workforce. **Chapter 5: General Discussion**

5.1. Main Research Objectives and Outcomes

The scientific investigation of life goals and work-family interference, while different views depending on the background discipline, is prevalent; this thesis mainly integrates perspectives from Psychology and Sociology, as well as multiple theoretical frameworks to understand and predict their role in improving employee well-being. Utilizing the Work-Home Resources model (ten Brummelhuis & Bakker, 2012), the Conservation of Resources theory (Hobfoll, 1989), the Strength Model of Self-Regulation (Muraven & Baumeister, 2000), and the Personal Resource Allocation framework (Grawitch et al., 2010), this thesis sought to answer and test the following research questions and hypotheses in Table 5.1 and Table 5.2.

Three distinct interconnected studies were conducted to address these research questions. The first study (Chapter 2) explored the theoretical link among work-family interference, wellbeing, and life goals, aiming to understand these relationships between all genders. Using data from Wave 8 (2015/2016) and Wave 10 (2017/2018) of the German Family Panel Survey (pairfam), structural equation modeling analyses were conducted. These results suggest that life goals may be one of the personal resources that can be used to manage work and family demands in the Work-Home Resources model (ten Brummelhuis & Bakker, 2012). The first study demonstrated that life goals significantly moderate the negative impacts of work-family interference on life satisfaction, showing that interference across life domains reduces overall satisfaction, particularly when primary life goals disrupt other domains. Specifically, those who prioritize agentic goals are negatively affected by work-to-family interference (WtFI) on life satisfaction, while those who prioritize communal goals are negatively affected by family-towork interference (FtWI) on life satisfaction.

Table 5. 1

Research Questions and Hypotheses Across Studies

Research Question	Study (Chapter)	Hypothesis
1) How do life goals shape the relationship between work-family interference and well-being, and in what ways do they impact this relationship?	Study 1 & 2 (Chapter 2 & 3)	 Individuals predominantly pursuing agentic life goals are prone to experiencing more pronounced negative impacts of WtFI on life satisfaction compared to those primarily focused on communal goals or maintaining a balance between them (Moderation). Individuals predominantly pursuing communal life goals are prone to experiencing more pronounced negative impacts of FtWI on life satisfaction compared to those primarily focused on agentic goals or maintaining a balance between them (Moderation). Pursuing agentic life goals has a positive relationship with well-being. However, these goals may negatively impact well-being through WtFI (Mediation). Pursuing communal life goals has a positive relationship with well-being. However, these goals may negatively impact well-being through FtWI (Mediation).
2) How do different gender roles in different cultural contexts (i.e., Turkish and German cultures) modify the relationship among the pursuit of different life goals, work-family interference, and overall well-being?	Study 1 & 2 (Chapter 2 & 3)	 Women who focus primarily on agentic life goals are likely to experience more substantial negative effects of WtFI on life satisfaction than their counterparts who predominantly pursue agentic life goals (in Germany). Men who focus primarily on communal life goals are likely to experience more substantial negative effects of FtWI on life satisfaction than their counterparts who predominantly pursue communal life goals (in Germany). Turkish women who pursue agentic life goals are likely to experience weaker positive effects on well-being and more negative effects from WtFI compared to German women with agentic life goals. German men who pursue communal life goals are likely to experience weaker positive effects on well-being and more negative effects from FtWI compared to Turkish men with communal life goals.
3) How do varying levels of self- control modify the relationship among the pursuit of different life goals, work-family interference, and overall well-being?	Study 2 & 3 (Chapter 3 & 4)	 Individuals who pursue agentic life goals and have high self-control are likely to experience stronger positive effects on well-being and weaker positive effects on WtFI compared to those with lower self-control. Individuals who pursue communal life goals and have high self-control are likely to experience stronger positive effects on well-being and weaker positive effects on FtWI compared to those with lower self-control.
4) What role do self-regulation interventions play in managing work-family interference?	Study 3 (Chapter 4)	• Individuals who improve their self-control through the intervention are likely to experience stronger positive effects from a reduction in WtFI compared to those whose self-control remained stable or reduced.

Table 5. 2

Summary of Hypotheses and Their Empirical Results

Hypothesis	Support	t Status
• Individuals predominantly pursuing agentic life goals are prone to experiencing more pro WtFI on life satisfaction compared to those primarily focused on communal goals or main (Moderation).	nounced negative impacts of • Supp ntaining a balance between them	ported.
• Individuals predominantly pursuing communal life goals are prone to experiencing more FtWI on life satisfaction compared to those primarily focused on agentic goals or maintai (Moderation).	pronounced negative impacts of • Part ning a balance between them	ially supported.
• Pursuing agentic life goals has a positive relationship with well-being. However, these go being through WtFI (Mediation).	als may negatively impact well- • Part	ially supported.
• Pursuing communal life goals has a positive relationship with well-being. However, these well-being through FtWI (Mediation).	goals may negatively impact • Part	ially supported.
• Women who focus primarily on agentic life goals are likely to experience more substantia life satisfaction than their counterparts who predominantly pursue agentic life goals (in G	al negative effects of WtFI on ermany).	supported.
• Men who focus primarily on communal life goals are likely to experience more substantial life satisfaction than their counterparts who predominantly pursue communal life goals (i	al negative effects of FtWI on n Germany).	supported.
• Turkish women who pursue agentic life goals are likely to experience weaker positive effects from WtFI compared to German women with agentic life goals.	• Part	ially supported.
• German men who pursue communal life goals are likely to experience weaker positive ef negative effects from FtWI compared to Turkish men with communal life goals.	• Not	supported.
 Individuals who pursue agentic life goals and have high self-control are likely to experier well-being and weaker positive effects on WtFI compared to those with lower self-control and initial who pursue communal life goals and have high self control are likely to average to the self control are likely to aver	 Ice stronger positive effects on I. Not 	supported.
• Individuals who pursue communal me goals and have high sen-control are likely to expe on well-being and weaker positive effects on FtWI compared to those with lower self-cor	itrol. • Part	ially supported.
• Individuals who improve their self-control through the intervention are likely to experien a reduction in WtFI compared to those whose self-control remained stable or reduced.	ce stronger positive effects from • Not	supported.

These findings aligned with Noor's (2004) study on role centrality, indicating that emphasizing goals in a specific domain increases sensitivity to stressors in that area, leading to heightened negative impacts on life satisfaction due to a perceived lack of control. While some literature indicated that individuals are willing to tolerate higher levels of work-family interference for domains they value more (Greenhaus et al., 2001; Matthews et al., 2012), current results showed that if the interference persists, the lack of return on resource investment and a sense of losing control over the targeted domain result in a greater decline in satisfaction compared to those who have not invested resources in that domain. It was found in the current study that there was no significant gender disparity in the influence of work-family interference on life satisfaction among individuals primarily focused on agentic or communal life goals. Namely, women and men with primarily agentic goals were similarly affected by the impact of WtFI on life satisfaction, and women and men with primarily communal goals were similarly affected by the impact of FtWI on life satisfaction. The social role theory (Eagly & Wood, 2012) posits that there are differences between men and women due to traditional gender roles. These differences are more pronounced in less egalitarian countries and less noticeable in more gender-egalitarian countries like Germany, where stereotypical gender expectations are less influential (Eagly & Wood, 2012; Kosakowska-Berezecka et al., 2023). In this respect, these results are consistent with social role theory, which says that in cultures where traditional gender roles are less dominant, the shared challenges of balancing demanding professional aspirations with family responsibilities might explain these similarities. In cultures where traditional gender roles are less dominant, the shared challenges of balancing demanding professional aspirations with family responsibilities might explain these similarities. Therefore, it is the centralization of specific life goals, regardless of gender roles in more egalitarian societies, that strengthens the negative impact of interference caused by the prioritized domain on life satisfaction.

Initial findings validated that life goals play a crucial role in shaping the association between work-family interference and life satisfaction. The specific content of life goals and the alignment of personal resources significantly influenced how individuals navigate workfamily interference and maintain well-being. Building on these insights, the second study (Chapter 3) shifted the focus from examining life goals as a moderating factor to investigating them as a predictor. Understanding life goals as a predictor allowed us to explore how primary life goals shape the interplay between work-family interference and well-being. This approach is guided by existing literature, which highlights the importance of considering various dimensions of life goals in predicting outcomes related to well-being (Sheldon et al., 2004; MacLeod, 2012). These guided findings indicate that life goals can have both direct and indirect effects. It has been shown that individuals who prioritize family goals over work and leisure goals experience higher life satisfaction, mediated by family satisfaction, highlighting the direct impact of family-oriented life goals on well-being (Masuda & Sortheix, 2012). These results underline the importance of considering the mediating role of work-family interference, as the pursuit of life goals dictates how personal resources are allocated across different life domains.

Building on the theories mentioned above, the second study investigated the predictor role of life goals on work-family interference and their cumulative impact on life satisfaction and stress, including the moderator role of culture and self-control. It aimed to address the first three research questions mentioned above. The research focused on white-collar employees from both Germany and Türkiye and utilized cross-sectional and self-report survey data for analyses. The findings indicated that prioritizing specific life goals leads to one domain taking precedence, causing interference in other areas of life, and this interference, regardless of the type of life goal, indirectly reduced overall well-being. Specifically, as individuals place greater value on agentic goals, they experience more WtFI, while as they place greater value on communal goals, they experience more FtWI, which ultimately negatively impacts their wellbeing.

In this study, a dual-perspective approach was employed: There were two narratives of the story. On the one hand, pursuing any life goal enhanced life satisfaction and presented a positive narrative. Confirming self-regulation theories, having goals brings a sense of fulfillment and life satisfaction (Emmons, 2003). On the other hand, excessive pursuit can lead to stress (Toth et al., 2018), especially when goals highlight deficiencies (Mayser et al., 2008). Since these pursuits require resource allocation, the more people prioritize a specific life goal, the more they experience a specific direction of work-family interference. Consequently, their life satisfaction levels decrease, and stress levels increase, forming a negative narrative. These findings align with previous boundary management research showing that individuals with a greater work focus experience more WtFI, while those with a stronger family focus face more FtWI (Ashforth et al., 2000; Kossek et al., 2012). Similarly, other research indicates that overworking, even voluntarily or because of motivation, can impair mental health, as the risks, such as WtFI, are often underestimated (Kuroda & Yamamoto, 2019; Virtanen et al., 2018).

Another aspect of the second study focused on the (cultural) differences between German and Turkish societies, using citizenship measurement to minimize the response burden. It is found that incongruence between life goals and societal gender role expectations negatively affects well-being. Specifically, German women enhance life satisfaction and eliminate stress by pursuing agentic goals valued by their community, consistent with previous literature (Myers & Diener, 1995; Sedikides et al., 2003). For Turkish women, endorsing agentic goals increases stress and reduces life satisfaction, suggesting misalignment with societal expectations suppresses positive effects. In contrast, there is no difference between German and Turkish men in the pursuit of communal goals concerning well-being, aligning with literature suggesting that valuing communal goals is crucial for well-being regardless of cultural background (Abele, 2014; Le et al., 2018).

There was no significant difference in the impact of WtFI on the well-being of Turkish and German women with agentic goals. However, the effect of FtWI on life satisfaction, but not on stress, differed significantly between Turkish and German men with communal goals, with Turkish men experiencing more detrimental effects. These findings suggest that WtFI uniformly harms the well-being of women with agentic goals across cultures, as supported by the findings of a previous study (Kusnierz et al., 2022). However, in societies with pronounced gender inequalities, men who see themselves as family providers may experience a greater negative impact on life satisfaction from FtWI compared to those in more gender-equal societies (Allen et al., 2015).

The second study also examined the differences between low and high self-control groups. The findings showed no differences between these groups for individuals who prioritize agentic goals, suggesting that agentic life goals may already manifest themselves through a greater level of self-focus (Garcia et al., 2015; Saragovi et al., 1997). A significant difference was observed for those valuing communal goals, but contrary to expectations: The high self-control group experienced more FtWI than the low self-control group when emphasizing communal goals. High self-control individuals often take on more responsibilities (Koval et al., 2015), potentially leading to an underestimation of demands and an overestimation of their capacities (Hetland et al., 2012). Their confidence in handling their responsibilities might lead them to underestimate the actual demands they face, resulting in greater FtWI as they stretch themselves too thin across different domains. Additionally, consistent with existing literature, focusing on communal goals correlated with higher life satisfaction for the high self-control group compared to the low self-control group (de Ridder et al., 2012; Tangney et al., 2004). This dual aspect of self-control—both as a source of taking

on more responsibilities and as a buffer against stress—highlights the complex role self-control plays in managing work-family dynamics.

Drawing from the rather negative narrative, it was demonstrated that preventing workfamily interference requires individuals to allocate their resources effectively while pursuing their goals. This means ensuring that the needs of both the work and family domains are met. One could imagine that achieving healthy resource allocation necessitates strong self-control skills (Baumeister et al., 2007; Muraven et al., 2006). Even though it has been shown before that higher self-control is associated with lower work-family interference (Chen et al., 2021), the current findings cross-sectionally revealed that individuals with relatively high self-control tend to experience greater FtWI. This counterintuitive result suggests that high self-control may not be sufficient alone to mitigate interference between family and work responsibilities. Consequently, a self-management training program was proposed with the aim of altering how individuals experience and manage work-family interference.

Lastly, the third study evaluated a novel intervention aimed at enhancing self-control to reduce WtFI among the same group of German white-collar employees from the second study, using longitudinal and self-report survey data for analysis. This experimental approach investigated valuable insights into the role of self-control in managing work-family dynamics, addressing a crucial gap in current research. The intervention was designed based on selfregulation theory (Vohs & Baumeister, 2004), implementation intentions (Gollwitzer et al., 2004), and Behavior Change Techniques (BCTs; Michie et al., 2013). Three groups were identified: Those whose self-control skills improved as an effect of the intervention, those whose skills remained the same, and those whose skills decreased, possibly due to gaining a more realistic understanding of self-control and reporting more accurate scores. The intervention was expected to have a more significant impact on the first group, as improved self-control would lead to better planning and control over actions, particularly in resource allocation. The results provided a detailed understanding of how changes in self-control can influence WtFI.

Although the intervention did not significantly impact the self-control levels of the experimental group, and there was no significant difference in self-control levels between the experimental and control groups post-intervention, there was a notable difference in WtFI levels depending on changes in self-control. It was found that participants who improved their self-control during the two-week intervention initially experienced heightened WtFI. Upon closer examination, it was revealed that those with moderate self-control improvements experienced a temporary rise in WtFI, likely due to increased cognitive and emotional demands. However, participants with more substantial self-control improvements eventually experienced a decrease in WtFI, indicating that greater improvements may offer long-term benefits by reducing interference. In contrast, individuals whose self-control remained stable or declined showed no significant changes in WtFI, likely due to maintaining consistent coping mechanisms. These findings suggest that improving self-control, especially through interventions, can initially increase cognitive and emotional demands, temporarily heightening WtFI until new strategies become habitual and less taxing. Previous studies (Reinke & Ohly, 2024) demonstrated that engaging in action control and planning exercises enables the management of work-family dynamics, with the benefits of self-control being most pronounced when there is a substantial change in self-regulatory behaviors (Muraven & Baumeister, 2000).

5.2. Main Contributions

One of the primary contributions of this thesis is its elucidation of how life goals function as both predictors and moderators of work-family interference and its subsequent impact on well-being. While the work-family interference literature has underexplored individual factors influencing these dynamics (see Allen et al., 2020; Byron, 2005 for an overview), this thesis incorporates the concept of life goals, which provide direction and purpose to people's lives (Baumeister, 2008). Life goals, alongside self-regulatory resources, are crucial for managing professional and personal demands, thereby maintaining overall well-being. However, the freedom to choose life goals also brings the freedom to choose one's struggles. Similar to how social support is examined in the literature as both a predictor and a moderator, life goals can directly determine resource loss and also serve as a mechanism to cope with this loss (French et al., 2018; Seiger & Wiese, 2009). By examining different types of life goals, such as agentic (career-oriented) and communal (family-oriented), this thesis highlights their significant influence on work-family interference and its subsequent effects on well-being. This dual role emphasizes the importance of understanding life goals not just as catalysts of interference but also as elements that modulate the impact of this interference on individual well-being. This thesis could also make a significant contribution to research on voluntary usage of information and communication technology. Individuals' perceptions of Information and Communication Technology use are shaped by their motivations (Schöllbauer et al., 2021); namely, those aiming for career advancement tend to use Information and Communication Technology more during non-work hours, increasing WtFI (Schlachter et al., 2018). Current results may suggest that individuals with communal goals may use non-work-related technology during work hours voluntarily, resulting in more FtWI.

This thesis further illuminates the critical impact of cultural and societal norms on the management of work-family interference and its repercussions on well-being. Previous studies have suggested that gender roles influence the effects of work-family interference differently, but the findings have been inconsistent (Hagqvist et al., 2017; Shockley et al., 2017). This thesis addresses this by considering the differential impact of work-family interference on men and women across various cultural contexts. By measuring individuals' pursued goals rather than relying solely on stereotypical cultural orientations, the research offers a more valid assessment, potentially explaining existing inconsistencies in the literature. The results

demonstrate that men and women, whether pursuing agentic or communal life goals, experience similar levels of impact from work-family interference on their well-being within the same society, as observed in Germany. However, societal expectations and cultural norms, especially in less egalitarian societies, can exacerbate these challenges (e.g., Drobnič et al., 2010). This is evident in the comparison between German and Turkish cultures; in more gender-equal societies like Germany, women who pursue agentic goals receive societal support, enhancing their life satisfaction. Conversely, in less egalitarian societies like Türkiye, traditional gender roles impose additional challenges for women with agentic goals, leading to reduced well-being. These cultural differences highlight the role of societal norms in shaping individual experiences of work-family interference and the varied well-being outcomes while pursuing life goals. Although cultural expectancies and gender roles may predispose certain groups to behave in specific ways, considering individuals' current circumstances within these contextual factors provides a better understanding of the outcomes. This thesis makes an important contribution to the literature by showing that the disparities observed are not merely gender-based but are deeply rooted in cultural values and expectations. Similar gender roles require different behavioral patterns in different societies, resulting in varied well-being outcomes. The empirical evidence from cross-cultural studies and the theoretical frameworks on cultural norms provide a robust basis for these conclusions⁶.

⁶ As previously noted, the agentic and communal values supported by societies are not mutually exclusive. These values represent tendencies in how individuals may behave in certain situations. In every society, individuals can prioritize and pursue both life goals depending on the context. However, the structure of collectivist cultures may expose individuals to settings that promote communion norms more frequently, thus shaping their life goals in that direction. Additionally, the reasons for pursuing particular life goals are also significant. For example,

Lastly, this thesis explores the potential "dark side" of self-control in the context of workfamily interference, revealing that while high self-control is generally seen as beneficial, it can paradoxically increase interference in certain conditions. The findings demonstrate that moderate increases in self-control can lead to heightened WtFI, likely due to the heightened cognitive and emotional demands associated with regulating both work and family life. However, the results also indicate that more substantial improvements in self-control can mitigate the negative effects, leading to reduced interference. This research challenges the conventional view that self-control uniformly reduces strain and highlights the importance of considering the short-term cognitive costs involved in improving self-regulatory capacity. By doing so, the thesis contributes to a more comprehensive understanding of the dynamics between self-control and work-family interference, suggesting that self-regulation interventions must account for these transitional costs to avoid unintended negative outcomes such as burnout. This perspective enriches the existing literature by emphasizing that the benefits of self-control are not immediate and may require sustained effort and becoming more habitual before yielding positive long-term outcomes.

Overall, this doctoral thesis contributes to the understanding of life goals, work-family interference, and well-being by offering a comprehensive framework that integrates key personal and macro-cultural resources. From an interdisciplinary viewpoint, the psychological and sociological foundations of the thesis provide insights into individual differences, perspectives on self-regulation and resource allocation, and the impact of cultural norms and societal expectations. This comprehensive approach not only enriches the theoretical landscape

agentic life goals may be adopted for communal reasons, such as being successful at work to serve the family. The current hypotheses in this thesis have been developed with the aim of comparing the relative importance of these life goals across different cultures.

but also offers practical implications for individuals and organizations aiming to foster a better work-family balance and improve overall life satisfaction.

5.3. Practical Recommendations

It is crucial to consider practical contributions that can be implemented based on these findings. One significant area of focus is managing work-family interference by understanding life goals and their impact on employee well-being.

First, understanding life goals as both a predictor and moderator allows us to explore how life goals directly influence the interplay between work-family interference and well-being and exacerbate the impact of work-family interference on well-being. Recognizing how their goal endorsements influence their tendencies can empower employees to manage their work and family dynamics more effectively. To support this, targeted training and support systems such as coaching and mentoring programs, advisory services, and workshops designed to align personal and professional goals can equip employees with the necessary skills and knowledge to identify their unique needs based on their life goals and manage work-family interference. The role of social support is also crucial in managing work-family interference and enhancing well-being. Employees with agentic life goals tend to experience and be affected by WtFI, impacting their well-being. Instead of focusing solely on work to control this interference, they can seek support from their coworkers and managers to mitigate these effects. Conversely, employees with communal life goals tend to experience and be affected by FtWI, impacting their well-being. These individuals can benefit from increased support from family and friends, helping to prevent personal domain stressors from interfering with their work.

Recognizing the good and bad outcomes that life goals can lead to is crucial not only for individuals but also for organizations. To manage the well-being of their valuable workforce, developing tailored interventions for individuals with different goal endorsements is more effective. For organizations, the findings offer practical strategies for managing work-family conflict and enhancing well-being. Tailored interventions focusing on managing WtFI specifically for employees with agentic life goals are crucial. Organizations should encourage familial support systems and promote a family-friendly workplace environment, especially for employees with communal life goals. Improving human resources professionals' skills to understand the diverse needs of employees based on their agentic and communal life goals and to recognize their work-family interference experiences. Organizations should also develop targeted guidelines for Information and Communication Technology use that accommodate different life goals. For instance, employees with agentic goals might benefit from structured breaks and clear boundaries for Information and Communication Technology use during non-work hours. Conversely, those with communal goals might need policies that allow for limited, purposeful use of non-work-related technology during work hours to manage family-related tasks without compromising work performance. This understanding enables professionals to offer the necessary support effectively.

Understanding the diverse needs of employees based on their life goals, gender, and cultural background is vital for effectively managing work-family interference and enhancing overall well-being. In different cultural settings, especially in gender-unequal contexts, women pursuing agentic goals and men pursuing communal goals experience varying impacts on well-being, as demonstrated by current findings. Therefore, cross-cultural sensitivity and tailored interventions are necessary to address these differences effectively. Promoting gender equality through programs and support groups can help mitigate the impact of societal gender norms on work-family dynamics. Cultural sensitivity in multinational companies is crucial. Organizations should adapt their practices to fit local cultural contexts, ensuring that interventions are relevant and effective across different regions. Increasing cross-cultural awareness among managers and employees can improve the effectiveness of work-life balance

strategies. Localized interventions that consider cultural differences can help create supportive environments that accommodate diverse life goals and work-family dynamics.

Additionally, while self-control is generally beneficial for managing demands, the current results reveal a more complex picture. Employees who moderately improved their self-control experienced an initial increase in work-family interference due to heightened cognitive and emotional demands. To mitigate this, employees should set realistic goals and be mindful of the potential for over-commitment. Programs aimed at managing work-family dynamics should focus not only on stress management and resource utilization but also on gradual, sustainable self-control development. Organizations should focus on developing the self-control skills of employees with lower levels of self-control while being mindful of the initial burden such efforts may impose. Interventions should be designed to gradually build self-control without overwhelming employees and should be offered over extended periods, allowing for a smoother integration of strategies into daily routines. This approach minimizes cognitive and emotional strain, leading to long-term improvements without the risk of burnout or negative outcomes.

In conclusion, implementing these practical contributions will create more supportive and balanced work environments, thereby enhancing employee well-being and organizational success while enabling people to pursue their life goals. By fostering an environment that recognizes and supports the diverse needs of employees, including training line managers, fostering a supportive climate, and providing managerial or collegial support, organizations can achieve sustainable success and maintain a competitive edge.

5.4. Main Limitations and Suggestions for Future Research

This doctoral thesis has several limitations that must be acknowledged to clarify the scope and areas for improvement. Firstly, while the first study utilized large-scale representative survey data from Germany, the primary data collected in the second study might

not fully represent the broader population. Future studies should strive to achieve a better balance in representativity. In the future, it is crucial to include more diverse samples, especially from various cultural contexts, to evaluate the universality of these findings effectively.

The cross-sectional design of the second study provides only a static snapshot of data, limiting the ability to make causal relationships and observe temporal dynamics. This design constrains the understanding of how life goals influence work-family interference and wellbeing, as well as the moderating roles of gender and culture. Additionally, common method bias may be present due to the reliance on self-reported data collected at a single point in time (Podsakoff et al., 2003). To better establish these causal relationships, future studies should adopt longitudinal designs, capturing temporal dynamics and causal relationships more comprehensively, thus enriching the understanding of the variables' interrelations.

Another critical issue is the measurement of some variables, and only subjective measures were taken. The use of a single-item self-control and life satisfaction instrument across all studies raises potential reliability concerns. Additionally, the method of measuring life goals by distributing 15 importance points among different domains has been criticized for potentially introducing biases in the first study. Using citizenship as a criterion for measuring culture is also insufficient. Future studies should consider more comprehensive measures, accounting for time and cost, and include both subjective and objective measures. Employing multi-item scales for measuring self-control, well-being, and culture will enhance the measurement reliability and reduce potential biases from the methods used in this study.

Furthermore, the design and duration of the intervention in the last study may have contributed to the small effect sizes observed in self-control improvements. Longer and more intensive interventions are generally more effective (Friese et al., 2017), suggesting that the current intervention duration might have been insufficient. Participant engagement posed another challenge, particularly in the longitudinal studies. The demands on time and effort, coupled with the absence of immediate, noticeable benefits, may have led to participant disengagement, hindering long-term positive effects. Future interventions should extend the duration and intensity to produce significant behavioral changes. Extended follow-up assessments are needed to determine if the effects of training persist over time (Lippke et al., 2021). Maintaining participant engagement is crucial, and future interventions should incorporate strategies such as motivational elements and real-time feedback mechanisms to sustain engagement over time. Integrating advanced technological platforms for real-time monitoring and feedback could improve adherence and data accuracy, enhancing the effectiveness of interventions and providing continuous support to participants (e.g., Reinke & Ohly, 2024).

Future research should also consider additional factors, such as the role of collegial and family support, as well as the alignment between the life goals of employees and their partners. Social support can significantly buffer the impact of life goals on work-family interference and, in turn, the effects of work-family interference on overall well-being (French et al., 2018; Seiger & Wiese, 2009). The alignment between partners' life goals can further mitigate conflicts and enhance mutual support, fostering a more harmonious balance between work and family life (Booth-LeDoux et al., 2020; Schooreel & Verbruggen, 2016). Understanding how the pursuit of life goals and self-control leads to various outcomes can provide deeper insights into managing work-family dynamics and overall well-being. Future research should also explore the long-term effects of voluntary Information and Communication Technology use on work-family interference across different life goals. A longitudinal study could provide deeper insights into how these dynamics evolve over time, considering both agentic and communal goal endorsements. In addition, incorporating qualitative approaches in future research could offer a more comprehensive perspective, uncovering the underlying mechanisms and personal

contexts that shape these complex relationships. While goal endorsements may appear dichotomous at the surface level, deeper qualitative analysis could reveal the fundamental processes, highlighting why both life goals are important and the purposes they serve.

In summary, addressing these directions in future research will help develop more robust and generalizable findings, ultimately contributing to a deeper understanding and improvement of balance between life domains and overall well-being while pursuing life goals. Combining the strengths of cross-sectional, longitudinal, and experimental designs offers a comprehensive approach to studying these intricate relationships. References

- Abele A. E. (2014). Pursuit of communal values in an agentic manner: A way to happiness? *Frontiers in Psychology*, 5, 1320. https://doi.org/10.3389/fpsyg.2014.01320
- Abele, A. E., & Wojciszke, B. (2014). Communal and agentic content in social cognition: A dual perspective model. In J. M. Olson & M. P. Zanna (Eds.), Advances in experimental social psychology (Vol. 50, pp. 195–255). Academic Press. https://doi.org/10.1016/B978-0-12-800284-1.00004-7
- Abele, A. E., Hauke, N., Peters, K., Louvet, E., Szymkow, A., & Duan, Y. (2016). Facets of the fundamental content dimensions: Agency with competence and assertivenesscommunion with warmth and morality. *Frontiers in Psychology*, 7, 1810. https://doi.org/10.3389/fpsyg.2016.01810
- Adams, G. A., & Jex, S. M. (1999). Relationships between time management, control, workfamily conflict, and strain. *Journal of Occupational Health Psychology*, 4(1), 72– 77. https://doi.org/10.1037/1076-8998.4.1.72
- Allen, T. D., & Paddock, E. L. (2015). How being mindful impacts individuals' work-family balance, conflict, and enrichment: a review of existing evidence, mechanisms and future directions. In J. Reb & P. W. B. Atkins (Eds.), *Mindfulness in organizations: Foundations, research, and applications* (pp. 213–238). Cambridge University Press. https://doi.org/10.1017/CBO9781107587793.011
- Allen, T. D., French, K. A., Dumani, S., & Shockley, K. M. (2015). Meta-analysis of work– family conflict mean differences: Does national context matter? *Journal of Vocational Behavior*, 90, 90–100. https://doi.org/10.1016/j.jvb.2015.07.006
- Allen, T. D., French, K. A., Dumani, S., & Shockley, K. M. (2020). A cross-national metaanalytic examination of predictors and outcomes associated with work-family conflict. *Journal of Applied Psychology*, 105(6), 539–576. https://doi.org/10.1037/APL0000442
- Allen, T. D., Herst, D. E. L., Bruck, C. S., & Sutton, M. (2000). Consequences associated with work-to-family conflict: A review and agenda for future research. *Journal of Occupational Health Psychology*, 5(2), 278–308. https://doi.org/10.1037/1076-8998.5.2.278
- Allen, T. D., Johnson, R. C., Saboe, K. N., Cho, E., Dumani, S., & Evans, S. (2012). Dispositional variables and work–family conflict: A meta-analysis. *Journal of Vocational Behavior*, 80(1), 17–26. https://doi.org/10.1016/J.JVB.2011.04.004
- Allen, T. D., Regina, J., Wiernik, B. M., & Waiwood, A. M. (2023). Toward a better understanding of the causal effects of role demands on work–family conflict: A genetic modeling approach. *Journal of Applied Psychology*, 108(3), 520– 539. https://doi.org/10.1037/ap10001032
- American Psychological Association. (2023). *Work in America Survey*. Retrieved from https://www.apa.org/pubs/reports/work-in-america/2023-workplace-health-well-being
- Amstad, F. T., Meier, L. L., Fasel, U., Elfering, A., & Semmer, N. K. (2011). A meta-analysis of work-family conflict and various outcomes with a special emphasis on cross-domain versus matching-domain relations. *Journal of Occupational Health Psychology*, 16(2), 151–169. https://doi.org/10.1037/A0022170

- Ashforth, B. E., Kreiner, G. E., & Fugate, M. (2000). All in a day's work: Boundaries and micro role transitions. *The Academy of Management Review*, 25(3), 472–491. https://doi.org/10.2307/259305
- Austin, J. T., & Vancouver, J. B. (1996). Goal constructs in psychology: Structure, process, and content. *Psychological Bulletin*, 120(3), 338–375. https://doi.org/10.1037/0033-2909.120.3.338
- Aviva. (2022). Working Lives Report 2022. Retrieved from https://static.aviva.io/content/dam/documentlibrary/adviser/workinglivesreport.pdf
- Aycan, Z. (2008). Cross-cultural approaches to work-family conflict. In K. Korabik, D. S. Lero,
 & D. L. Whitehead (Eds.), *Handbook of work-family integration* (pp. 353–370). Academic Press. https://doi.org/https://doi.org/10.1016/B978-012372574-5.50022-3
- Aycan, Z., & Eskin, M. (2005). Relative contributions of childcare, spousal support, and organizational support in reducing work-family conflict for men and women: The case of Turkey. Sex Roles, 53(7–8), 453–471. https://doi.org/10.1007/S11199-005-7134-8
- Aycan, Z., & Korabik, K. (2017). An integrative model of work-family conflict: Pan-cultural effects and cross-cultural differences. In K. Korabik, Z. Aycan, & R. Ayman (Eds.), *The work-family interface in global context* (pp. 269–289). Routledge.
- Babalola, M. T., Kwan, H. K., Ren, S., Agyemang-Mintah, P., Chen, H., & Li, J. (2021). Being ignored by loved ones: Understanding when and why family ostracism inhibits creativity at work. *Journal of Organizational Behavior*, 42(3), 349–364. https://doi.org/10.1002/job.2499
- Bakan, D. (1966). *The duality of human existence: An essay on psychology and religion*. Rand Mcnally.
- Bakker, A. B., & Demerouti, E. (2007). The Job Demands-Resources model: State of the art. *Journal of Managerial Psychology*, 22(3), 309–328. https://doi.org/10.1108/02683940710733115
- Bakker, A. B., Du, D., & Derks, D. (2019). Major life events in family life, work engagement, and performance: A test of the work-home resources model. *International Journal of Stress Management*, 26(3), 238–249. https://doi.org/10.1037/str0000108
- Baltes, B. B., & Heydens-Gahir, H. A. (2003). Reduction of work-family conflict through the use of selection, optimization, and compensation behaviors. *Journal of Applied Psychology*, 88(6), 1005–1018. https://doi.org/10.1037/0021-9010.88.6.1005
- Baumeister R.F., Heatherton T.F., Tice D.M. (1994). *Losing control: How and why people fail at self-regulation*. San Diego, CA: Academic Press.
- Baumeister, R. F. (1991). Meanings of Life. The Guilford Press.
- Baumeister, R. F. (2008). Free will in scientific psychology. *Perspectives on Psychological Science*, *3*(1), 14–19. https://doi.org/10.1111/j.1745-6916.2008.00057.x

- Baumeister, R. F., & Alquist, J. L. (2009). Is there a downside to good self-control? *Self and Identity*, 8(2–3), 115–130. https://doi.org/10.1080/15298860802501474
- Baumeister, R. F., & Heatherton, T. F. (1996). Self-regulation failure: An overview. *Psychological Inquiry*, 7(1), 1–15. https://doi.org/10.1207/s15327965pli0701_1
- Baumeister, R. F., & Vohs, K. D. (2007). Self-regulation, ego depletion, and motivation. *Social and Personality Psychology Compass*, *1*(1), 115–128. https://doi.org/https://doi.org/10.1111/j.1751-9004.2007.00001.x
- Baumeister, R. F., Bratslavsky, E., Finkenauer, C., & Vohs, K. D. (2001). Bad is stronger than good. *Review of General Psychology*, 5(4), 323–370. https://doi.org/10.1037/1089-2680.5.4.323
- Baumeister, R. F., Bratslavsky, E., Muraven, M., & Tice, D. M. (1998). Ego depletion: Is the active self a limited resource? *Journal of Personality and Social Psychology*, 74(5), 1252–1265. https://doi.org/10.1037/0022-3514.74.5.1252
- Baumeister, R. F., Tice, D. M., & Vohs, K. D. (2018). The strength model of self-regulation: Conclusions from the second decade of willpower research. *Perspectives on Psychological Science*, 13(2), 141–145. https://doi.org/10.1177/1745691617716946
- Baumeister, R. F., Vohs, K. D., & Tice, D. M. (2007). The strength model of self-control. *Current Directions in Psychological Science*, 16(6), 351–355. https://doi.org/10.1111/j.1467-8721.2007.00534.x
- Beames, J. R., Schofield, T. P., & Denson, T. F. (2017). A meta-analysis of improving selfcontrol with practice. In *The Routledge international handbook of self-control in health and well-being* (pp. 405–417). Routledge. https://doi.org/10.4324/9781315648576-32
- Beham, B., & Drobnič, S. (2011). Job demands and work-home interference: Empirical evidence from service sector employees in eight European countries. In S. Drobnič & A. Guillén (Eds.) Work-life balance in Europe: The role of job quality (pp. 95-119). Palgrave Macmillan.
- Berkman, E. T., Hutcherson, C. A., Livingston, J. L., Kahn, L. E., & Inzlicht, M. (2017). Selfcontrol as value-based choice. *Current Directions in Psychological Science*, 26(5), 422– 428. https://doi.org/10.1177/0963721417704394
- Berkman, L. F. (2016). Self-regulation training. In K. D. Vohs & R. F. Baumeister (Eds.), Handbook of self-regulation: Research, theory and applications (3rd ed.). Guilford.
- Biswas, T., Mäkelä, L., & Andresen, M. (2022). Work and non-work-related antecedents of expatriates' well-being: A meta-analysis. *Human Resource Management Review*, *32*(3). https://doi.org/10.1016/j.hrmr.2021.100889
- Boals, A., vanDellen, M. R., & Banks, J. B. (2011). The relationship between self-control and health: The mediating effect of avoidant coping. *Psychology & Health*, *26*(8), 1049–1062. https://doi.org/10.1080/08870446.2010.529139

- Boehnke, M. (2011). Gender role attitudes around the Globe: Egalitarian vs. traditional views. *Asian Journal of Social Science*, 39(1), 57–74. https://doi.org/ 10.1163/156853111X554438
- Bolger, N., & Zuckerman, A. (1995). A framework for studying personality in the stress process. *Journal of Personality and Social Psychology*, *69*(5), 890–902. https://doi.org/10.1037/0022-3514.69.5.890
- Booth-LeDoux, S. M., Matthews, R. A., & Wayne, J. H. (2020). Testing a resource-based spillover-crossover-spillover model: Transmission of social support in dual-earner couples. *Journal of Applied Psychology*, 105(7), 732-747. https://doi.org/10.1037/ap10000460
- Brassey, J., Hartenstein, L., Jeffery, B., & Simon, P. (2024). Working nine to thrive. McKinsey Health Institute. Retrieved from https://www.mckinsey.com/mhi/our-insights/workingnine-to-thrive
- Brislin, R. W. (1970). Back-Translation for Cross-Cultural Research. Journal of Cross-Cultural Psychology, 1(3), 185–216. https://doi.org/10.1177/135910457000100301
- Brüderl, J., Drobnič, S., Hank, K., Neyer, F. J., Walper, S., Alt, P., Borschel, E., . . . Wetzel, M. (2021). The German Family Panel (Pairfam). GESIS, Cologne. ZA5678 Data file Version 12.0.0. https://doi.org/10.4232/pairfam.5678.12.0.0
- Brunstein, J. C. (1993). Personal goals and subjective well-being: A longitudinal study. *Journal of Personality and Social Psychology*, 65(5), 1061–1070. https://doi.org/10.1037/0022-3514.65.5.1061
- Byron, K. (2005). A meta-analytic review of work–family conflict and its antecedents. *Journal* of Vocational Behavior, 67(2), 169–198. https://doi.org/https://doi.org/10.1016/j.jvb.2004.08.009
- Cantor, N. (1990). From thought to behavior: "Having" and "doing" in the study of personality and cognition. *American Psychologist*, 45(6), 735–750. https://doi.org/10.1037/0003-066X.45.6.735
- Cantor, N., & Blanton, H. (1996). Effortful pursuit of personal goals in daily life. In *The psychology of action: Linking cognition and motivation to behavior.* (pp. 338–359). The Guilford Press.
- Carlson, D. S., & Grzywacz, J. G. (2008). Reflections and future directions on measurement in work-family research. In K. Korabik, D. S. Lero, & D. L. Whitehead (Eds.), *Handbook* of work-family integration: Research, theory, and best practices (pp. 57-73). Academic Press. https://doi.org/10.1016/B978-012372574-5.50007-7
- Carlson, D. S., & Kacmar, K. M. (2000). Work–Family Conflict in the Organization: Do Life Role Values make a Difference? *Journal of Management*, 26(5), 1031-1054. https://doi.org/10.1177/014920630002600502
- Carlson, D. S., Kacmar, K. M., & Williams, L. J. (2000). Construction and initial validation of a multidimensional measure of work–family conflict. *Journal of Vocational Behavior*, 56(2), 249–276. https://doi.org/10.1006/JVBE.1999.1713

- Carr, J. C., Boyar, S. L., & Gregory, B. T. (2008). The moderating effect of work—family centrality on work—family conflict, organizational attitudes, and turnover behavior. *Journal of Management*, 34(2), 244-262. https://doi.org/10.1177/0149206307309262
- Chen, Y., Gao, L., Lippke, S., Xiang, Z., & Gan, Y. (2021). Harmonious personality and work-family conflicts: The multiple mediating roles of social support and selfcontrol. *PsyCh Journal*, 10(6), 889–897. https://doi.org/10.1002/pchj.476
- Cheung, F., Tang, C. S.-k., & Tang, S. (2011). Psychological capital as a moderator between emotional labor, burnout, and job satisfaction among school teachers in China. *International Journal of Stress Management*, 18(4), 348– 371. https://doi.org/10.1037/a0025787
- Choi, Y. E., Cho, E., Jung, H. J., & Sohn, Y. W. (2018). Calling as a predictor of life satisfaction: the roles of psychological capital, work–family enrichment, and boundary management strategy. *Journal of Career Assessment*, 26(4), 567-582. https://doi.org/10.1177/1069072717723092
- Cinamon, R. G., & Rich, Y. (2002). Gender differences in the importance of work and family roles: Implications for work-family conflict. *Sex Roles*, 47(11–12), 531–541. https://doi.org/10.1023/A:1022021804846
- Clinton, M. E., Conway, N., Sturges, J., & Hewett, R. (2020). Self-control during daily work activities and work-to-nonwork conflict. *Journal of Vocational Behavior*, 118, 103410. https://doi.org/10.1016/j.jvb.2020.103410
- Cohen, A. (2009). Individual values and the work/family interface. *Journal of Managerial Psychology*, *24*(8), 814–832. https://doi.org/10.1108/02683940910996815
- Colby, P. M. (1996). Individual differences in identity development status: Social and cognitive correlates of goal integration in early adulthood. University of California.
- Colby, P. M., Rabin, N., & Emmons, R. A. (1994). *Intimacy strivings and well-being: The role of social support*. Poster presented at the 102nd Annual Convention of the American Psychological Association, Los Angeles, CA.
- Das, K. V., Jones-Harrell, C., Fan, Y., Ramaswami, A., Orlove, B., & Botchwey, N. (2020). Understanding subjective well-being: perspectives from psychology and public health. *Public Health Reviews*, 41(1), 1–32. https://doi.org/10.1186/S40985-020-00142-5
- de Ridder, D. T. D., Lensvelt-Mulders, G., Finkenauer, C., Stok, F. M., & Baumeister, R. F. (2011). Taking stock of self-control: a meta-analysis of how trait self-control relates to a wide range of behaviors. *Personality and Social Psychology Review*, 16(1), 76–99. https://doi.org/10.1177/1088868311418749
- de Simone, S., Lampis, J., Lasio, D., Serri, F., Cicotto, G., & Putzu, D. (2014). Influences of work-family interface on job and life satisfaction. *Applied Research in Quality of Life*, 9(4), 831–861. https://doi.org/10.1007/s11482-013-9272-4
- Deci, E. L., & Ryan, R. M. (2000). The "what" and "why" of goal pursuits: Human needs and the self-determination of behavior. *Psychological Inquiry*, *11*(4), 227–268. https://doi.org/10.1207/S15327965PLI1104_01

- Del Líbano, M., Llorens, S., Salanova, M., & Schaufeli, W. B. (2012). About the dark and bright sides of self-efficacy: Workaholism and work engagement. *The Spanish Journal of Psychology*, *15*(2), 688–701. https://doi.org/10.5209/rev_SJOP.2012.v15.n2.38883
- Deloitte. (2022). Employee experience trends report: 2022.
- Diekman, A. B., & Eagly, A. H. (2008). Of men, women, and motivation: A role congruity account. In J. Y. Shah & W. L. Gardner (Eds.), *Handbook of motivation science* (pp. 434–447). The Guilford Press.
- Diekman, A. B., Brown, E. R., Johnston, A. M., & Clark, E. K. (2010). Seeking congruity between goals and roles: a new look at why women opt out of science, technology, engineering, and mathematics careers. *Psychological Science*, 21(8), 1051-1057. https://doi.org/10.1177/0956797610377342
- Diekman, A. B., Joshi, M. P., & Benson-Greenwald, T. M. (2020). Goal congruity theory: Navigating the social structure to fulfill goals. In B. Gawronski (Ed.), Advances in experimental social psychology (Vol. 62, pp. 189–244). Academic Press. https://doi.org/https://doi.org/10.1016/bs.aesp.2020.04.003
- Diekman, A. B., Steinberg, M., Brown, E. R., Belanger, A. L., & Clark, E. K. (2017). A goal congruity model of role entry, engagement, and exit: understanding communal goal processes in stem gender gaps. *Personality and Social Psychology Review*, 21(2), 142– 175. https://doi.org/10.1177/1088868316642141
- Diener, E. (1984). Subjective well-being. *Psychological Bulletin*, 95(3), 542–575. https://doi.org/10.1037/0033-2909.95.3.542
- Diener, E., & Diener, M. (1995). Cross-cultural correlates of life satisfaction and self-esteem. *Journal of Personality and Social Psychology*, *68*(4), 653–663. https://doi.org/10.1037/0022-3514.68.4.653
- Diener, E., & Fujita, F. (1995). Resources, personal strivings, and subjective well-being: a nomothetic and idiographic approach. *Journal of Personality and Social Psychology*, 68(5), 926–935. https://doi.org/10.1037/0022-3514.68.5.926
- Diener, E., & Ryan, K. (2009). Subjective well-being: A general overview. South African Journal of Psychology, 39(4), 391–406. https://doi.org/10.1177/008124630903900402
- Diener, E., Emmons, R. A., Larsen, R. J., & Griffin, S. (1985). The satisfaction with life scale. *Journal of Personality Assessment, 49*(1), 71–75. https://doi.org/10.1207/s15327752jpa4901_13
- Diener, E., Lucas, R. E., & Oishi, S. (2002). Subjective well-being: The science of happiness and life satisfaction. In C. R. Snyder & S. J. Lopez (Eds.), *Handbook of positive psychology*. Oxford University Press.
- Diener, E., Oishi, S., & Tay, L. (2018). Advances in subjective well-being research. *Nature Human Behaviour*, 2(4), 253–260. https://doi.org/10.1038/s41562-018-0307-6

- Drobnič, S., Beham, B., & Präg, P. (2010). Good job, good life? Working conditions and quality of life in Europe. *Social Indicators Research*, 99(2), 205-225. https://doi.org/10.1007/s11205-010-9586-7
- Duckworth, A., & Gross, J. J. (2014). Self-control and grit: Related but separable determinants of success. *Current Directions in Psychological Science*, 23(5), 319–325. https://doi.org/10.1177/0963721414541462
- Eagly, A. H., & Wood, W. (2012). Social role theory. In P. A. M. Van Lange, A. W. Kruglanski, & E. T. Higgins (Eds.), *Handbook of Theories of Social Psychology* (pp. 458–476). Sage Publications Ltd. https://doi.org/10.4135/9781446249222.n49
- Eagly, A. H., Nater, C., Miller, D. I., Kaufmann, M., & Sczesny, S. (2020). Gender stereotypes have changed: A cross-temporal meta-analysis of U.S. public opinion polls from 1946 to 2018. American Psychologist, 75(3), 301–315. https://doi.org/10.1037/amp0000494
- Eby, L. T., Casper, W. J., Lockwood, A., Bordeaux, C., & Brinley, A. (2005). Work and family research in IO/OB: Content analysis and review of the literature (1980–2002). *Journal of Vocational Behavior*, *66*(1), 124–197. https://doi.org/10.1016/J.JVB.2003.11.003
- Elliot, A. J. (1999). Approach and avoidance motivation and achievement goals. *Educational Psychologist*, *34*(3), 169–189. https://doi.org/10.1207/S15326985EP3403_3
- Elloy, D. F., & Smith, C. (2004). Antecedents of work-family conflict among dual-career couples: an Australian study. *Cross Cultural Management: An International Journal*, 11(4), 17–27. https://doi.org/10.1108/13527600410797927
- Elman, M., & Gilbert, L. (1984). Coping strategies for role conflict in married professional women with children. *Family Relations*, 33, 317. https://doi.org/10.2307/583799
- Elo, A.-L., Leppänen, A., & Jahkola, A. (2003). Validity of a single-item measure of stress symptoms. *Scandinavian Journal of Work, Environment & Health*, 29(6), 444–451. https://doi.org/10.5271/sjweh.752
- Emmons, R. A. (1986). Personal strivings. An approach to personality and subjective wellbeing. *Journal of Personality and Social Psychology*, 51(5), 1058–1068. https://doi.org/10.1037/0022-3514.51.5.1058
- Emmons, R. A. (1989). The personal striving approach to personality. In *Goal concepts in personality and social psychology*. (pp. 87–126). Lawrence Erlbaum Associates, Inc.
- Emmons, R. A. (1991). Personal strivings, daily life events, and psychological and physical well-being. *Journal of Personality*, 59(3), 453–472. https://doi.org/10.1111/J.1467-6494.1991.TB00256.X
- Emmons, R. A. (1996). Striving and feeling: Personal goals and subjective well-being. In P.M. Gollwitzer & J. A. Bargh (Eds.), *The psychology of action: Linking cognition and motivation to behavior* (pp. 313–337). The Guilford Press.
- Emmons, R. A. (2003). Personal goals, life meaning, and virtue: Wellsprings of a positive life.In C. L. M. Keyes & J. Haidt (Eds.), *Flourishing: Positive psychology and the life well-*
lived. (pp. 105–128). American Psychological Association. https://doi.org/10.1037/10594-005

- Emmons, R. A., & King, L. A. (1988). Conflict among personal strivings: Immediate and longterm implications for psychological and physical well-being. *Journal of Personality and Social Psychology*, 54(6), 1040–1048. https://doi.org/10.1037/0022-3514.54.6.1040
- Erarslan-Başkurt, A. B., & Aycan, Z. (2017). The work-family interface in Turkey. In K. Korabik, Z. Aycan, & R. Ayman (Eds.), *The work-family interface in global context* (pp. 175–194). Routledge.
- Erdoğan, I. (2009). The relationship between work-family conflict, life role salience and individual coping styles: A gender-sensitive study [Doctoral dissertation]. Marmara University.
- Externbrink, K., Diestel, S., & Krings, M. (2019). When do those high in trait self-control suffer from strain? *Journal of Personnel Psychology*, 18(1), 23-33. https://doi.org/10.1027/1866-5888/a000218
- Fishbach, A., Zhang, Y., & Koo, M. (2009). The dynamics of self-regulation. *European Review* of Social Psychology, 20(1), 315–344. https://doi.org/10.1080/10463280903275375
- Folkman, S. (1984). Personal control and stress and coping processes: A theoretical analysis. *Journal of Personality and Social Psychology*, 46(4), 839–852. https://doi.org/10.1037/0022-3514.46.4.839
- Ford, M. T., Heinen, B. A., & Langkamer, K. L. (2007). Work and family satisfaction and conflict: A meta-analysis of cross-domain relations. *Journal of Applied Psychology*, 92(1), 57–80. https://doi.org/10.1037/0021-9010.92.1.57
- Frear, K. A., Paustian-Underdahl, S. C., Heggestad, E. D., & Walker, L. S. (2019). Gender and career success: A typology and analysis of dual paradigms. *Journal of Organizational Behavior*, 40(4), 400–416. https://doi.org/10.1002/job.2338
- French, K. A., Dumani, S., Allen, T. D., & Shockley, K. M. (2018). A meta-analysis of work– family conflict and social support. *Psychological Bulletin*, 144(3), 284– 314. https://doi.org/10.1037/bul0000120
- Freund, A. M., & Baltes, P. B. (2002). Life-management strategies of selection, optimization and compensation: Measurement by self-report and construct validity. *Journal of Personality and Social Psychology*, 82(4), 642–662. https://doi.org/10.1037/0022-3514.82.4.642
- Friese, M., Frankenbach, J., Job, V., & Loschelder, D. D. (2017). Does self-control training improve self-control? A meta-analysis. *Perspectives on Psychological Science*, 12(6), 1077–1099. https://doi.org/10.1177/1745691617697076
- Frone, M. R., Russell, M., & Barnes, G. M. (1996). Work-family conflict, gender, and healthrelated outcomes: a study of employed parents in two community samples. *Journal of Occupational Health Psychology*, 1(1), 57–69. https://doi.org/10.1037//1076-8998.1.1.57

- Frone, M. R., Russell, M., & Cooper, M. L. (1992). Antecedents and outcomes of work-family conflict: Testing a model of the work-family interface. *Journal of Applied Psychology*, 77(1), 65–78. https://doi.org/10.1037/0021-9010.77.1.65
- Frone, M. R., Russell, M., & Cooper, M. L. (1992). Prevalence of work-family conflict: Are work and family boundaries asymmetrically permeable? *Journal of Organizational Behavior*, 13(7), 723-729. https://doi.org/10.1002/job.4030130708
- Frone, M. R., Russell, M., & Cooper, M. L. (1997). Relation of work–family conflict to health outcomes: A four-year longitudinal study of employed parents. *Journal of Occupational and Organizational Psychology*, 70(4), 325–335. https://doi.org/10.1111/J.2044-8325.1997.TB00652.X
- Frone, M. R., Yardley, J. K., & Markel, K. S. (1997). Developing and testing an integrative model of the work–family interface. *Journal of Vocational Behavior*, 50(2), 145-167. https://doi.org/10.1006/jvbe.1996.1577
- Gailliot, M. T., Baumeister, R. F., Dewall, C. N., Maner, J. K., Plant, E. A., Tice, D. M., Brewer, L. E., & Schmeichel, B. J. (2007). Self-control relies on glucose as a limited energy source: Willpower is more than a metaphor. *Journal of Personality and Social Psychology*, 92(2), 325–336. https://doi.org/10.1037/0022-3514.92.2.325
- Galinsky, A. D., Hall, E. V., & Cuddy, A. J. C. (2013). Gendered races: Implications for interracial marriage, leadership selection, and athletic participation. *Psychological Science*, 24, 498–506. https://doi.org/10.1177/0956797612457783
- Garcia, D., Anckarsäter, H., Kjell, O. N. E., Archer, T., Rosenberg, P., Cloninger, C. R., & Sikström, S. (2015). Agentic, communal, and spiritual traits are related to the semantic representation of written narratives of positive and negative life events. *Psychology of Well-Being*, 5(1), 8. https://doi.org/10.1186/s13612-015-0035-x
- Gebauer, J. E., Wagner, J., Sedikides, C., & Neberich, W. (2013). Agency-communion and self-esteem relations are moderated by culture, religiosity, age, and sex: evidence for the "self-centrality breeds self-enhancement" principle. *Journal of Personality*, 81(3), 261– 275. https://doi.org/10.1111/j.1467-6494.2012.00807.x
- Geurts, S.A.E. and Demerouti, E. (2003). Work/non-work interface: A review of theories and findings. In M. J. Schabracq, J. A. M. Winnubst and C. L. Cooper (Eds.), *The handbook* of work and health psychology. John Wiley & Sons. https://doi.org/10.1002/0470013400.ch14
- Gollwitzer, P. M., & Bargh, J. A. (Eds.). (1996). *The psychology of action: Linking cognition and motivation to behavior.* The Guilford Press.
- Gollwitzer, P. M., & Sheeran, P. (2006). Implementation intentions and goal achievement: A meta-analysis of effects and processes. In *Advances in experimental social psychology* (Vol. 38, pp. 69–119). Academic Press. https://doi.org/https://doi.org/10.1016/S0065-2601(06)38002-1
- Gollwitzer, P. M., Fujita, K., & Oettingen, G. (2004). Planning and the implementation of goals. In R. F. Baumeister & K. D. Vohs (Eds.), *Handbook of self-regulation: Research, theory, and applications* (pp. 211–228). The Guilford Press

- Goode, W. J. (1960). A theory of role strain. *American Sociological Review*, 25(4), 483–496. https://doi.org/10.2307/2092933
- Goschke, T., & Job, V. (2023). The willpower paradox: possible and impossible conceptions of self-control. *Perspectives on Psychological Science*, 18(6), 1339–1367. https://doi.org/10.1177/17456916221146158
- Grandey, A. A., & Cropanzano, R. (1999). The conservation of resources model applied to work–family conflict and strain. *Journal of Vocational Behavior*, 54(2), 350–370. https://doi.org/10.1006/jvbe.1998.1666
- Grant-Vallone, E. J., & Donaldson, S. I. (2001). Consequences of work-family conflict on employee well-being over time. *Work & Stress*, 15(3), 214–226. https://doi.org/10.1080/02678370110066544
- Grawitch, M. J., Barber, L. K., & Justice, L. (2010). Rethinking the work–life interface: it's not about balance, it's about resource allocation. *Applied Psychology: Health and Well-Being*, 2(2), 127–159. https://doi.org/10.1111/j.1758-0854.2009.01023.x
- Greenhaus, J. H., & Beutell, N. J. (1985). Sources of conflict between work and family roles. *The Academy of Management Review*, *10*(1), 76. https://doi.org/10.2307/258214
- Greenhaus, J. H., & Powell, G. N. (2006). When work and family are allies: A theory of workfamily enrichment. *The Academy of Management Review*, 31(1), 72–92. https://doi.org/10.2307/20159186
- Greenhaus, J. H., & Powell, G. N. (2017). Making work and family work: From hard choices to smart choices. New York, NY: Routledge. https://doi.org/10.4324/9781315780511
- Greenhaus, J. H., Parasuraman, S., & Collins, K. M. (2001). Career involvement and family involvement as moderators of relationships between work–family conflict and withdrawal from a profession. *Journal of Occupational Health Psychology*, 6(2), 91– 100. https://doi.org/10.1037/1076-8998.6.2.91
- Gröpel, P., & Kuhl, J. (2006). Having time for life activities: Life balance and selfregulation. Zeitschrift für Gesundheitspsychologie, 14(2), 54– 63. https://doi.org/10.1026/0943-8149.14.2.54
- Gutek, B. A., Searle, S., & Klepa, L. (1991). Rational versus gender role explanations for workfamily conflict. *Journal of Applied Psychology*, 76(4), 560–568. https://doi.org/10.1037/0021-9010.76.4.560
- Haar, J. M., Russo, M., Suñe, A., & Ollier-Malaterre, A. (2014). Outcomes of work–life balance on job satisfaction, life satisfaction and mental health: A study across seven cultures. *Journal of Vocational Behavior*, 85(3), 361–373. https://doi.org/10.1016/J.JVB.2014.08.010
- Hagger, M. S. (2015). Conservation of resources theory and the 'strength' model of selfcontrol: Conceptual overlap and commonalities. *Stress and Health*, 31(2), 89–94. https://doi.org/10.1002/SMI.2639

- Hagqvist, E., Gådin, K. G., & Nordenmark, M. (2017). Work–family conflict and well-being across Europe: The role of gender context. *Social Indicators Research*, *132*(2), 785–797. https://doi.org/10.1007/s11205-016-1301-x
- Haines, E. L., & Stroessner, S. J. (2019). The role prioritization model: How communal men and agentic women can (sometimes) have it all. *Social and Personality Psychology Compass*, 13(12). https://doi.org/10.1111/spc3.12504
- Hargis, M. B., Kotrba, L. M., Zhdanova, L., & Baltes, B. B. (2011). What really important? Examining the relative importance of antecedents to work-family conflict. *Journal of Managerial Issues*, 23(4), 386–408.
- Haring-Hidore, M., Stock, W. A., Okun, M. A., & Witter, R. A. (1985). Marital status and subjective well-being: A research synthesis. *Journal of Marriage and the Family*, 47(4), 947-953. https://www.jstor.org/stable/352338
- Hauser, A., Weisweiler, S., & Frey, D. (2018). Does the motivational orientation matter? A regulatory focus perspective work-life enrichment on and work-life conflict. International Journal of Stress Management, 25(4), 357 -376. https://doi.org/10.1037/str0000074
- Headey, B. (2008). Life goals matter to happiness: A revision of set-point theory. *Social Indicators Research*, *86*(2), 213–231. https://doi.org/10.1007/s11205-007-9138-y
- Helgeson, V. S. (1994). Relation of agency and communion to well-being: Evidence and potential explanations. *Psychological Bulletin*, 116(3), 412–428. https://doi.org/10.1037/0033-2909.116.3.412
- Helgeson, V. S., & Fritz, H. L. (1999). Unmitigated agency and unmitigated communion: distinctions from agency and communion. *Journal of Research in Personality*, 33(2), 131–158. https://doi.org/10.1006/jrpe.1999.2241
- Heller, D., Watson, D., & Ilies, R. (2004). The role of person versus situation in life satisfaction: a critical examination. *Psychological Bulletin*, 130(4), 574–600. https://doi.org/10.1037/0033-2909.130.4.574
- Helliwell, J. F. (2003). How's life? Combining individual and national variables to explain subjective well-being. *Economic Modelling*, 20(2), 331–360. https://doi.org/10.1016/S0264-9993(02)00057-3
- Hennecke, M., & Brandstätter, V. (2017). Means, Ends, and Happiness: The Role of Goals for Subjective Well-Being. In *The Happy Mind: Cognitive Contributions to Well-Being* (pp. 235–251). Springer International Publishing. https://doi.org/10.1007/978-3-319-58763-9_13
- Hetland, H., Saksvik, I. B., Albertsen, H., Berntsen, L. S., & Henriksen, A. (2012). "All work and no play..." Overcommitment and personality among university and college students. *College Student Journal*, 46, 470–482. Retrieved from http://search.ebscohost.com/login.aspx?direct=true&db=psyh&AN=2012-24487-002&site=ehost-live&scope=site

- Hirschi, A., Shockley, K. M., & Zacher, H. (2019). Achieving work-family balance: an action regulation model. *Academy of Management Review*, 44(1), 150–171. https://doi.org/10.5465/amr.2016.0409
- Hobfoll, S. E. (1989). Conservation of resources: A new attempt at conceptualizing stress. *American Psychologist*, 44(3), 513–524. https://doi.org/10.1037/0003-066X.44.3.513
- Hobfoll, S. E. (2011). Conservation of resource caravans and engaged settings. *Journal of Occupational and Organizational Psychology*, 84(1), 116-122. https://doi.org/10.1111/j.2044-8325.2010.02016.x
- Hobfoll, S. E. (2011). Conservation of resources theory: Its implication for stress, health, and resilience. In S. Folkman (Ed.), *The Oxford handbook of stress, health, and coping* (pp. 127–147). Oxford University Press. https://doi.org/10.1093/oxfordhb/9780195375343.013.0007
- Hobfoll, S. E., Halbesleben, J., Neveu, P., & Westman, M. (2018). Conservation of resources in the organizational context: The reality of resources and their consequences. *Annual Review of Organizational Psychology and Organizational Behavior*, 5, 103–28. https://doi.org/10.1146/annurev-orgpsych-032117-104640
- Hofmann, W., Baumeister, R. F., Förster, G., & Vohs, K. D. (2012). Everyday temptations: An experience sampling study of desire, conflict, and self-control. *Journal of Personality* and Social Psychology, 102(6), 1318–1335. https://doi.org/10.1037/a0026545
- Hofmann, W., Luhmann, M., Fisher, R. R., Vohs, K. D., & Baumeister, R. F. (2014). Yes, but are they happy? Effects of trait self-control on affective well-being and life satisfaction. *Journal of Personality*, 82(4), 265–277. https://doi.org/10.1111/jopy.12050
- Hofstede, G. (2001). Culture's consequences: Comparing values, behaviors, institutions and organizations across nations. Sage Publications.
- Hogan, R. (1982). A socioanalytic theory of personality. *Nebraska Symposium on Motivation*, 29, 55–89.
- Horowitz, J. L., & Garber, J. (2006). The prevention of depressive symptoms in children and adolescents: A meta-analytic review. *Journal of Consulting and Clinical Psychology*, 74(3), 401–415. https://doi.org/10.1037/0022-006X.74.3.401
- Huinink, J., & Schröder, T. (2003). Das Timing der Familiengründung. Antrag auf Gewährung einer Sachbeihilfe im Rahmen des DFG-Schwerpunktprogramms" Beziehungs-und Familienentwicklung.
- Imhoff, R., & Koch, A. (2017). How orthogonal are the big two of social perception? On the curvilinear relation between agency and communion. *Perspectives on Psychological Science*, 12(1), 122–137. https://doi.org/10.1177/1745691616657334
- Johnson, R. E., Lin, S.-H., & Lee, H. W. (2018). Self-Control as the Fuel for Effective Self-Regulation at Work: Antecedents, Consequences, and Boundary Conditions of Employee Self-Control. In A. J. Elliot (Ed.), Advances in motivation science (pp. 87–128). Elsevier Academic Press. https://doi.org/10.1016/bs.adms.2018.01.004

- Kagitcibasi, C. (1996). The autonomous-relational self. *European Psychologist*, 1(3), 180–186. https://doi.org/10.1027/1016-9040.1.3.180
- Kagitcibasi, C. (2005). Autonomy and relatedness in cultural context: Implications for self and family. *Journal of Cross-Cultural Psychology*, *36*(4), 403–422. https://doi.org/10.1177/0022022105275959
- Kahn, R. L., Wolfe, D. M., Quinn, R. P., Snoek, J. D., & Rosenthal, R. A. (1964). Organizational stress: Studies in role conflict and ambiguity. In *Organizational stress: Studies in role conflict and ambiguity*. John Wiley.
- Kahneman, D., Diener, E., & Schwarz, N. (1999). *Well-being: The foundations of hedonic psychology*. Russell Sage Foundation.
- Kalliath, P., Kalliath, T., & Chan, C. (2017). Work–family conflict, family satisfaction and employee well-being: a comparative study of Australian and Indian social workers. *Human Resource Management Journal*, 27(3), 366–381. https://doi.org/10.1111/1748-8583.12143
- Kasser, T., & Ryan, R. M. (1993). A dark side of the American dream: Correlates of financial success as a central life aspiration. *Journal of Personality and Social Psychology*, 65(2), 410–422. https://doi.org/10.1037/0022-3514.65.2.410
- Kasser, T., & Ryan, R. M. (2001). Be careful what you wish for: Optimal functioning and the relative attainment of intrinsic and extrinsic goals. In P. Schmuck & K. M. Sheldon (Eds.), *Life goals and well-being: Towards a positive psychology of human striving* (pp. 116–131). Hogrefe & Huber Publishers.
- Kaufman, G., & Taniguchi, H. (2019). Gender equality and work–family conflict from a crossnational perspective. International *Journal of Comparative Sociology*, 60(6), 385-408. https://doi.org/10.1177/0020715219893750
- Khatri, P., & Shukla, S. (2022). A review and research agenda of work-life balance: An agentic approach. *Community, Work & Family,* 1-35. https://doi.org/10.1080/13668803.2022.2127348
- Kiburz, K. M., Allen, T. D., & French, K. A. (2017). Work–family conflict and mindfulness: Investigating the effectiveness of a brief training intervention. *Journal of Organizational Behavior*, 38(7), 1016–1037. https://doi.org/10.1002/JOB.2181
- Kinnunen, U., & Mauno, S. (1998). Antecedents and outcomes of work–family conflict among employed women and men in Finland. *Human Relations*, 51(2), 157–177. https://doi.org/10.1023/A:1016962202639
- Kinnunen, U., Feldt, T., Geurts, S., & Pulkkinen, L. (2006). Types of work-family interface: Well-being correlates of negative and positive spillover between work and family. *Scandinavian Journal of Psychology*, 47(2), 149–162. https://doi.org/10.1111/j.1467-9450.2006.00502.x
- Kinnunen, U., Rantanen, J., Mauno, S., & Peeters, M. C. W. (2013). Work–family interaction. In *An introduction to contemporary work psychology* (pp. 267–289). https://doi.org/10.1002/9781119941467.ch11

- Kinnunen, U., Vermulst, A., Gerris, J., & Mäkikangas, A. (2003). Work–family conflict and its relations to well-being: the role of personality as a moderating factor. *Personality and Individual Differences*, 35(7), 1669–1683. https://doi.org/10.1016/S0191-8869(02)00389-6
- Kivetz, R., & Keinan, A. (2006). Repenting hyperopia: An analysis of self-control regrets. *Journal of Consumer Research*, 33(2), 273–282. https://doi.org/10.1086/506308
- Klinger, E. (1975). Consequences of commitment to and disengagement from incentives. *Psychological Review*, 82(1), 1–25. https://doi.org/10.1037/H0076171
- Klug, H. J. P., & Maier, G. W. (2015). Linking goal progress and subjective well-being: A meta-analysis. *Journal of Happiness Studies*, 16(1), 37–65. https://doi.org/10.1007/s10902-013-9493-0
- Knecht, M. K., Bauer, G. F., Gutzwiller, F., & Hämmig, O. (2011). Persistent work-life conflict and health satisfaction - a representative longitudinal study in Switzerland. *BMC Public Health*, 11, 271. https://doi.org/10.1186/1471-2458-11-271
- Kokkoris, M. D., Hoelzl, E., & Alós-Ferrer, C. (2019). True to which self? Lay rationalism and decision satisfaction in self-control conflicts. *Journal of Personality and Social Psychology*, 117(2), 417–447. https://doi.org/10.1037/pspp0000242
- Korabik, K. (2017). The role of work-family guilt in the work-family interface: A cross-cultural analysis. In K. Korabik, Z. Aycan, & R. Ayman (Eds.), *The work-family interface in global context* (pp. 368–391). Routledge.
- Kosakowska-Berezecka, N., Bosson, J. K., Jurek, P., Besta, T., Olech, M., Vandello, J. A., ... & van der Noll, J. (2023). Gendered self-views across 62 countries: A test of competing models. *Social Psychological and Personality Science*, 14(7), 808-824. https://doi.org/10.1177/19485506221129687
- Kossek, E. E. (2016). Managing work-life boundaries in the digital age. *Organizational Dynamics*, 45(3), 258–270. https://doi.org/10.1016/j.orgdyn.2016.07.010
- Kossek, E. E., & Ozeki, C. (1998). Work-family conflict, policies, and the job-life satisfaction relationship: A review and directions for organizational behavior-human resources research. *Journal of Applied Psychology*, 83(2), 139–149. https://doi.org/10.1037/0021-9010.83.2.139
- Kossek, E. E., Ruderman, M. N., Braddy, P. W., & Hannum, K. M. (2012). Work–nonwork boundary management profiles: A person-centered approach. *Journal of Vocational Behavior*, 81(1), 112-128. https://doi.org/10.1016/j.jvb.2012.04.003
- Koval, C. Z., van Dellen, M. R., Fitzsimons, G. M., & Ranby, K. W. (2015). The burden of responsibility: Interpersonal costs of high self-control. *Journal of Personality and Social Psychology*, 108(5), 750–766. https://doi.org/10.1037/pspi0000015
- Krampen, G. (2010). Evaluation of a program on systematic self-monitoring and reflection of health behavior in organisations: Results of two randomised controlled studies on wellbeing and absenteeism of employees and skilled workers. *Applied Psychology: Health and Well-Being*, 2(1), 105-125. https://doi.org/10.1111/j.1758-0854.2009.01025.x

- Kuroda, S., & Yamamoto, I. (2019). Why do people overwork at the risk of impairing mental health? *Journal of Happiness Studies: An Interdisciplinary Forum on Subjective Well-Being*, 20(5), 1519–1538. https://doi.org/10.1007/s10902-018-0008-x
- Kuśnierz, C., Rogowska, A. M., Chilicka, K., Pavlova, I., & Ochnik, D. (2022). Associations of work-family conflict with family-specific, work-specific, and well-being-related variables in a sample of Polish and Ukrainian adults during the second wave of the COVID-19 pandemic: A cross-sectional study. *International Journal of Environmental Research and Public Health*, 19(17), 10954. https://doi.org/10.3390/ijerph191710954
- Lanaj, K., Johnson, R. E., & Barnes, C. M. (2014). Beginning the workday yet already depleted? Consequences of late-night smartphone use and sleep. Organizational Behavior and Human Decision Processes, 124(1), 11–23. https://doi.org/10.1016/j.obhdp.2014.01.001
- Lapierre, L. M., & Allen, T. D. (2012). Control at work, control at home, and planning behavior: Implications for work–family conflict. *Journal of Management*, 38(5), 1500– 1516. https://doi.org/10.1177/0149206310385868
- Lavallee, L. F., & Campbell, J. D. (1995). Impact of personal goals on self-regulation processes elicited by daily negative events. *Journal of Personality and Social Psychology*, 69(2), 341–352. https://doi.org/10.1037/0022-3514.69.2.341
- Layton, R. L., & Muraven, M. (2014). Self-control linked with restricted emotional extremes. *Personality and Individual Differences, 58, 48–53.* https://doi.org/10.1016/j.paid.2013.10.004
- Lazarus, R. S. (1991). *Emotion and adaptation*. Oxford University Press. https://doi.org/10.2307/2075902
- Lazarus, R. S. (1999). Stress and emotion: A new synthesis. Springer Publishing Co.
- Lazarus, R. S., & Folkman, S. (1984). Stress, appraisal, and coping. Springer Publishing Co.
- Le, B. M., Impett, E. A., Lemay, E. P., Muise, A., & Tskhay, K. O. (2018). Communal motivation and well-being in interpersonal relationships: An integrative review and meta-analysis. *Psychological Bulletin*, 144(1), 1–25. https://doi.org/10.1037/bul0000133
- Li, V., Jiang, L., & Xu, X. V. (2020). From workplace mistreatment to job insecurity: The moderating effect of work centrality. *Stress and Health*, 36(3), 249–263. https://doi.org/10.1002/smi.2915
- Lippke, S., Derksen, C., Keller, F. M., Kötting, L., Schmiedhofer, M., & Welp, A. (2021). Effectiveness of communication interventions in Obstetrics—A systematic review. *International Journal of Environmental Research and Public Health*, 18(5), 2616. https://doi.org/10.3390/ijerph18052616
- Little, B. R. (1983). Personal projects: A rationale and method for investigation. *Environment and Behavior*, 15(3), 273–309. https://doi.org/10.1177/0013916583153002

- Livingston, B. A., & Judge, T. A. (2008). Emotional responses to work-family conflict: An examination of gender role orientation among working men and women. *Journal of Applied Psychology*, 93(1), 207–216. https://doi.org/10.1037/0021-9010.93.1.207
- Locke, E. A., & Kristof, A. L. (1996). Volitional choices in the goal achievement process. In *The psychology of action: Linking cognition and motivation to behavior*. (pp. 365–384). The Guilford Press.
- MacLeod, A. (2012). Goals and Plans: Their Relationship to Well-Being. In A. Efklides & D. Moraitou (Eds.), *A Positive Psychology Perspective on Quality of Life* (Vol. 51, pp. 33–50). Springer Publishing Co. https://doi.org/10.1007/978-94-007-4963-4_3
- Marks, S. R. (1977). Multiple roles and role strain: some notes on human energy, time and commitment. *American Sociological Review*, 42(6), 921. https://doi.org/10.2307/2094577
- Markus, H. R., & Kitayama, S. (1991). Culture and the self: Implications for cognition, emotion, and motivation. *Psychological Review*, 98(2), 224–253. https://doi.org/10.1037/0033-295X.98.2.224
- Marsiske, M., Lang, F. B., Baltes, P. B., & Baltes, M. M. (1995). Selective optimization with compensation: Life-span perspectives on successful human development. In R. A. Dixon & L. Bäckman (Eds.), *Compensating for psychological deficits and declines: Managing losses and promoting gains* (pp. 35–79). Lawrence Erlbaum Associates, Inc.
- Masuda, A. D., & Sortheix, F. M. (2012). Work-family values, priority goals and life satisfaction: A seven year follow-up of MBA students. *Journal of Happiness Studies: An Interdisciplinary Forum on Subjective Well-Being*, 13(6), 1131– 1144. https://doi.org/10.1007/s10902-011-9310-6
- Matias, M., & Fontaine, A. (2015). Coping with work and family: How do dual-earners interact? *Scandinavian Journal of Psychology*, 56 2, 212-22. https://doi.org/10.1111/sjop.12195
- Matthews, R. A., Swody, C. A., & Barnes-Farrell, J. L. (2012). Work hours and work–family conflict: The double-edged sword of involvement in work and family. *Stress and Health*, 28(3), 234-247. https://doi.org/10.1002/smi.1431
- Matthews, R. A., Wayne, J. H., & Ford, M. T. (2014). A work-family conflict/subjective wellbeing process model: A test of competing theories of longitudinal effects. *Journal of Applied Psychology*, 99(6), 1173–1187. https://doi.org/10.1037/a0036674
- Mauno, S., Kinnunen, U., & Ruokolainen, M. (2006). Exploring work- and organization-based resources as moderators between work-family conflict, well-being, and job attitudes. *Work & Stress*, 20(3), 210–233. https://doi.org/10.1080/02678370600999969
- Mayser, S., Scheibe, S., & Riediger, M. (2008). (Un)reachable? An empirical differentiation of goals and life longings. *European Psychologist*, 13(2), 126–140. https://doi.org/10.1027/1016-9040.13.2.126
- McClelland, D. C. (1985). Human motivation. Scott, Foresman and Comp.

- Mellner, C., Aronsson, G., & Kecklund, G. (2014). Boundary management preferences, boundary control, and work-life balance among full-time employed professionals in knowledge-intensive, flexible work. *Nordic Journal of Working Life Studies*, 4(4), 7– 23. https://doi.org/10.19154/njwls.v4i4.4705
- Mesmer-Magnus, J. R., & Viswesvaran, C. (2005). Convergence between measures of workto-family and family-to-work conflict: A meta-analytic examination. *Journal of Vocational Behavior*, 67(2), 215–232. https://doi.org/10.1016/J.JVB.2004.05.004
- Michel, J. S., Kotrba, L. M., Mitchelson, J. K., Clark, M. A., & Baltes, B. B. (2011). Antecedents of work–family conflict: A meta-analytic review. *Journal of Organizational Behavior*, 32(5), 689–725. https://doi.org/10.1002/job.695
- Michie, S., Richardson, M., Johnston, M., Abraham, C., Francis, J., Hardeman, W., Eccles, M. P., Cane, J., & Wood, C. E. (2013). The behavior change technique taxonomy (v1) of 93 hierarchically clustered techniques: Building an international consensus for the reporting of behavior change interventions. *Annals of Behavioral Medicine: A Publication of The Society of Behavioral Medicine*, 46(1), 81–95. https://doi.org/10.1007/s12160-013-9486-6
- Mickel, A. E., & Dallimore, E. J. (2009). Life-quality decisions: Tension-management strategies used by individuals when making tradeoffs. *Human Relations*, 62(5), 627–668. https://doi.org/10.1177/0018726709103453
- Miller, B. K., Wan, M., Carlson, D., Kacmar, K. M., & Thompson, M. (2022). Antecedents and outcomes of work-family conflict: A mega-meta path analysis. *PLOS ONE*, 17(2). https://doi.org/10.1371/journal.pone.0263631
- Mullen, J., Kelley, E., & Kelloway, E. K. (2008). Health and well-being outcomes of the work-family interface. In K. Korabik, D. S. Lero, & D. L. Whitehead (Eds.), *Handbook of work-family integration* (pp. 191–214). Academic Press. https://doi.org/https://doi.org/10.1016/B978-012372574-5.50014-4
- Müller, A., Heiden, B., Herbig, B., Poppe, F., & Angerer, P. (2016). Improving well-being at work: A randomized controlled intervention based on selection, optimization, and compensation. *Journal of Occupational Health Psychology*, 21(2), 169–181. https://doi.org/10.1037/a0039676
- Muraven, M. (2010). Building self-control strength: Practicing self-control leads to improved self-control performance. *Journal of Experimental Social Psychology*, 46(2), 465–468. https://doi.org/10.1016/J.JESP.2009.12.011
- Muraven, M., & Baumeister, R. F. (2000). Self-regulation and depletion of limited resources: Does self-control resemble a muscle? *Psychological Bulletin*, 126(2), 247–259. https://doi.org/10.1037/0033-2909.126.2.247
- Muraven, M., Baumeister, R. F., & Tice, D. M. (1999). Longitudinal improvement of self-regulation through practice: Building self-control strength through repeated exercise. *The Journal of Social Psychology*, 139(4), 446-457. https://doi.org/10.1080/00224549909598404

- Muraven, M., Shmueli, D., & Burkley, E. (2006). Conserving self-control strength. Journal of Personality and Social Psychology, 91(3), 524–537. https://doi.org/10.1037/0022-3514.91.3.524
- Myers, D. G., & Diener, E. (1995). Who is happy? *Psychological Science*, 6(1), 10–19. https://doi.org/10.1111/j.1467-9280.1995.tb00298.x
- Netemeyer, R. G., Boles, J. S., & McMurrian, R. (1996). Development and validation of work– family conflict and family–work conflict scales. *Journal of Applied Psychology*, 81(4), 400–410. https://doi.org/10.1037/0021-9010.81.4.400
- Ng, A. K., Ho, D. Y. F., Wong, S. S., & Smith, I. (2003). In search of the good life: A cultural odyssey in the East and West. *Genetic, Social, and General Psychology Monographs*, *129*(4), 317–363.
- Nohe, C., Meier, L. L., Sonntag, K., & Michel, A. (2015). The chicken or the egg? A metaanalysis of panel studies of the relationship between work–family conflict and strain. *Journal of Applied Psychology*, 100(2), 522–536. https://doi.org/10.1037/a0038012
- Noor, N. M. (2004). Work-family conflict, work-and family-role salience, and women's wellbeing. *The Journal of Social Psychology*, *144*(4), 389-406. https://doi.org/10.3200/SOCP.144.4.389-406
- Obrenovic, B., Jianguo, D., Khudaykulov, A., & Khan, M. A. S. (2020). Work-family conflict impact on psychological safety and psychological well-being: A job performance model. *Frontiers in Psychology*, *11*. https://doi.org/10.3389/fpsyg.2020.00475
- Oishi, S. (2000). Goals as cornerstones of subjective well-being: Linking individuals and cultures. In E. Diener & E. M. Suh (Eds.), *Culture and subjective well-being* (pp. 87–112). The MIT Press.
- Oishi, S., & Diener, E. (2001). Goals, culture, and subjective well-being. *Personality and Social Psychology Bulletin*, 27(12), 1674–1682. https://doi.org/10.1177/01461672012712010
- Organization for Economic Cooperation and Development. (2020). *OECD Better Life Index*. Retrieved from https://www.oecdbetterlifeindex.org
- Ouellette, J. A., & Wood, W. (1998). Habit and intention in everyday life: The multiple processes by which past behavior predicts future behavior. *Psychological Bulletin*, 124(1), 54–74. https://doi.org/10.1037/0033-2909.124.1.54
- Pavot, W., & Diener, E. (2008). The satisfaction with life scale and the emerging construct of life satisfaction. *The Journal of Positive Psychology*, 3(2), 137–152. https://doi.org/10.1080/17439760701756946
- Peng, A. C., Ilies, R., & Dimotakis, N. (2011). Work-Family Balance, Role Integration and Employee Well-Being. In S. Kaiser, M. Ringlstetter, D. Eikhof, & M. Pina e Cunha (Eds.), *Creating Balance?* (pp. 121–140). Springer. https://doi.org/10.1007/978-3-642-16199-5_7

- Podsakoff, P. M., MacKenzie, S. B., Lee, J.-Y., & Podsakoff, N. P. (2003). Common method biases in behavioral research: A critical review of the literature and recommended remedies. *Journal of Applied Psychology*, 88(5), 879–903. https://doi.org/10.1037/0021-9010.88.5.879
- Poelmans, S., Spector, P. E., Cooper, C. L., Allen, T. D., O'Driscoll, M., & Sanchez, J. I. (2003). A cross-national comparative study of work/family demands and resources. *International Journal of Cross Cultural Management*, 3(3), 275-288. https://doi.org/10.1177/1470595803003002
- Pöhlmann, K. (2001). Agency- and communion-orientation in life goals: Impacts on goal pursuit strategies and psychological well-being. In P. Schmuck & K. M. Sheldon (Eds.), *Life goals and well-being: Towards a positive psychology of human striving* (pp. 68–84). Hogrefe & Huber Publishers.
- Pöhlmann, K. , & Brunstein, J. C. (1997). GOALS: Ein Fragebogen zur Messung von Lebenszielen [GOALS: A questionnaire for assessing life goals]. *Diagnostica*, 43(1), 63– 79.
- Pöhlmann, K., Brunstein, J. C., Koch, R., Brähler, E., & Joraschky, P. (2010). Der Lebenszielfragebogen GOALS: Befunde zur internen und externen validität auf der basis einer repräsentativen bevölkerungsstichprobe und einer klinischen stichprobe [The Life Goal Questionnaire GOALS: Data on internal and external validity in a representative population survey and a clinical sample]. Zeitschrift für Medizinische Psychologie, 19(2), 70–80.
- Poortvliet, P. M., & Darnon, C. (2010). Toward a more social understanding of achievement goals. *Current Directions in Psychological Science*, *19*(5), 324–328. https://doi.org/10.1177/0963721410383246
- Prendergast, M., Podus, D., Finney, J., Greenwell, L., & Roll, J. (2006). Contingency management for treatment of substance use disorders: A meta-analysis. *Addiction*, 101(11), 1546–1560. https://doi.org/10.1111/j.1360-0443.2006.01581.x
- Reimann, M., & Diewald, M. (2022). Good mental health despite work-family conflict? The within-domain and cross-domain buffering potentials of family and work resources. *Journal of Family Research*, 34(4), 1126–1150. https://doi.org/10.20377/jfr-726
- Reinke, K., & Ohly, S. (2024). Examining the training design and training transfer of a boundary management training: A randomized controlled intervention study. *Journal of Occupational* and Organizational Psychology, 00, 1– 25. https://doi.org/10.1111/joop.12497
- Richardson, K. M. (2017). Managing employee stress and wellness in the new millennium. *Journal of Occupational Health Psychology*, 20, 423-428. https://doi.org/10.1037/ocp0000066
- Riva, E., Lucchini, M., & Russo, M. (2019). Societal gender inequality as moderator of the relationship between work–life fit and subjective well-being: A multilevel analysis across European countries. *Social Indicators Research*, 143(2), 657-691. https://doi.org/10.1007/s11205-018-1986-0

- Rothbard, N. P. (2001). Enriching or depleting? The dynamics of engagement in work and family roles. *Administrative Science Quarterly*, 46(4), 655–684. https://doi.org/10.2307/3094827
- Rothbard, N. P., & Edwards, J. R. (2003). Investment in work and family roles: A test of identity and utilitarian motives. *Personnel Psychology*, 56(3), 699– 730. https://doi.org/10.1111/j.1744-6570.2003.tb00755.x
- Rothbard, N. P., & Ollier-Malaterre, A. (2016). Boundary management. In T. D. Allen & L. T. Eby (Eds.), *The Oxford handbook of work and family* (pp. 109–122). Oxford University Press.
- Rubenstein, A. L., Morrison, H. M., Whiting, S. W., & Bosco, F. A. (2022). More money, more problems? An examination of the dynamic relationship between income and work– family conflict. *Journal of Occupational and Organizational Psychology*, 95(2), 305– 331. https://doi.org/10.1111/joop.12387
- Rubin, D.B. (1987). Multiple imputation for nonresponse in surveys. John Wiley & Sons Inc., New York. http://dx.doi.org/10.1002/9780470316696
- Rudman, L. A., Moss-Racusin, C. A., Phelan, J. E., & Nauts, S. (2012). Status incongruity and backlash effects: Defending the gender hierarchy motivates prejudice against female leaders. *Journal of Experimental Social Psychology*, 48(1), 165–179. https://doi.org/10.1016/j.jesp.2011.10.008
- Santos, H. C., Varnum, M. E. W., & Grossmann, I. (2017). Global increases in individualism. *Psychological Science*, 28(9), 1228-1239. https://doi.org/10.1177/0956797617700622
- Saragovi, C., Aubé, J., Koestner, R., & Zuroff, D. (2002). Traits, motives, and depressive styles as reflections of agency and communion. *Personality and Social Psychology Bulletin*, 28(5), 563–577. https://doi.org/10.1177/0146167202288001
- Saragovi, C., Koestner, R., Di Dio, L., & Aubé, J. (1997). Agency, communion, and wellbeing: Extending Helgeson's (1994) model. *Journal of Personality and Social Psychology*, 73(3), 593–609. https://doi.org/10.1037/0022-3514.73.3.593
- Schimmack, U., Radhakrishnan, P., Oishi, S., Dzokoto, V., & Ahadi, S. (2002). Culture, personality, and subjective well-being: Integrating process models of life satisfaction. *Journal of Personality and Social Psychology*, 82(4), 582–593. https://doi.org/10.1037/0022-3514.82.4.582
- Schlachter, S., McDowall, A., Cropley, M., & Inceoglu, I. (2018). Voluntary work-related technology use during non-work time: A narrative synthesis of empirical research and research agenda. *International Journal of Management Reviews*, 20(4), 825– 846. https://doi.org/10.1111/ijmr.12165
- Schmeichel, B. J., & Vohs, K. D. (2009). Self-affirmation and self-control: Affirming core values counteracts ego depletion. *Journal of Personality and Social Psychology*, 96(4), 770–782. https://doi.org/10.1037/A0014635
- Schmuck, P., & Sheldon, K. M. (Eds.). (2001). *Life goals and well-being: Towards a positive psychology of human striving*. Hogrefe & Huber Publishers.

- Schöllbauer, J., Hartner-Tiefenthaler, M., & Kelliher, C. (2021). ICT-enabled work extension and its consequences: A paradoxical situation between high performance and low wellbeing. In C. Korunka (Ed.), Flexible working practices and approaches (pp. 149– 165). Springer International Publishing. https://doi.org/10.1007/978-3-030-74128-0_8
- Schooreel, T., & Verbruggen, M. (2016). Use of family-friendly work arrangements and work– family conflict: Crossover effects in dual-earner couples. *Journal of Occupational Health Psychology*, 21(1), 119–132. https://doi.org/10.1037/a0039669
- Schwartz, S. H. (2012). An overview of the Schwartz theory of basic values. *Online Readings in Psychology and Culture*, 2(1). https://doi.org/10.9707/2307-0919.1116
- Schwarzer, R., Lippke, S., & Luszczynska, A. (2011). Mechanisms of health behavior change in persons with chronic illness or disability: The health action process approach (HAPA). *Rehabilitation Psychology*, 56(3), 161–170. https://doi.org/10.1037/A0024509
- Sczesny, S., Nater, C., & Eagly, A. H. (2018). Agency and communion: Their implications for gender stereotypes and gender identities. In A. E. Abele & B. Wojciszke (Eds.), Agency and Communion in Social Psychology (pp. 103–116). Routledge. https://doi.org/10.4324/9780203703663-9
- Sedikides, C., Gaertner, L., & Toguchi, Y. (2003). Pancultural self-enhancement. Journal of Personality and Social Psychology, 84(1), 60–79. https://doi.org/10.1037/0022-3514.84.1.60
- Seiger, C. P., & Wiese, B. S. (2009). Social support from work and family domains as an antecedent or moderator of work-family conflicts? *Journal of Vocational Behavior*, 75(1), 26–37. https://doi.org/10.1016/j.jvb.2009.03.001
- Senécal, C., Vallerand, R. J., & Guay, F. (2001). Antecedents and outcomes of work-family conflict: Toward a motivational model. *Personality and Social Psychology Bulletin*, 27(2), 176-186. https://doi.org/10.1177/0146167201272004
- Shaffer, M. A., Joplin, J. R. W., & Hsu, Y.-S. (2011). Expanding the boundaries of work family research: A review and agenda for future research. *International Journal of Cross Cultural Management*, 11(2), 221-268. https://doi.org/10.1177/1470595811398800
- Sheldon, K. M. (2018). Understanding the good life: Eudaimonic living involves well-doing, not well-being. In J. P. Forgas & R. F. Baumeister (Eds.), *The Social Psychology of Living Well* (pp. 116–136). New York: Routledge.
- Sheldon, K. M., & Cooper, M. L. (2008). Goal striving within agentic and communal roles: Separate but functionally similar pathways to enhanced well-being. *Journal of Personality*, 76(3), 415–448. https://doi.org/10.1111/j.1467-6494.2008.00491.x
- Sheldon, K. M., & Elliot, A. J. (1999). Goal striving, need satisfaction, and longitudinal wellbeing: The self-concordance model. *Journal of Personality and Social Psychology*, 76(3), 482–497. https://doi.org/10.1037/0022-3514.76.3.482
- Sheldon, K. M., & Kasser, T. (1995). Coherence and congruence: Two aspects of personality integration. *Journal of Personality and Social Psychology*, 68(3), 531–543. https://doi.org/10.1037/0022-3514.68.3.531

- Sheldon, K. M., Abad, N., Ferguson, Y., Gunz, A., Houser-Marko, L., Nichols, C. P., & Lyubomirsky, S. (2010). Persistent pursuit of need-satisfying goals leads to increased happiness: A 6-month experimental longitudinal study. *Motivation and Emotion*, 34(1), 39–48. https://doi.org/10.1007/s11031-009-9153-1
- Sheldon, K. M., Elliot, A. J., Ryan, R. M., Chirkov, V., Kim, Y., Wu, C., Demir, M., & Sun, Z. (2004). Self-concordance and subjective well-being in four cultures. *Journal of Cross-Cultural Psychology*, 35(2), 209-223. https://doi.org/10.1177/0022022103262245
- Sheldon, K. M., Ryan, R. M., Deci, E. L., & Kasser, T. (2004). The independent effects of goal contents and motives on well-being: It's both what you pursue and why you pursue it. *Personality and Social Psychology Bulletin, 30*(4), 475-486. https://doi.org/10.1177/0146167203261883
- Shockley, K. M., & Singla, N. (2011). Reconsidering work—family interactions and satisfaction: A meta-analysis. *Journal of Management*, 37(3), 861–886. https://doi.org/10.1177/0149206310394864
- Shockley, K. M., Douek, J., Smith, C. R., Yu, P. P., Dumani, S., & French, K. A. (2017). Crosscultural work and family research: A review of the literature. *Journal of Vocational Behavior*, 101, 1–20. https://doi.org/10.1016/j.jvb.2017.04.001
- Shockley, K. M., Shen, W., DeNunzio, M. M., Arvan, M. L., & Knudsen, E. A. (2017). Disentangling the relationship between gender and work-family conflict: An integration of theoretical perspectives using meta-analytic methods. *Journal of Applied Psychology*, 102(12), 1601–1635. https://doi.org/10.1037/apl0000246
- Sonnentag, S., & Fritz, C. (2007). The Recovery Experience Questionnaire: Development and validation of a measure for assessing recuperation and unwinding from work. *Journal of Occupational Health Psychology*, 12(3), 204–221. https://doi.org/10.1037/1076-8998.12.3.204
- Spector, P. E., Allen, T. D., Poelmans, S. A., Lapierre, L. M., Cooper, C. L., Michael, O. D., ... & Widerszal-Bazyl, M. (2007). Cross-national differences in relationships of work demands, job satisfaction, and turnover intentions with work–family conflict. *Personnel Psychology*, 60(4), 805–835. https://doi.org/10.1111/j.1744-6570.2007.00092.x
- Stahnke, B., & Cooley, M. (2021). A systematic review of the association between partnership and life satisfaction. *The Family Journal*, 29(2), 182–189. https://doi.org/10.1177/1066480720977517
- Steca, P., Monzani, D., Greco, A., D'Addario, M., Cappelletti, E., & Pancani, L. (2016). The effects of short-term personal goals on subjective well-being. *Journal of Happiness Studies*, 17(4), 1435–1450. https://doi.org/10.1007/s10902-015-9651-7
- Stubbe, J. H., Posthuma, D., Boomsma, D. I., & De Geus, E. J. (2005). Heritability of life satisfaction in adults: A twin-family study. *Psychological Medicine*, 35(11), 1581-1588. https://doi.org/10.1017/S0033291705005374
- Suh, E. M., & Oishi, S. (2002). Subjective well-being across cultures. *Online Readings in Psychology and Culture*, *10*(1), 1. https://doi.org/10.9707/2307-0919.1076

- Sun, S. H., & Frese, M. (2013). Multiple goal pursuit. In *New developments in goal setting and task performance*. (pp. 177–194). Routledge. https://doi.org/10.4324/9780203082744
- Tangney, J. P., Baumeister, R. F., & Boone, A. L. (2004). High self-control predicts good adjustment, less pathology, better grades, and interpersonal success. *Journal of Personality*, 72(2), 271–324. https://doi.org/10.1111/j.0022-3506.2004.00263.x
- ten Brummelhuis, L. L., & Bakker, A. B. (2012). A resource perspective on the work-home interface: The work-home resources model. *American Psychologist*, 67(7), 545–556. https://doi.org/10.1037/a0027974
- Thoits, P. A. (1994). Stressors and problem-solving: the individual as psychological activist. *Journal of Health and Social Behavior*, 35(2), 143–160. https://doi.org/10.2307/2137362
- Tóth, Á., Wisse, B., & Faragó, K. (2018). The impact of goal attainment and goal importance on satisfaction with life a polynomial regression and response surface analysis. *Mentálhigiéné És Pszichoszomatika*, 19(1), 80–101. https://doi.org/10.1556/0406.19.2018.004
- Trapnell, P. D., & Paulhus, D. L. (2012). Agentic and communal values: Their scope and measurement. *Journal of Personality Assessment*, 94(1), 39–52. https://doi.org/10.1080/00223891.2011.627968
- Triandis, H. C. (1995). Individualism & collectivism. Westview Press.
- Triandis, H.C. (2001). Individualism-collectivism and personality. *Journal of Personality*, 69, 907-924. https://doi.org/10.1111/1467-6494.696169
- Tucak Junaković, I. (2015). The contribution of life goals to some aspects of mental well-being and health. *Medica Jadertina*, 45(3–4), 87–95.
- Uchronski, M. (2008), Agency and communion in spontaneous self-descriptions: Occurrence and situational malleability. *European Journal of Social Psychology, 38*, 1093-1102. https://doi.org/10.1002/ejsp.563
- Unsworth, K., Yeo, G., & Beck, J. (2014). Multiple goals: A review and derivation of general principles. *Journal of Organizational Behavior*, *35*(8), 1064–1078. https://doi.org/10.1002/job.1963
- Ursin, H., & Eriksen, H. R. (2010). Cognitive activation theory of stress (CATS). *Neuroscience* & *Biobehavioral Reviews*, 34, 877– 881. https://doi.org/10.1016/j.neubiorev.2009.03.001
- Uskul, A. K., Kirchner-Häusler, A., Vignoles, V. L., Rodriguez-Bailón, R., Castillo, V. A., Cross, S. E., Yalçın, M. G., Harb, C., Husnu, S., Ishii, K., Jin, S., Karamaouna, P., Kafetsios, K., Kateri, E., Matamoros-Lima, J., Liu, D., Miniesy, R., Na, J., Özkan, Z., . . Uchida, Y. (2023). Neither Eastern nor Western: Patterns of independence and interdependence in Mediterranean societies. *Journal of Personality and Social Psychology*, 125(3), 471–495. https://doi.org/10.1037/pspa0000342
- Virtanen, M., Jokela, M., Madsen, I. E., Magnusson Hanson, L. L., Lallukka, T., Nyberg, S. T., Alfredsson, L., Batty, G. D., Bjorner, J. B., Borritz, M., Burr, H., Dragano, N., Erbel,

R., Ferrie, J. E., Heikkilä, K., Knutsson, A., Koskenvuo, M., Lahelma, E., Nielsen, M. L., Oksanen, T., ... Kivimäki, M. (2018). Long working hours and depressive symptoms: systematic review and meta-analysis of published studies and unpublished individual participant data. *Scandinavian Journal of Work, Environment & Health, 44*(3), 239–250. https://doi.org/10.5271/sjweh.3712

- Vohs, K. D., & Baumeister, R. F. (2004). Understanding self-regulation: An introduction. In R. F. Baumeister & K. D. Vohs (Eds.), *Handbook of self-regulation: Research, Theory,* and Applications (pp. 1-9). New York, NY: The Guilford Press
- Vohs, K. D., Baumeister, R. F., Schmeichel, B. J., Twenge, J. M., Nelson, N. M., & Tice, D. M. (2008). Making choices impairs subsequent self-control: a limited-resource account of decision making, self-regulation, and active initiative. *Journal of Personality and Social Psychology*, 94(5), 883–898. https://doi.org/10.1037/0022-3514.94.5.883
- Volpp, K. G., John, L. K., Troxel, A. B., Norton, L., Fassbender, J., & Loewenstein, G. (2008). Financial incentive-based approaches for weight loss: A randomized trial. *JAMA*, 300(22), 2631–2637. https://doi.org/10.1001/jama.2008.804
- Wadsworth, M., & Ford, D. H. (1983). Assessment of personal goal hierarchies. *Journal of Counseling Psychology*, 30(4), 514–526. https://doi.org/10.1037/0022-0167.30.4.514
- Wang, Y., Liu, L., Wang, J., & Wang, L. (2012). Work-family conflict and burnout among Chinese doctors: the mediating role of psychological capital. *Journal of Occupational Health*, 54(3), 232–240. https://doi.org/10.1539/joh.11-0243-oa
- Wei, H., Wu, B., Park, H., & Deng, C. (2022). Is exercise good for all? Time- and strain-based work–family conflict and its impacts. *The Journal of Social Psychology*, 163(2), 230– 247. https://doi.org/10.1080/00224545.2022.2111251
- Wicker, F. W., Lambert, F. B., Richardson, F. C., & Kahler, J. (1984). Categorical goal hierarchies and classification of human motives. *Journal of Personality*, *52*(3), 285–305. https://doi.org/10.1111/j.1467-6494.1984.tb00883.x
- Wiggins, J. S. (1991). Agency and communion as conceptual coordinates for the understanding and measurement of interpersonal behavior. In D. Cicchetti & W. M. Grove (Eds.), *Thinking clearly about psychology: Essays in honor of Paul E. Meehl, Vol. 1: Matters of public interest; Vol. 2: Personality and psychopathology.* (pp. 89–113). University of Minnesota Press.
- Williams, G. C., McGregor, H. A., Sharp, D., Levesque, C., Kouides, R. W., Ryan, R. M., & Deci, E. L. (2006). Testing a self-determination theory intervention for motivating tobacco cessation: Supporting autonomy and competence in a clinical trial. *Health Psychology*, 25(1), 91–101. https://doi.org/10.1037/0278-6133.25.1.91
- Wojciszke, B. (1997). Parallels between competence- versus morality-related traits and individualistic versus collectivistic values. *European Journal of Social Psychology*, 27(3), 245–256. https://doi.org/10.1002/(SICI)1099-0992(199705)27:3<245::AID-EJSP819>3.0.CO;2-H

- Wojciszke, B., & Abele, A. E. (2008). The primacy of communion over agency and its reversals in evaluations. *European Journal of Social Psychology*, 38(7), 1139– 1147. https://doi.org/10.1002/ejsp.549
- Wojciszke, B., & Bialobrzeska, O. (2014). Agency versus Communion as Predictors of Selfesteem: Searching for the Role of Culture and Self-construal. *Polish Psychological Bulletin*, 45(4), 469–479. https://doi.org/10.2478/ppb-2014-0057
- Wojdylo, K., Baumann, N., & Kuhl, J. (2017). The firepower of work craving: When selfcontrol is burning under the rubble of self-regulation. *PLOS ONE*, 12(1). https://doi.org/10.1371/journal.pone.0169729
- Wolff, H.-G., & Höge, T. (2011). Konflikte zwischen Arbeit und Familie. Zeitschrift Für Arbeits- Und Organisationspsychologie A&O, 55(3), 143–152. https://doi.org/10.1026/0932-4089/a000053
- Wolff, W., Bieleke, M., Englert, C., Bertrams, A., Schüler, J., & Martarelli, C. S. (2022). A single item measure of self-control – validation and location in a nomological network of self-control, boredom, and if-then planning. *Social Psychological Bulletin*, 17. https://doi.org/10.32872/spb.7453
- Wood, W., & Eagly, A. H. (2012). Biosocial construction of sex differences and similarities in behavior. In *Advances in experimental social psychology* (pp. 55-123). Academic Press. https://doi.org/10.1016/B978-0-12-394281-4.00002-7.
- Wood, W., Christensen, P. N., Hebl, M. R., & Rothgerber, H. (1997). Conformity to sex-typed norms, affect, and the self-concept. *Journal of Personality and Social Psychology*, 73(3), 523–535. https://doi.org/10.1037/0022-3514.73.3.523
- World Economic Forum. (2023). Global gender gap report 2023. Retrieved from https://www.weforum.org/reports/global-gender-gap-report-2023
- Wrosch, C., Scheier, M. F., Carver, C. S., & Schulz, R. (2003). The importance of goal disengagement in adaptive self-regulation: When giving up is beneficial. *Self and Identity*, 2(1), 1–20. https://doi.org/10.1080/15298860309021
- Xanthopoulou, D., Bakker, A. B., Demerouti, E., & Schaufeli, W. B. (2007). The role of personal resources in the job demands-resources model. *International Journal of Stress Management*, 14(2), 121–141. https://doi.org/10.1037/1072-5245.14.2.121
- Yang, N. (2005). Individualism-collectivism and work-family interfaces: A Sino-U.S. comparison. In *Work and family: An international research perspective*. (pp. 287–318). Lawrence Erlbaum Associates Publishers.
- Yavas, U., Babakus, E., & Karatepe, O. M. (2008). Attitudinal and behavioral consequences of work-family conflict and family-work conflict: Does gender matter? *International Journal of Service Industry Management*, 19(1), 7–31. *https://doi.org/10.1108/09564230810855699*
- Yucel, D., & Borgmann, L. S. (2022). Work-family conflict and depressive symptoms among dual-earner couples in Germany: A dyadic and longitudinal analysis. *Social Science Research*, 104. https://doi.org/10.1016/j.ssresearch.2021.102684

- Yucel, D., & Fan, W. (2019). Work–family conflict and well-being among German couples: A longitudinal and dyadic approach. *Journal of Health and Social Behavior*, 60(3), 377– 395. https://doi.org/10.1177/0022146519870535
- Zapf, D., Dormann, C., & Frese, M. (1996). Longitudinal studies in organizational stress research: A review of the literature with reference to methodological issues. *Journal of Occupational Health Psychology*, 1(2), 145–169. https://doi.org/10.1037//1076-8998.1.2.145
- Zimmerman, B. J. (2000). Attaining self-regulation: A social cognitive perspective. In M. Boekaerts, P. R. Pintrich, & M. Zeidner (Eds.), *Handbook of Self-Regulation* (pp. 13–39). Academic Press. https://doi.org/10.1016/B978-012109890-2/50031-7

Appendix

Appendix of Chapter 2 Table A2. 1 *Descriptive Statistics*

		WtFI	FtWI	Life Sat. T1	Life Sat. T2	
Life-Goal Gr	oups	Range 4–20	Range 4–20	Range 1–10	Range 1–10	
		Mean (SD)	Mean (SD)	Mean (SD)	Mean (SD)	
	Men					
	(n = 673)	9.86 (3.43)	6.49 (2.44)	7.74 (1.53)	7.70 (1.50)	
Low Agentic	Women					
High	(<i>n</i> = 634)	9.25 (3.56)	6.42 (2.62)	7.74 (1.53)	7.78 (1.42)	
Communal	Total					
	(<i>n</i> = 1307)	9.57 (3.51)	6.46 (2.53)	7.74 (1.53)	7.74 (1.46)	
	Men					
	(n = 375)	9.76 (3.54)	6.65 (2.34)	7.45 (1.53)	7.42 (1.57)	
High Agentic—	Women					
Low	(<i>n</i> = 312)	9.17 (3.60)	6.37 (2.63)	7.54 (1.67)	/.58 (1.54)	
Communal	Total					
	(<i>n</i> = 687)	9.49 (3.58)	6.52 (2.48)	/.49 (1.59)	7.49 (1.56)	
	Men					
	(n = 313)	9.79 (3.26)	6.64 (2.54)	7.68 (1.27)	7.63 (1.35)	
High Agentic—	Women				7.60 (1.58)	
High	(<i>n</i> = 349)	9.21 (3.76)	6.60 (2.61)	7.62 (1.46)		
Communai	Total					
	(<i>n</i> = 662)	9.49 (3.54)	6.62 (2.58)	7.65 (1.38)	7.61 (1.48)	
Total	Men	0.00 (0.40)		7 (1 (1 1 0)	7 (1 (1 40)	
	(<i>n</i> = 1361)	9.82 (3.42)	6.57 (2.44)	7.64 (1.48)	7.61 (1.49)	
	Women					
	(<i>n</i> = 1295)	9.22 (3.62)	6.46 (2.62)	7.66 (1.55)	7.68 (1.50)	
	Total					
	(n = 2656)	9.53 (3.53)	6.51 (2.53)	7.65 (1.51)	7.64 (1.49)	

Table A2. 2

Correlation Matrix

	Gender	LA_HC	HA_LC	HA_HC	WtFI	FtWI	Life Sat. T1	Life Sat. T2
Gender								
LA_HC	005 .800							
HA_LC	040* .042							
HA_HC	.046* .019							
WtFI	084*** <.001	.011 .582	006 .767	007 .737				
FtWI	021 .275	022 .248	.002 .921	.024 .217	.330*** <.001			
Life Sat. T1	.006	.056**	064***	.000	260***	216***		
	.747	.004	.001	.984	<.001	<.001		
Life Sat. T2	.026	.063***	061**	011	194***	158***	.474***	
	.187	.001	.002	.559	<.001	<.001	<.001	

Note: $p \le 0.05$; $p \le 0.01$; $p \le 0.001$.

Figure A2. 1

Changes in Mean Values of WtFI from T1 to T2 for Different Life-Goal Groups



Estimated Marginal Means of WtFI

Covariates appearing in the model are evaluated at the following values: Married = ,5230, Full-Time Employed = ,6638, Have a Child = ,5994

Figure A2. 2

Changes in Mean Values of FtWI from T1 to T2 for Different Life-Goal Groups



Estimated Marginal Means of FtWI

Covariates appearing in the model are evaluated at the following values: Married = ,5230, Full-Time Employed = ,6638, Have a Child = ,5994

Figure A2. 3



Changes in Life Satisfaction from T1 to T2 for Different Life-Goal Groups

Covariates appearing in the model are evaluated at the following values: Married = .5230, Full-Time Employed = .6638, Have a Child = .5994

Appendix of Chapter 3

Table A3. 1

Descriptive Statistics

		Agentic	Goals	Communal	Goals	WtH	FI	FtW	/I	Satisfa	ction	Stre	SS	Self-Co	ntrol
Na	tionality	Range	1-5	Range 1	-5	Range	1–5	Range	1–5	Range	1-11	Range	1-5	Range	1-11
		Mean	SD	Mean	SD	Mean	SD	Mean	SD	Mean	SD	Mean	SD	Mean	SD
	Male $(n = 292)$	3,97	0,60	4,09	0,55	2,89	1,07	2,48	0,96	7,13	2,26	3,23	1,22	8,07	1,63
Turkish	Female $(n = 295)$	4,09	0,61	4,11	0,56	3,09	1,06	2,37	0,90	7,02	2,10	3,33	1,22	8,06	1,63
	Divers $(n = 3)$	3,36	0,43	3,97	0,55	3,94	0,69	3,28	1,17	7,33	3,22	3,67	1,53	8,00	2,85
	Total $(n = 590)$	4,02	0,61	4,10	0,56	2,99	1,07	2,43	0,93	7,07	2,18	3,28	1,22	8,06	1,63
	Male $(n = 358)$	3,25	0,70	3,44	0,66	2,49	0,99	2,11	0,92	7,47	1,86	2,83	1,12	7,95	1,54
German	Female $(n = 304)$	3,14	0,66	3,64	0,54	2,56	0,99	2,01	0,89	7,50	1,93	3,04	1,13	7,71	1,65
	Total (<i>n</i> = 662)	3,20	0,68	3,53	0,61	2,52	0,99	2,06	0,91	7,48	1,89	2,92	1,13	7,84	1,59
	Male (<i>n</i> = 650)	3,57	0,75	3,73	0,69	2,67	1,05	2,28	0,95	7,32	2,05	3,01	1,18	8,00	1,58
T. (1	Female $(n = 599)$	3,61	0,79	3,87	0,60	2,82	1,06	2,19	0,91	7,26	2,03	3,18	1,18	7,88	1,65
Total	Divers $(n = 3)$	3,36	0,43	3,97	0,55	3,94	0,69	3,28	1,17	7,33	3,22	3,67	1,53	8,00	2,85
	Total $(n = 1252)$	3,59	0,77	3,80	0,65	2,74	1,05	2,24	0,93	7,29	2,04	3,09	1,18	7,94	1,61

Mean Level Differences

		Turkish - German	Male - Female	Turkish Male - Turkish Female	German Male - German Female	Turkish Male - German Male	Turkish Female - German Female
Agentic	t-Test	22.677***	847	-2.324*	1.903*	14.039***	18.118***
Goals	Latent Structure	20.206***	677	-2.438**	2.113*	12.941***	15.917***
Communal Goals	t-Test	17.331***	-3.785****	400	-4.286***	13.832***	10.560***
	Latent Structure	13.074***	-3.897***	-1.196	-4.247***	10.190***	8.326***
WtFI	t-Test	8.107***	-2.560**	-2.314*	896	4.889***	6.347***
	Latent Structure	7.991***	-2.541**	-2.290*	-1.017	4.774***	6.462***
FtWI	t-Test	7.032***	1.711	1.406	1.481	4.973***	4.986***
	Latent Structure	6.815***	1.621	1.219	1.613	4.530***	5.058***
Life Satisfaction	t-Test	-3.523***	.502	.629	167	-2.076*	-2.914**
One-Item	Latent Structure	-	-	-	-	-	-
Stress	t-Test	5.335***	-2.534**	955	-2.357*	4.320***	3.014***
One-Item	Latent Structure	-	-	-	-	-	-
Self-Control	t-Test	2.483**	1.315	.098	1.890*	1.011	2.588**
	Latent Structure	2.321*	1.225	.224	1.590	.904	2.544**

Correlation Matrix

	Agentic Goals	Communal Goals	WtFI	FtWI	Life Satisfaction	Stress
Agentic Goals						
Communal Goals	.704**					
WtFI	.250**	.172**				
FtWI	.281**	.186**	.597**			
Life Satisfaction	.042	.120**	327**	205**		
Stress	.203**	.170**	.527**	.409**	353**	

Note. ** p < 0.01

The Relationship between Agentic Goals, WtFI, Life Satisfaction, and Stress after Bootstrapping

Std. Indirect Effects	Estimates	Bias-Corrected CI
Agentic -> Life Satisfaction	120	p=.000, 95% CI:155,089
Agentic -> Stress	.163	p=.000, 95% CI: .123, .205
Std. Direct Effects		
Agentic -> WtFI	.304	p=.000, 95% CI: .233, .369
Agentic -> Life Satisfaction	.146	p=.000, 95% CI: .079, .213
Agentic -> Stress	.075	p=.020, 95% CI: .012, .138
WtFI -> Life Satisfaction	394	p=.000, 95% CI:452,332
WtFI -> Stress	.536	p=.000, 95% CI: .483, .585
Std. Total Effects		
Agentic -> WtFI	.304	p=.000, 95% CI: .233, .369
Agentic -> Life Satisfaction	.027	p=.444, 95% CI:042, .095
Agentic -> Stress	.238	p=.000, 95% CI: .172, .304
WtFI -> Life Satisfaction	394	p=.000, 95% CI:452,332
WtFI -> Stress	.536	p=.000, 95% CI: .483, .585

The Relationship between Communal Goals, FtWI, Life Satisfaction and Stress after Bootstrapping

Std. Indirect Effects	Estimates	Bias-Corrected CI
Communal -> Life Satisfaction	053	p=.000, 95% CI:080,031
Communal -> Stress	.079	p=.000, 95% CI: .048, .113
Std. Direct Effects		
Communal -> FtWI	.191	p=.000, 95% CI: .115, .265
Communal -> Life Satisfaction	.157	p=.000, 95% CI: .089, .226
Communal -> Stress	.121	p=.000, 95% CI: .055, .184
FtWI -> Life Satisfaction	276	p=.000, 95% CI:338,213
FtWI -> Stress	.413	p=.000, 95% CI: .362, .462
Std. Total Effects		
Communal -> FtWI	.191	p=.000, 95% CI: .115, .265
Communal -> Life Satisfaction	.105	p=.003, 95% CI: .035, .174
Communal -> Stress	.200	p=.000, 95% CI: .133, .264
FtWI -> Life Satisfaction	276	p=.000, 95% CI:338,213
FtWI -> Stress	.413	p=.000, 95% CI: .362, .462

The Relationship between Agentic Goals, WtFI, Life Satisfaction, and Stress after

Std. Indirect Effects	Estimates	Bias-Corrected CI
Agentic -> Life Satisfaction	119	p=.000, 95% CI:169,078
Agentic -> Stress	.152	p=.000, 95% CI: .103, .208
Std. Direct Effects		
Agentic -> WtFI	.310	p=.000, 95% CI: .214, .404
Agentic -> Life Satisfaction	.061	p=.182, 95% CI:030, .155
Agentic -> Stress	.069	p=.138, 95% CI:022, .158
WtFI -> Life Satisfaction	384	p=.000, 95% CI:459,302
WtFI -> Stress	.491	p=.000, 95% CI: .415, .562
Std. Total Effects		
Agentic -> WtFI	.310	p=.000, 95% CI: .214, .404
Agentic -> Life Satisfaction	058	p=.247, 95% CI:148, .037
Agentic -> Stress	.221	p=.000, 95% CI: .125, .313
WtFI -> Life Satisfaction	384	p=.000, 95% CI:459,302
WtFI -> Stress	.491	p=.000, 95% CI: .415, .562

Bootstrapping for Low Self Control Group

The Relationship between Agentic Goals, WtFI, Life Satisfaction, and Stress after

Std. Indirect Effects	Estimates	Bias-Corrected CI
Agentic -> Life Satisfaction	126	p=.000, 95% CI:179,082
Agentic -> Stress	.185	p=.000, 95% CI: .126, .247
Std. Direct Effects		
Agentic -> WtFI	.324	p=.000, 95% CI: .221, .416
Agentic -> Life Satisfaction	.157	p=.001, 95% CI: .058, .262
Agentic -> Stress	.094	p=.042, 95% CI: .004, .182
WtFI -> Life Satisfaction	389	p=.000, 95% CI:480,294
WtFI -> Stress	.570	p=.000, 95% CI: .492, .641
Std. Total Effects		
Agentic -> WtFI	.324	p=.000, 95% CI: .221, .416
Agentic -> Life Satisfaction	.032	p=.510, 95% CI:066, .142
Agentic -> Stress	.278	p=.000, 95% CI: .180, .368
WtFI -> Life Satisfaction	389	p=.000, 95% CI:480,294
WtFI -> Stress	.570	p=.000, 95% CI: .492, .641

Bootstrapping for High Self-Control Group

The Relationship between Communal Goals, FtWI, Life Satisfaction, and Stress after

Estimates	Bias-Corrected CI
051	p=.001, 95% CI:088,020
.062	p=.001, 95% CI: .025, .103
.164	p=.001, 95% CI: .064, .263
.068	p=.138, 95% CI:023, .162
.143	p=.002, 95% CI: .057, .225
309	p=.000, 95% CI:385,228
.380	p=.000, 95% CI: .309, .448
.164	p=.001, 95% CI: .064, .263
.018	p=.694, 95% CI:076, .117
.205	p=.000, 95% CI: .116, .291
309	p=.000, 95% CI:385,228
.380	p=.000, 95% CI: .309, .448
	Estimates 051 .062 .164 .068 .143 309 .380 .164 .018 .205 309 .380

Bootstrapping for Low Self Control Group

The Relationship between Communal Goals, FtWI, Life Satisfaction, and Stress after Bootstrapping for High Self Control Group

Std. Indirect Effects	Estimates	Bias-Corrected CI
Communal -> Life Satisfaction	065	p=.000, 95% CI:116,031
Communal -> Stress	.124	p=.000, 95% CI: .073, .190
Std. Direct Effects		
Communal -> FtWI	.285	p=.000, 95% CI: .169, .404
Communal -> Life Satisfaction	.169	p=.002, 95% CI: .058, .288
Communal -> Stress	.110	p=.057, 95% CI:004, .210
FtWI -> Life Satisfaction	229	p=.000, 95% CI:327,128
FtWI -> Stress	.436	p=.000, 95% CI: .356, .510
Std. Total Effects		
Communal -> FtWI	.285	p=.000, 95% CI: .169, .404
Communal -> Life Satisfaction	.104	p=.053, 95% CI:002, .218
Communal -> Stress	.234	p=.000, 95% CI: .131, .332
FtWI -> Life Satisfaction	229	p=.000, 95% CI:327,128
FtWI -> Stress	.436	p=.000, 95% CI: .356, .510

The Relationship between Agentic Goals, WtFI, Life Satisfaction, and Stress after

Std. Indirect Effects	Estimates	Bias-Corrected CI
Agentic -> Life Satisfaction	156	p=.000, 95% CI:262,076
Agentic -> Stress	.168	p=.000, 95% CI: .083, .270
Std. Direct Effects		
Agentic -> WtFI	.322	p=.000, 95% CI: .161, .482
Agentic -> Life Satisfaction	.337	p=.000, 95% CI: .191, .479
Agentic -> Stress	.042	p=.585, 95% CI:113, .193
WtFI -> Life Satisfaction	484	p=.000, 95% CI:609,363
WtFI -> Stress	.521	p=.000, 95% CI: .403, .621
Std. Total Effects		
Agentic -> WtFI	.322	p=.000, 95% CI: .161, .482
Agentic -> Life Satisfaction	.181	p=.024, 95% CI: .021, .332
Agentic -> Stress	.210	p=.010, 95% CI: .053, .355
WtFI -> Life Satisfaction	484	p=.000, 95% CI:609,363
WtFI -> Stress	.521	p=.000, 95% CI: .403, .621

Bootstrapping for German Women

The Relationship between Agentic Goals, WtFI, Life Satisfaction, and Stress after

Std. Indirect Effects	Estimates	Bias-Corrected CI
Agentic -> Life Satisfaction	076	p=.011, 95% CI:147,019
Agentic -> Stress	.095	p=.009, 95% CI: .022, .166
Std. Direct Effects		
Agentic -> WtFI	.202	p=.012, 95% CI: .044, .344
Agentic -> Life Satisfaction	.035	p=.658, 95% CI:105, .168
Agentic -> Stress	.136	p=.036, 95% CI: .008, .256
WtFI -> Life Satisfaction	379	p=.000, 95% CI:488,262
WtFI -> Stress	.470	p=.000, 95% CI: .366, .562
Std. Total Effects		
Agentic -> WtFI	.202	p=.012, 95% CI: .044, .344
Agentic -> Life Satisfaction	042	p=.568, 95% CI:181, .100
Agentic -> Stress	.230	p=.002, 95% CI: .087, .359
WtFI -> Life Satisfaction	379	p=.000, 95% CI:488,262
WtFI -> Stress	.470	p=.000, 95% CI: .366, .562

Bootstrapping for Turkish Women
Table A3. 12

The Relationship between Communal Goals, FtWI, Life Satisfaction and Stress after

Bootstrapping for German Men

Std. Indirect Effects	Estimates	Bias-Corrected CI
Communal -> Life Satisfaction	017	p=.078, 95% CI:064, .002
Communal -> Stress	.052	p=.110, 95% CI:013, .134
Std. Direct Effects		
Communal -> FtWI	.119	p=.122, 95% CI:034, .288
Communal -> Life Satisfaction	.312	p=.000, 95% CI: .161, .448
Communal -> Stress	.048	p=.493, 95% CI:093, .182
FtWI -> Life Satisfaction	139	p=.019, 95% CI:252,023
FtWI -> Stress	.434	p=.000, 95% CI: .337, .526
Std. Total Effects		
Communal -> FtWI	.119	p=.122, 95% CI:034, .288
Communal -> Life Satisfaction	.296	p=.000, 95% CI: .149, .428
Communal -> Stress	.100	p=.169, 95% CI:043, .240
FtWI -> Life Satisfaction	139	p=.019, 95% CI:252,023
FtWI -> Stress	.434	p=.000, 95% CI: .337, .526

Table A3. 13

The Relationship between Communal Goals, FtWI, Life Satisfaction and Stress after

Bootstrapping for Turkish Men

Std. Indirect Effects	Estimates	Bias-Corrected CI
Communal -> Life Satisfaction	016	p=.415, 95% CI:065, .026
Communal -> Stress	.022	p=.434, 95% CI:036, .087
Std. Direct Effects		
Communal -> FtWI	.054	p=.442, 95% CI:095, .206
Communal -> Life Satisfaction	.218	p=.001, 95% CI: .089, .336
Communal -> Stress	.091	p=.188, 95% CI:046, .221
FtWI -> Life Satisfaction	285	p=.000, 95% CI:397,171
FtWI -> Stress	.400	p=.001, 95% CI: .285, .504
Std. Total Effects		
Communal -> FtWI	.054	p=.442, 95% CI:095, .206
Communal -> Life Satisfaction	.203	p=.003, 95% CI: .069, .328
Communal -> Stress	.113	p=.128, 95% CI:033, .250
FtWI -> Life Satisfaction	285	p=.000, 95% CI:397,171
FtWI -> Stress	.400	p=.001, 95% CI: .285, .504

Appendix of Chapter 4

Table A4.

Intervention and Control Group Differences Regarding Baseline Self-Control Scores

Group	Self-Control T1			Self-Control T2				
	Below Median	Median	Above Median	Total	Below Median	Median	Above Median	Total
Control	119	26	113	258	120	32	105	257
Intervention	80	23	80	183	94	14	72	180
Total	199	49	193	441	214	46	177	437

Note. No significant differences were found between intervention and control groups in

baseline, $\chi^2(2) = .736$, p = .69, and in post-intervention, $\chi^2(2) = 2.88$, p = .24.

Figure A4.

Action Planning and Action Control Exercise

	oura nave maintained a planning journal.				
ACT	IONS				
)	pps on my phone.				
)					
WHEN	WHERE				
At the end of working hours.	At home or on the way there.				
<u> </u>	<u> </u>				
IF SOMETHING CROPS UP	THEN				
I receive a call from my manager,	I will verify whether the call is				
	urgent.				
Of its urgent,	J will arrange a time to do the work				
	when the children are in bed.				

Declaration of Own Contribution

I hereby declare that this dissertation has been produced by myself. No unauthorized aids were used. I declare that all citations, excerpts, and ideas of others are indicated.

Chapter 2 was co-authored with Sonja Drobnič and Klaus Boehnke, and Chapter 4 was coauthored with Sonia Lippke. I am the corresponding author and made the main contributions to each paper.

I permit the review of the thesis via qualified software in case of an accusation of plagiarism. No part of the thesis has been accepted or submitted for any other qualification at the University of Bremen, Constructor University Bremen, or any other university.

Gamze Ipek

Bremen,